PREVALENCE OF AND RISK FACTORS FOR BODY FATNESS AND NUTRITIONAL STATUS OF URBAN AND RURAL PRIMARY SCHOOL CHILDREN BETWEEN THE AGES OF SIX AND NINE YEARS IN THE POLOKWANE LOCAL MUNICIPALITY, LIMPOPO PROVINCE, SOUTH AFRICA

by

MAMOGOBO NELLY MOKABANE

DISSERTATION

Submitted in fulfilment of the requirements for the degree of **MASTER OF SCIENCE**

in

PHYSIOLOGY

in the

FACULTY OF SCIENCE AND AGRICULTURE (School of Molecular and Life Science)

at the

UNIVERSITY OF LIMPOPO

SUPERVISOR: Dr. M van Staden CO-SUPERVISOR: Mr. H Malan

2019

DECLARATION

I, Mamogobo Nelly Mokabane, declare that "Differences in body fatness and nutritional status of urban and rural primary school children between the ages of six and nine years in the Polokwane Local Municipality, Limpopo Province, South Africa," hereby submitted to the University of Limpopo, in fulfilment of the requirements for the degree of Master of Science in Physiology, is my own work and that all the sources that I have used or quoted have been indicated and acknowledged by means of complete references, and that this work has not been submitted before for any other degree at any other institution.

Ms MN Mokabane

Date

DEDICATION

This work is dedicated to the following people:

- **God**, thank you for helping me turn all my setbacks and failures into a stepping stone. You are my saving grace.
- **To my family**: my parents and brother for their prayers, emotional, financial, academic and moral support.

ACKNOWLEDGEMENTS

I would like to express my appreciation and thanks to the following people for their respective contributions in making this study possible:

- A special thank you to **my supervisor**, Dr M van Staden, for giving me the opportunity to pursue this Masters degree, and for her guidance and support throughout this journey.
- My **co-supervisor**, Mr H Malan, thank you for your guidance on language, writing, and technical editing.
- **My Honours supervisor**, Ms MM Mashao, for her patience, guidance, constructive criticism and words of encouragement throughout my postgraduate years.
- My parents, Matthews and Matsie, thank you for your sacrifices, motivating and believing in me whenever I lost hope. My sibling Bernard, thank you for your support and encouragement. My aunt Daphney and her son Lebogang, thank you for your love, support and encouragement. My cousin Sekana and his wife Ester, thank you for your guidance, encouragement, love and support throughout my academic jounery.
- My dear friends, thank you for your support and encouragement.
- The **Honours students**, class of 2016, for their hard work and contribution towards data collection.
- The National Research Foundation for funding my postgraduate studies.
- The Limpopo Department of Education, district offices, circuit managers, school authorities, principals, teachers and parents in the Polokwane Local Municipality for their cooperation.
- The primary school children who voluntarily participated in this project.
- The biostatistician, **Dr S Ntuli**, for helping and sharing his excellent statistical knowledge with me.

ABSTRACT

Aim: The purpose of this study was to assess the prevalence and risk factors associated with malnutrition and nutritional status of rural and urban primary school children aged 6–9 years.

Literature background: Malnutrition is one of the leading causes of childhood morbidity and mortality in developing countries, affecting 10.9% of people globally, particularly in Southern Asian (15.7%) and sub-Saharan African (23.2%) countries undergoing urbanisation. Urbanisation affects diet, physical activity levels, body fatness, body composition and socio-economic factors. There is currently a shortage of information on the effect of urbanisation on nutritional status, especially in poor areas such as the Limpopo Province of South Africa. It is important to understand the effects that body fatness and associated risk factors have on stunting, wasting, underweight, and overweight/obesity in urban and rural children.

Subjects and design: This was a cross-sectional study including rural (n=106) and urban (n=68) primary school children aged 6–9 years. Anthropometric (weight and height) and skinfold measurements were taken and a 24-hour recall dietary assessment was conducted twice to include a week day and a weekend day. A questionnaire was used to gather demographic, health, dietary and physical activity information.

Results: The prevalence of stunting, wasting, underweight, and overweight/obese children in this population was 14%, 6%, 20%, and 26% respectively. Furthermore, the prevalence stunting, wasting, and underweight were higher in rural areas compared to urban areas. Meanwhile, the prevalence of overweight/obesity was higher in urban areas as compared to rural areas. Gender, area (urban/rural) (p=0.0001), birthweight, time spent on sedentary activities, monthly household income (p=0.0210), mode of transport to school, and breastfeeding (p=0.0560) were all found to be significantly associated with malnutrition. Weekday dietary intake of Vitamins A and D was significally associated with BF%, whereas weekend consumption of energy, protein, calcium, iron, phosphorus, and zinc were not significantly associated with BF%. The mean energy and calcium intake between the urban and rural

۷

population only differed significantly over weekends and not during the week, while vitamin D differed significantly between urban and rural populations during the week only.

Conclusions: The current study demonstrates that children in rural areas were significantly more wasted than those residing in urban areas, while those residing in urban areas were significantly more overweight/obese compared to rural areas. All indicators used for undernutrition were associated significantly with gender and the prevalence were higher males compared to females, but this significance disappeared for overnutrition. Families with a higher income tended to have children who were overweight or obese compared to families with a lower income. Children spending a lot of time on sedentary activities were more likely to be overweight or obese, and breastfeeding seemed to protect children from becoming overweight or obese, while low birth weight was associated with stunting. The weekaday and week-end diets indicate that school feeding schemes in the rural areas may be effective in increasing total energy intake in children and this should be further investigated. In this population, stunting, wasting, and underweight were not limited to rural areas, and should still be a health concern in urban areas, despite the fact that overweight/obesity is also prevalent in urban areas. Thus, overweight/obesity in this population can be prevented by promoting breastfeeding and physical activity, while at the same time discouraging children from partaking in too many sedentary activities.

Key words: Rural, urban, children, malnutrition, nutritional status, prevalence, urbanisation, stunting, wasting, underweight, overweight/obesity.

vi

Table of Contents

DECLARATION	
DEDICATION	
ACKNOWLEDGEMENTS	IV
ABSTRACT	V
LIST OF FIGURES	XII
LIST OF TABLES	XIII
LIST OF ADDENDA	XIV
LIST OF ABBREVIATIONS	XV

CH	APTER ONE: INTRODUCTION AND BACKGROUND INFORMATION	1
1	INTRODUCTION/BACKGROUND	1
2	RESEARCH PROBLEM	2
3	SIGNIFICANCE OF THE RESEARCH	3
4	PURPOSE OF THE STUDY	3
	4.1 Aim	3
	4.2 Objectives	3
5	SCOPE AND LIMITATIONS OF THE STUDY	4
	5.1 Scope	4
	5.2 Limitations	4
	5.3 Permission to conduct the study	5
	5.4 Harm to respondents	5
	5.5 Informed consent	5
	5.6 Confidentiality and anonymity	6
	5.7 Release or publication of the findings	6
6	STRUCTURE OF DISSERTATION	6
СН	APTER TWO: LITERATURE REVIEW	7
1	OVERVIEW OF MALNUTRITION	7
2		8
	2.1 Overview of overnutrition	8
	2.2 The prevalence of childhood overweight/obesity	9

	2.2.1	The prevalence of childhood overweight/obesity among children globa	lly . 9
	2.2.2	The prevalence of childhood overweight/obesity in South Africa	. 9
	2.2.3	The prevalence of childhood overweight/obesity in urban and rural area	as 10
	2.3 The	e risk factors associated with childhood overweight/obesity	10
	2.3.1	Age	11
	2.3.2	Transition	11
	2.3.3	The socioeconomic and cultural factors	11
	2.3.4	The genetic contribution of obesity	12
	2.3.5	The link between obesity and family history	12
	2.3.6	Intra-uterine environment and early life	12
	2.3.7	The effect of dietary energy intake on body fatness	13
	2.3.8	The effects of physical activity on body weight	14
	2.3.9	The effect of leptin on the development of obesity	15
	2.4 The	e causal mechanism associated with increased body fatness	15
	2.5 The	e complications associated with overweight/obesity	18
3	Under	RNUTRITION	19
	3.1 Ove	erview of stunting	19
	3.1 Ove 3.1.1	erview of stunting The prevalence of stunting globally	<i>19</i> 19
	3.1 Ove 3.1.1 3.1.2	erview of stunting The prevalence of stunting globally The prevalence of stunting in South Africa	<i>19</i> 19 19
	3.1 Ove 3.1.1 3.1.2 3.1.3	erview of stunting The prevalence of stunting globally The prevalence of stunting in South Africa The prevalence of stunting in urban and rural areas	<i>19</i> 19 19 19
	3.1 Ove 3.1.1 3.1.2 3.1.3 3.2 Ove	erview of stunting The prevalence of stunting globally The prevalence of stunting in South Africa The prevalence of stunting in urban and rural areas	<i>19</i> 19 19 19 20
	3.1 Ove 3.1.1 3.1.2 3.1.3 3.2 Ove 3.2.1	The prevalence of stunting globally The prevalence of stunting globally The prevalence of stunting in South Africa The prevalence of stunting in urban and rural areas <i>Erview of wasting</i> The prevalence of wasting globally	<i>19</i> 19 19 20 20
	3.1 Ove 3.1.1 3.1.2 3.1.3 3.2 Ove 3.2.1 3.2.2	The prevalence of stunting globally The prevalence of stunting in South Africa The prevalence of stunting in urban and rural areas <i>The prevalence of stunting in urban and rural areas</i> <i>erview of wasting</i> The prevalence of wasting globally The prevalence of wasting in South Africa	19 19 19 20 20 20
	3.1 Ove 3.1.1 3.1.2 3.1.3 3.2 Ove 3.2.1 3.2.2 3.2.3	The prevalence of stunting globally The prevalence of stunting in South Africa The prevalence of stunting in urban and rural areas erview of wasting	19 19 19 20 20 20 20
	3.1 Ove 3.1.1 3.1.2 3.1.3 3.2 Ove 3.2.1 3.2.2 3.2.3 3.3 Ove	Prevaluation of stunting Image: Stunting and Stunting and Stunting in South Africa The prevalence of stunting in South Africa Image: Stunting in Urban and rural areas The prevalence of stunting in urban and rural areas Image: Stunting in Urban and rural areas The prevalence of stunting in Urban and rural areas Image: Stunting in Urban and rural areas The prevalence of wasting globally Image: Stunting in Urban and rural areas The prevalence of wasting in South Africa Image: Stunting in Urban and rural areas The prevalence of wasting in Urban and rural areas Image: Stunting in Urban and rural areas The prevalence of wasting in Urban and rural areas Image: Stunting in Urban and rural areas The prevalence of wasting in Urban and rural areas Image: Stunting in Urban and rural areas The prevalence of wasting in Urban and rural areas Image: Stunting in Urban and rural areas The prevalence of wasting in Urban and rural areas Image: Stunting in Urban and rural areas The prevalence of wasting in Urban and rural areas Image: Stunting in Urban and rural areas The prevalence of wasting in Urban and rural areas Image: Stunting in Urban and rural areas The prevalence of wasting in Urban and rural areas Image: Stunting in Urban and rural areas The prevalence of wasting in Urban and rural areas Image: Stuning	19 19 19 20 20 20 20 21
	3.1 Ove 3.1.1 3.1.2 3.1.3 3.2 Ove 3.2.1 3.2.2 3.2.3 3.3 Ove 3.3.1	erview of stunting The prevalence of stunting globally The prevalence of stunting in South Africa The prevalence of stunting in urban and rural areas erview of wasting The prevalence of wasting globally The prevalence of wasting globally The prevalence of wasting globally The prevalence of wasting in South Africa The prevalence of wasting in South Africa The prevalence of wasting in urban and rural areas The prevalence of wasting in urban and rural areas The prevalence of wasting in urban and rural areas The prevalence of wasting in urban and rural areas The prevalence of wasting in urban and rural areas The prevalence of underweight	19 19 19 20 20 20 20 21 21
	3.1 Ove 3.1.1 3.1.2 3.1.3 3.2 Ove 3.2.1 3.2.2 3.2.3 3.3 Ove 3.3.1 3.3.2	erview of stunting The prevalence of stunting globally The prevalence of stunting in South Africa The prevalence of stunting in urban and rural areas erview of wasting The prevalence of stunting globally The prevalence of wasting globally The prevalence of wasting globally The prevalence of wasting in South Africa The prevalence of wasting in urban and rural areas The prevalence of wasting in urban and rural areas Erview of underweight The prevalence of underweight globally The prevalence of underweight in South Africa	 19 19 19 20 20 20 20 21 21 21
	3.1 Ove 3.1.1 3.1.2 3.1.3 3.2 Ove 3.2.1 3.2.2 3.2.3 3.3 Ove 3.3.1 3.3.2 3.3.3	erview of stunting The prevalence of stunting globally The prevalence of stunting in South Africa The prevalence of stunting in urban and rural areas erview of wasting The prevalence of stunting in urban and rural areas Erview of wasting The prevalence of wasting globally The prevalence of wasting globally The prevalence of wasting in South Africa The prevalence of wasting in urban and rural areas Erview of underweight The prevalence of underweight globally The prevalence of underweight in South Africa The prevalence of underweight in South Africa	 19 19 19 20 20 20 20 21 21 21 21 21
	3.1 Ove 3.1.1 3.1.2 3.1.3 3.2 Ove 3.2.1 3.2.2 3.2.3 3.3 Ove 3.3.1 3.3.2 3.3.3 3.4 The	erview of stunting The prevalence of stunting globally The prevalence of stunting in South Africa The prevalence of stunting in urban and rural areas erview of wasting The prevalence of stunting in urban and rural areas erview of wasting The prevalence of stunting in urban and rural areas The prevalence of wasting globally The prevalence of wasting in South Africa The prevalence of wasting in urban and rural areas	19 19 19 20 20 20 21 21 21 21 21 21
	3.1 Ove 3.1.1 3.1.2 3.1.3 3.2 Ove 3.2.1 3.2.2 3.2.3 3.3 Ove 3.3.1 3.3.2 3.3.3 3.4 The 3.4.1	erview of stunting The prevalence of stunting globally The prevalence of stunting in South Africa The prevalence of stunting in urban and rural areas erview of wasting The prevalence of wasting globally The prevalence of wasting globally The prevalence of wasting globally The prevalence of wasting in South Africa The prevalence of wasting in South Africa The prevalence of wasting in urban and rural areas erview of underweight The prevalence of underweight globally The prevalence of underweight in South Africa The prevalence of underweight in South Africa The prevalence of underweight in urban and rural areas erview of undermutrition in children The prevalence of underweight in urban and rural areas	19 19 19 20 20 20 21 21 21 21 21 21 21
	3.1 Ove 3.1.1 3.1.2 3.1.3 3.2 Ove 3.2.1 3.2.2 3.2.3 3.3 Ove 3.3.1 3.3.2 3.3.3 3.4 The 3.4.1 3.4.2	erview of stunting The prevalence of stunting globally The prevalence of stunting in South Africa The prevalence of stunting in urban and rural areas erview of wasting The prevalence of wasting globally The prevalence of wasting globally The prevalence of wasting globally The prevalence of wasting in South Africa The prevalence of wasting in urban and rural areas The prevalence of wasting in urban and rural areas Erview of underweight The prevalence of underweight globally The prevalence of underweight in South Africa The prevalence of underweight in urban and rural areas Ervisk factors of undernutrition in children The socio economic factors on undernutrition in children The link between undernutrition and family history	 19 19 19 20 20 20 20 21 2

	3.4.4	Birthweight	22
	3.4.5	Breastfeeding	23
	3.4.6	The effect of diet on undernutrition diet	24
	3.4.7	Infections	24
	3.5 Cau	usal mechanisms associated with undernutrition (stunting, wasting and	1
	una	lerweight)	25
	3.6 Cor	nplications associated with undernutrition	27
4	SUMM	ARY	28

CH/	APTER THREE: RESEARCH METHODOLOGY	29
1	INTRODUCTION	29
2	STUDY AREA	29
3	STUDY DESIGN	29
4	EXCLUSION AND INCLUSION CRITERIA	29
5	SAMPLE SIZE AND TECHNIQUE	30
6	STUDY POPULATION	30
7	RESEARCH METHODOLOGY	31
8	DATA COLLECTION	32
	8.1 Organisational procedure	. 32
	8.2 Questionnaire	. 32
	8.3 Measuring malnutrition in children	33
	8.3.1 Criteria for determining malnutrition in children	33
	8.3.2 Criteria for determining the recommended daily allowance	of
	macronutrients and micronutrients in children.	34
	8.3.3 Anthropometric measurements	36
	8.3.3.1 Anthropometric and nutritional status calculations	36
	8.4 Measuring dietary intake in children	. 36
	8.5 24-hour recall test	. 37
9	STATISTICAL ANALYSIS	38
10	0 ETHICAL CONSIDERATIONS	38
1	1 RELIABILITY, VIABILITY AND OBJECTIVITY	39
1:	2 BIAS	39

СН	APTER FOUR: PRESENTATION AND INTERPRETATION OF FINDINGS 40
1	INTRODUCTION
2	DEMOGRAPHIC PROFILE AND MEDICAL HISTORY OF THE STUDY PARTICIPANTS 40
	2.1 Age distribution of the children
	2.1.1 The age distribution of the children in the rural versus the urban areas .
	2.1.2 The age distribution of the children between males and females 41
	2.2 Gender distribution of the children
	2.3 Age distribution of the parents/guardians
	2.4 Monthly household income
	2.5 Monthly household income between urban and rural population
	2.6 Availability of motor vehicle
	2.7 Parents medical history
3	PREVALENCE OF MALNUTRITION (STUNTING, WASTING, UNDERWEIGHT,
	OVERWEIGHT/OBESITY) IN CHILDREN BETWEEN URBAN AND RURAL AREAS
4	PREVALENCE OF RISK FACTORS FOR STUNTING, WASTING, UNDERWEIGHT,
	OVERWEIGHT/OBESITY
	4.1 Stunting
	4.2 Wasting
	4.3 Underweight
	4.4 Overweight/obesity
5	PEARSON CORRELATION BETWEEN WEEKDAY AND WEEKEND DIETARY INTAKE AND
	BODY FAT PERCENTAGE OF THE STUDY POPULATION
CH/	APTER FIVE: DISCUSSION
1	INTRODUCTION
2	PREVALENCE OF STUNTING, WASTING, UNDERWEIGHT, OVERWEIGHT/OBESITY IN
	CHILDREN BETWEEN URBAN AND RURAL AREAS
	2.1 Stunting
	2.2 Wasting
	2.3 Underweight
	2.4 Overweight/obesity 60
3	THE PREVALENCE OF RISK FACTORS OF STUNTING, WASTING, UNDERWEIGHT,
	OVERWEIGHT/OBESITY

	3.1 Stunting
	3.2 Wasting
	3.3 Underweight
	3.4 Overweight/obesity
4	PEARSON CORRELATION BETWEEN WEEKDAY AND WEEKEND DIETARY INTAKE AND
	BF% OF THE STUDY POPULATION
CH	APTER SIX: CONCLUSION AND RECOMMENDATIONS
1	CONCLUSION
2	LIMITATIONS
3	RECOMMENDATIONS
REI	FERENCES
REI A	FERENCES
REI A A	FERENCES
REI A A A	FERENCES
REI A A A	FERENCES 75 DDENDUM A: TREC CERTIFICATE 98 DDENDUM B: LETTER FROM THE CIRCUIT MANAGER 99 DDENDUM C: A LETTER FROM DEPARTMENT OF EDUCATION 100 DDENDUM D: A FORMAL LETTER FOR PERMISSION TO CONDUCT THE
REI A A A	FERENCES 75 DDENDUM A: TREC CERTIFICATE 98 DDENDUM B: LETTER FROM THE CIRCUIT MANAGER 99 DDENDUM C: A LETTER FROM DEPARTMENT OF EDUCATION 100 DDENDUM D: A FORMAL LETTER FOR PERMISSION TO CONDUCT THE STUDY 102
REI A A A	FERENCES 75 DDENDUM A: TREC CERTIFICATE 98 DDENDUM B: LETTER FROM THE CIRCUIT MANAGER 99 DDENDUM C: A LETTER FROM DEPARTMENT OF EDUCATION 100 DDENDUM D: A FORMAL LETTER FOR PERMISSION TO CONDUCT THE STUDY 102 DDENDUM E: A FORMAL LETTER TO REQUEST PERMISSION TO COLLECT 102
REI A A A	FERENCES 75 DDENDUM A: TREC CERTIFICATE 98 DDENDUM B: LETTER FROM THE CIRCUIT MANAGER 99 DDENDUM C: A LETTER FROM DEPARTMENT OF EDUCATION 100 DDENDUM D: A FORMAL LETTER FOR PERMISSION TO CONDUCT THE STUDY 102 DDENDUM E: A FORMAL LETTER TO REQUEST PERMISSION TO COLLECT DATA 106
REI A A A A	FERENCES 75 DDENDUM A: TREC CERTIFICATE 98 DDENDUM B: LETTER FROM THE CIRCUIT MANAGER 99 DDENDUM C: A LETTER FROM DEPARTMENT OF EDUCATION. 100 DDENDUM D: A FORMAL LETTER FOR PERMISSION TO CONDUCT THE STUDY. 102 DDENDUM E: A FORMAL LETTER TO REQUEST PERMISSION TO COLLECT DATA 106 DDENDUM F: CONSENT FORMS (ENGLISH, SEPEDI AND AFRIKAANS). 109
REI A A A A A	FERENCES 75 DDENDUM A: TREC CERTIFICATE 98 DDENDUM B: LETTER FROM THE CIRCUIT MANAGER 99 DDENDUM C: A LETTER FROM DEPARTMENT OF EDUCATION 100 DDENDUM D: A FORMAL LETTER FOR PERMISSION TO CONDUCT THE STUDY 102 DDENDUM E: A FORMAL LETTER TO REQUEST PERMISSION TO COLLECT DATA 106 DDENDUM F: CONSENT FORMS (ENGLISH, SEPEDI AND AFRIKAANS) 109 DDENDUM G: QUESTIONNAIRES (ENGLISH, SEPEDI AND AFRIKAANS) 115

LIST OF FIGURES

- - parents/guardians45

LIST OF TABLES

Table 3.1:	Composition of the intended study population	
Table 3.2:	The criteria used to classify malnutrition in children aged six to nine years	
	(WHO, 2007)	
Table 3.3:	The criteria used for determining recommended dietary allowance of	
	micronutrients and macronutrients requirements for children between 4-	
	13 years of age according to http://www.nap.edu/openbook.php?isbn	
	=0309085373	
Table 4.1:	The age distribution of children from urban versus rural areas	
Table 4.2:	The age distribution of the children between males and females42	
Table 4.3:	Monthly household income	
Table 4.4:	Descriptive statistics of anthropometric measurements and calculations	
	used to determine nutritional status in the urban and rural population. 46	
Table 4.5:	Prevalence of stunting, wasting, underweight, overweight/obesity 47	
Table 4.6:	Risk factors for stunting	
Table 4.7:	Risk factors for wasting	
Table 4.8:	Risk factors for underweight	
Table 4.9:	Risk factors for overweight/obesity	
Table 4.10:	0: Descriptive statistics for food consumed per week in the urban and rural	
	population	
Table 4.11:	Descriptive statistics of weekday macronutrients and micronutrients	
	between urban and rural population54	
Table 4.12:	Descriptive statistics of weekend macronutrients and micronutrients	
	between urban and rural population54	
Table 4.13:	Pearson correlation between weekday dietary intake and body fat	
	percentage of the study population	
Table 4.14:	Pearson correlation between weekend dietary intake and body fat	
	percentage of the study population	
Table 6.1:	The risk factors associated with malnutrition in this study population 72	

LIST OF ADDENDA

Addendum	Name of addendum	Page number
Addendum A	TREC Certificate	83
Addendum B	Letter from the circuit manager	84
Addendum C	Letter from Department of Education	85
Addendum D	A formal letter for permission to conduct the study	87
Addendum E	A formal letter to request permission to collect data	91
Addendum F	Consent forms (English, Sepedi and Afrikaans)	94
Addendum G	Questionnaires (English, Sepedi and Afrikaans)	100
Addendum H	Feedback chart	116

LIST OF ABBREVIATIONS

Abbreviation	Description
BF%	Body fat percentage
BMI	Body mass index
BMR	Basal metabolic rate
Ca	Calcium
CCK	Cholecystokinin
CDLs	Chronic diseases of lifestyle
Cu	Copper
DHHS	Department of Health and Human Services
FAO	Food and Agriculture Organisation
FDA	Food and drug administration
Fe	Iron
HFA	Height-for-age
WFH	Weight for height
ISAK	International Society for the Advancement of Kinanthropometry
Kg	Kilogram
Р	Phosphorus
PLM	Polokwane Local Municipality
POMC	Pro-opiomelanocortin
PUFA	Polyunsaturated fatty acids
PYY	Peptide YY
RDA	Recommended Daily Allowance
ROS	Reactive oxygen species
SA	South Africa
SAMRC	South African Medical Research Council
SD	Standard deviation
Se	Selenium
Vit A	Vitamin A
Vit D	Vitamin D
Vit E	Vitamin E
WFA	Weight-for-age
WHO	World Health Organization
Zn	Zinc

Chapter One: Introduction and background information

1 Introduction/background

Malnutrition is a major public health problem affecting 10.9% of people globally, particularly in Southern Asian (15.7%) and sub-Saharan African (23.2%) countries undergoing urbanisation (FAO, 2015). Urbanisation can be defined as a movement of individuals from rural to urban areas and the development of rural areas into urban areas (Ojiambo *et al.*, 2012). It has a huge effect on the lives of individuals and is associated with changes in, amongst other factors, diet, physical activity levels (Yamauchi *et al.*, 2001), body fatness, body composition, and socio-economic factors (Dinsa *et al.*, 2012). Currently South Africa (SA) is in the midst of a huge sector of the population moving from rural areas towards urban areas (Monyeki *et al.*, 2015). To complicate matters, the rural areas themselves are becoming more and more urban due to developments such as tar roads, shops, hospitals and clinics in these areas (Tydeman-Edwards *et al.*, 2018). This has a huge impact on the lifestyle of the population such as abnormal body fatness (You & Henneberg, 2016).

Abnormal body fatness and stunting are caused by a variety of factors such as diet, maternal age, body composition, and socio-economic status (Leddy *et al.*, 2008). Body composition profiles such as stunting, wasting, underweight, overweight/obesity can result in poor growth, poor development, and negative health outcomes, such as type 2 diabetes mellitus and hypertension (Kimani- Murage *et al.*, 2010). Stunting and abnormal body fatness also have negative effects on learning abilities in children (Caulfield *et al.*, 2004).

Globally, over 22 million children are overweight (Musaiger, 2011). Childhood obesity is not only a problem in developed countries, but it is also becoming an increasing problem in countries undergoing epidemiological transition, such as SA; this is seen especially in girls (Puoane *et al.*, 2002). Girls between the ages of four and fourteen from countries such as Ethiopia, SA, and Zimbabwe have higher mean body mass

1

indices as compared to boys of the same ages (Mascie-Taylor & Goto, 2007). Childhood obesity in SA is a major public health problem and is on the increase (Karnik & Kanekar, 2012). The prevalence of overweight/obesity in children varies with age, gender, population group and level of urbanisation (Lobstein & Frelut, 2003).

Low birth weight is a risk factor for both overweight and stunting (Arifeen *et al.*, 2000). Worldwide more than 20 million babies are born with low birth weight every year due to factors such as poor maternal nutrition (UNICEF, 2012). Developing and middle income country such as SA has a relatively higher levels of under-nutrition compared to other middle-income countries (Sunguya *et al.*, 2014). Malnutrition is one of the leading causes of death in Limpopo Province (UNICEF, 2009). It is likely that SA (including Limpopo Province) has a high prevalence of low birth weight due to the high prevalence of malnutrition (Kleynhans *et al.*, 2006). As far as the author could ascertain, this was the first study comparing body fatness and nutritional status in rural and urban children to have been conducted in the Polokwane Local Municipality (PLM), Capricorn District, Limpopo Province, SA.

In a developing country such as SA, where a large sector of the population is currently affected by urbanisation (Monyeki *et al.*, 2015), it is possible that the changes in lifestyle associated with urbanisation negatively affects the body fatness and body composition of prepubertal children (between the ages of six and nine years). It is of the utmost importance to increase the knowledge on the effect that urbanisation has on the development of these conditions and on the co-occurrence of stunting and overweight, especially in low-income areas such as the Limpopo Province.

2 Research problem

Stunting, wasting, underweight, overweight and obesity are all common conditions in developing countries such as SA (Tydeman-Edwards *et al.*, 2018) and their prevalence is increasing with increasing urbanisation and modernisation (Monyeki *et al.*, 2015). Overweight and stunting can even co-occur in one child (Symington *et al.*, 2016). All these conditions may have a negative impact on children later in life by increasing their risk for the development of other chronic diseases of lifestyle (CDL), including type 2 diabetes mellitus, hypertension, and stroke (Pulos *et al.*, 2014). There

currently is a shortage of information on the effect that urbanisation has on the development of these conditions and on the co-occurrence of stunting and overweight, especially in poor areas such as the Limpopo Province.

3 Significance of the research

The findings of this study will make a valuable contribution to knowledge of how urbanisation affects body fatness and nutritional status of primary school children. The Department of Health and the Department of Basic Education may be able to use the results to design effective programmes to prevent abnormal body fatness and stunting in children, especially those from rural environments. The results may also be used to ensure that children (and probably parents) are informed about practical ways to ensure that a prudent diet is followed. This study will also help in understanding the effects that body fat and diet have on the development and growth of the children.

4 Purpose of the study

4.1 Aim

The aim of this study was to assess the prevalence, risk factors associated with malnutrition and nutritional status of urban (n = 68) and rural (n = 106) primary school children aged 6–9 years in the PLM, Capricorn District, Limpopo Province, SA.

4.2 Objectives

The objectives of this study were to:

- gather demographic and health information with a questionnaire to compare the prevalence of stunting, wasting, underweight, overweight/obesity between rural and urban children;
- measure weight, height, and skinfolds in children between the ages of six and nine years to compare the prevalence of risk factors associated with stunting, wasting, underweight, overweight, and obesity in children; and
- perform a repeat 24-hour recall dietary assessment (one week-day and one weekend-day) on 32% of the study population to determine correlations between dietary intake and body fatness.

5 Scope and limitations of the study

5.1 Scope

Anthropometric measurements were taken using the standardized techniques of the International Society for the Advancement of Kinanthropometry (ISAK) (Marfell-Jones *et al.*, 2006). The following anthropometric measurements were taken: weight and height, and the following skinfolds: biceps, triceps, subscapular, and iliac crest.

A 24-hour recall test was used to analyse micronutrient and macronutrient intake of a randomly selected subgroup of the subject population (32% of the children). Recommended energy and nutrient intakes vary according to gender and age. Information from the 24-hour recall test was used to calculate total energy intake as well as the percentage energy that was contributed by each of the macronutrient groups and was compared to the recommended daily allowances (RDAs) for each individual. Similarly, all macronutrient and micronutrient intakes were calculated for each individual and compared to RDAs for that age and gender group. The test was done according to the guidelines in Wolmarans *et al.* (2009) and was repeated twice to include one week-day and one weekend day. The results were coded and analysed using the FoodFinder software developed by the South African Medical Research Council (SAMRC) (Wolmarans *et al.*, 2009).

5.2 Limitations

This research project was subject to the following limitations:

- This research project was limited to the rural and urban primary schools of PLM, Capricorn District, Limpopo Province which were selected by the Senior District manager of the Capricorn District.
- The rural primary schools selected consisted of only black children, whereas the urban primary schools had white, coloured, Indian and black children.
- The study only included children from six to nine years of age.
- The fact that the study population was not randomly selected could have introduced bias to the study. The Department of Basic Education selected the schools that participated in the study and the first children in each age and gender group who brought back signed consent forms was selected to participate in the study.

- A 24-hour recall questionnaire tends to underestimate the amount of an individual's intake because of day to day variation; however, the selection of this method was justified based on a review of the existing literature, particularly on national studies, with consideration of the unique characteristics of the South African population, and on the basis that it is the best method to use on children (Oldewage-Theron *et al.*, 2011).
- The young children between the ages of six and nine found it difficult to accurately recall the food that had been consumed in the previous 24 hours. In this regard, the parents were interviewed telephonically to verify intakes if the children were unsure about what they ate.

5.3 Permission to conduct the study

An official letter for authorization to undertake the study at the nominated schools was submitted to the Senior Manager of Primary schools at the Department of Education, Capricorn District, and principals of selected schools. The letter entailed permission to take anthropometric measurements and assess the nutritional status of the children using the 24-hour recall test. The letter from the circuit manager (Addendum B), as well as the letter from the Limpopo Department of Education (Addendum C), both served as permission to allow the principals to let us conduct the study at their schools.

5.4 Harm to respondents

The researcher ensured that the participants were protected from any harm. Participants would have been referred to social workers and psycologists for professional intervention if it was found that harm had been inflicted due to the nature of questions posed.

5.5 Informed consent

The purpose of the study, the extent of their involvement, and what will happen to their private information was explained to the participants and their parens/guardians in the letter that accompanied the informed consent form. Thus both the participants and parents knew what the procedures involved before they gave consent to participate in the study. Parents of participating children gave informed consent on behalf of the child to participate in the study. Before data collection on any child commenced, the child was asked for ascent. The participants were also informed that the research was

5

voluntary and if any individual chose to withdraw from the research he/she could do so at any time. In case some questions to the participants might sound offensive and uncomfortable to respond to, they were informed not to respond.

5.6 Confidentiality and anonymity

Confidentiality was maintained regarding the information which was collected from participants. Each participant was given a subject number which was used to link the child's questionnaire and data sheet. The only people who had access to this information were the researcher and supervisors. The information was stored securely in the office of the Head of the Department of Physiology and Environmental Health.

5.7 Release or publication of the findings

The research results will be made available to the participating schools and Department of Basic Education through a report and publishable articles which allows the participants access to the results.

6 Structure of dissertation

The outline of the various chapters of this dissertation is as follows:

- Chapter two deals with the background and literature review relevant to the current research focus.
- Chapter three outlines the procedures that were followed, including the sampling method, study design, scope, and data collection processes.
- Chapter four deals specifically with the results of the study.
- Chapter five deals with comparing the results to other studies.
- Chapter six contains the conclusion and recommendations.

Chapter Two: Literature review

1 Overview of malnutrition

According to Christopher *et al.* (2008), 54% of child mortality worldwide is due to malnutrition. Malnutrition in children develops due to insufficient consumption of calories, proteins, iron (Fe), and other nutrients, as well as consumption of food high in saturated fats and sugars and reduced physical activity (Brabin & Goulter, 2003). Little information is available about the prevalence of malnutrition in children between the ages of 5-19 years. (World bank, 2015).

Malnutrition is broadly classified into undernutrition (which presents as stunting, wasting, and underweight) and overnutrition (which presents as overweight/obesity) (Tathiah *et al.*, 2013). Undernutrition is characterised by protein energy malnutrition (growth failure) and micronutrient deficiency, whereas overnutrition is characterised by overconsumption of macronutrients and energy dense foods (Faber & Wenhold, 2007). Wasting is defined as low BMI-for-age (below-2 z-score/SD) (WHO, 2010), and is an indicator of acute nutrient depreviation due to recent food shortage (famine) or illness (Corsi et al., 2015). Stunting is defined as low HFA; (below -2 Z-score/SD) (WHO, 2010) and is usually a linear growth delay due to chronic insufficient food and nutrient intake and frequent infections (Das & Bose 2011). Underweight is characterized by low weight-for-age (WFA) and is a result of short-term and long-term malnutrition (Faber & Wenhold, 2007). Overweight is a situation where body's fat stores have increased above normal limits, but have not reached a level classified as obesity yet. Thus it is an intermediate position between normal weight (lean) and obese (Ogden & Flegal, 2010). Obesity is defined by the WHO as the excessive accumulation of fat in the body, resulting in adverse health effects such as the development of metabolic diseases (Osuji et al., 2010). Overweight and obesity are two separate classes, but because of the small number of overweight subjects in this study, it was combined into one group. For this reason, it is treated as a single group throughout the study, including the literature review.

In many developing and even some developed countries, the prevalence of overweight co-exists with stunting in children (Lee *et al.*, 2012). This double burden of malnutrition

is advancing to a severe level, especially in developing countries such as SA, China, and Russia (Monyeki *et al.*, 2015).

Due to urbanisation, the diets of many South Africans are changing from a traditional African (prudent) diet to a Western diet (Jinabhai *et al.*, 2003). As a result of this nutritional transition in SA there is a shift in trends where undernutrition and overnutrition overlap, resulting in the co-existence of stunting and overweight in the population and even in the same individual (Atsu *et al.*, 2017). Therefore, a child can be both stunted and overweight at the same time (Jinabhai *et al.*, 2003).

2 Overnutrition

2.1 Overview of overnutrition

Obesity develops when the total energy intake is greater than the total energy expended by the body (Bray & Bellanger, 2006). The development of overweight/obesity occurs through interactions of various factors, including genetics, (Badawi *et al.*, 2013) environmental, cultural and behavioural factors (Doak *et al.*, 2006). In certain cultures, an abundance of food may result in social meetings which cause overeating, while foods rich in fats and energy are associated with prestige in other cultures (Kruger *et al.*, 2005). Genetic factors associated with overweight/obesity include deficiencies in leptin and growth hormone, which result in the rising prevalence of obesity. However, most cases of genetic predisposition only become visible or are expressed in the presence of behavioural and environmental factors associated with a positive energy balance (Badawi *et al.*, 2013).

Diseases associated with obesity have became a major public health concern in both developed and developing countries (Crowther & Norris, 2012). Manifestations of obesity include abdominal obesity (presented as an increased waist circumference), which is associated with the metabolic syndrome (Ntyintyane *et al.*, 2006). Lower body obesity is another form of obesity which tend to have an inverse relationship with cardiovascular diseases and type 2 diabetes mellitus (Karpe & Pinnick, 2015).

There are different terms, metrics, indices and cut-off values used to classify and calculate overweight/obesity (Flegal *et al.*, 2006). The body mass index (BMI), also

8

known as the "Quetelet index", uses the variables weight and height, and is calculated by dividing weight in kilograms (Kg) by height in metres squared (Kg/m²) (Chung, 2015). It is used to measure body fat stores, usually in adults rather than children, because the BMI of children changes as they grow older. BMI-for-age is useful for screening for overweight and obesity (WHO, 2008). Cole *et al.* (2000) determined agespecific BMI cut-off points for children between the ages of six and eighteen. Children are at high risk for overweight/obesity if their BMI is between the 85th and 95th percentile and higher than the 95th percentile for age and gender respectively (Cole *et al.*, 2000).

2.2 The prevalence of childhood overweight/obesity

2.2.1 The prevalence of childhood overweight/obesity among children globally Childhood overweight/obesity are pandemic worldwide, but are most prevalent in developed countries (Rahman et al., 2014). Guo et al. (2015) found that globally, more than 4 billion children were overweight. Majority of these children were living in developed countries (Guo et al., 2015). In Portugal, a study of children between the ages of seven and nine years found that 20.3% of the children were overweight and 11.3% were obese (Padez et al., 2004). The prevalence of overweight/obesity are increasing drastically in central and eastern Europe and the Middle East (Jackson-Leach & Lobstein, 2006). The prevalence of overweight/obesity amongst six to nineyear-old children was 5.6% and 6.3% respectively in a study conducted by Mosha and Fungo (2010) of Dodoma and Kinondoni Municipalities, Tanzania. Badawi et al. (2013) undertook a study of six to nine-year-old Port Said City primary school children and the results indicated that 17.7% of the children were overweight and 13.5% were obese. Studies on overweight/obesity in developing countries show that obesity increased from 4% in 1990 to 6% in 2010, and these percentages are expected to increase to 9% in 2020 (Gupta et al., 2013). Thus, overweight/obesity are now also increasing in underdeveloped and developing countries (Koirala et al., 2015), especially those undergoing nutritional and economic transition.

2.2.2 The prevalence of childhood overweight/obesity in South Africa

South Africa is a developing country, in a high transition state (Rocco *et al.*, 2014), and is experiencing a rapid increase in the prevalence of overweight/obesity (Griffiths *et al.*, 2013). The prevalence of overweight/obesity differs between ethnic groups,

genders and geographical areas (WHO, 2015). A study conducted by Jinabhai *et al.* (2001) amongst Kwazulu-Natal primary school children between the ages of eight and ten reported that 3.1% of the children were obese. Another study conducted in Kwazulu-Natal amongst children between the ages of eight to eleven, found that the prevalence of overweight varied between 0.4% and 13.3% and obesity between 0.1% and 3.7% according to WHO criteria (Jinabhai *et al.*, 2003). Overweight/obesity are now showing a dramatic increase in the low- and middle-income population in SA (An *et al.*, 2013). Studies also indicate that there is an increase in prevalence of obesity in the previously disadvantaged population (Guo *et al.*, 2015). If actions are not taken against the high prevalence of overweight/obesity in SA, the mortality rate due to obesity related conditions or the complications associated with obesity is likely to increase (Flegal *et al.*, 2012).

2.2.3 The prevalence of childhood overweight/obesity in urban and rural areas

In SA, high prevalences of overweight/obesity were found in rural communities of Limpopo, Kwazulu-Natal and Eastern Cape (Motadi *et al.*, 2015), but studies report that the prevalence of overweight/obesity is increasing most rapidly in urban areas due to the reduction of physical activity, consumption of energy dense Western diets containing refined carbohydrates, and meals full of refined oil (Duh *et al.*, 2016). In agreement with previous studies, Monyeki *et al.*, (2015) also found a higher prevalence of overweight/obesity in South African children residing in urban areas than in rural areas. The South African black population constitutes the largest percentage in the country and large numbers of people reside in rural areas (Stats SA, 2018). However, currently the percentage of urban blacks is increasing with more black Africans starting to reside in urban settings (Arndt *et al.*, 2018).

2.3 The risk factors associated with childhood overweight/obesity

Childhood overweight/obesity is complex and multifactorial. The major determinants of childhood overweight/obesity are urban residences, nutritional transition, higher socio-economic status, genetic predisposition, female children, and lack of physical activity (Han & Powell, 2013).

2.3.1 Age

The development of obesity is rapid during early life periods, childhood and adolescence (Dietz, 1994). Children and adolescents who become overweight or obese seldom lose the weight and rather tend to gain more weight as they grow older, thus becoming increasingly overweight and obese (Sutharsan *et al.*, 2015). Thus the prevalence of overweight/obesity in any group of children is likely to increase as the children grows older rather than to decrease.

2.3.2 Transition

The increased prevalence of overweight/obesity in developing countries is due to changes in lifestyle, particularly in diet and physical activity that is associated with urbanisation (Navaneetham & Dharmalingam, 2013). Childhood obesity has also been a problem over the past few decades because people in developing countries believed that increased body fatness improves chances of survival during famine (Parizkova & Hills, 2001) and thus overweight/obese children are viewed as being healthy and happy (Motadi *et al.*, 2015).

2.3.3 The socioeconomic and cultural factors

The socioeconomic status of a family is positively associated with overweight/obesity (Wang *et al.*, 2012). Children from families with high incomes have a high prevalence of overweight/obesity (He *et al.*, 2014). Socioeconomic factors that are associated with overweight/obesity include, amongst others, parents' level of education and occupation, household size and geographical region (Meko *et al.*, 2015). Mchiza *et al*, (2015) reported that there is higher BMI in high socio-economic areas than in low socio-economic areas of SA (Mchiza *et al*, 2015). A study conducted by Meko *et al*. (2015) in Bloemfontein, SA, found that children whose parents were degree holders were more overweight and obese than those whose parents did not have degrees.

In the black population of SA residing in rural areas, being overweight is reflected as a sign of being wealthy and happy, therefore people have a fear of being lean (Motadi *et al.*, 2015). However, black South Africans residing in urban areas believe that people who are not overweight are not at risk of developing CDLs, such as type 2 diabetes mellitus, cardiovascular diseases, and hypertension (Puoane *et al.*, 2010).

11

2.3.4 The genetic contribution of obesity

Genetic predisposition is one of the three main causes of childhood obesity, the other two being lack of exercise and unhealthy diet (Chamieh *et al.*, 2015). Monogenic obesity (obesity due to a specific genetic mutation) is very rare (Dubern *et al.*, 2007). Only a small percentage of childhood obesity is due to monogenic obesity (Sheikh *et al.*, 2017).

However, mutations on one or more of a variety of genes (Farooqi & O'Rahilly, 2005) can affect factors such as basal metabolic rate (BMR) (Luczynski *et al.*, 2016) or the mechanisms responsible for energy intake and expenditure (Nordang *et al.*, 2017). This will then affect the degree of body fatness or the relative ease with which one gains weight (Luczynski *et al.*, 2016). It is called polygenic obesity and is recognised as a CDL (Nordang *et al.*, 2017). Various genes have been linked with polygenic obesity, including: leptin, leptin receptor and pro-opiomelanocortin (POMC) genes (Cummings & Schwartz, 2003). These genes stimulate hunger and satiety signals, are involved in the growth and differentiation of adipose cells and control the amount of energy expenditure (Farooqi, 2011). The Y chromosome is involved in the development of obesity in some males and thus increases the chances of males to be obese (Perusse *et al.*, 2005).

2.3.5 The link between obesity and family history

Due to the genetic contribution to obesity, the condition has a trace of family history (Chamieh *et al.*, 2015). An overweight parent is at high risk of having a child with obesity (Bahreynian *et al.*, 2017), especially if the mother is obese (Reilly *et al.*, 2005). If one of the parents is obese, there is a high risk of the children to become obese, and if both the parents are obese, then the risk is two times higher (Lifshitz, 2008).

2.3.6 Intra-uterine environment and early life

The environment that a developing foetus is exposed to during pregnancy affects foetal development and this have an affect on the development of obesity after birth (Whitaker *et al.*, 1997). Such babies are often small at birth (Williams *et al.*, 2014). Birth weight of 2500 g or less for a full term baby, is regarded as the cut-off point for low birth weight (Hughes *et al.*, 2017). Certain factors associated with pregnancy, such as age of the mother, causes sub-optimal foetal development, which can increase the

chances for becoming obese later in life, especially when faced with an energy dense diet and high energy intake (Williams *et al.*, 2014). Aras, 2013 found that females with low body weight before pregnancy and those with low body weight during pregnancy gave birth to infants with low birth weight. These findings suggest that there is a significant correlation between maternal nutrition and low birth weight. A decrease in maternal nutrition affects the development of the foetus, resulting in low birth weight, which is a risk factor for obesity and type 2 diabetes mellitus (Galtier-Dereure *et al.*, 2000).

Gestational weight gain is the best predictor for early childhood obesity and overweight later in life (Oken & Gillman, 2003). Driul *et al.* (2008) found that females who gained weight before pregnancy and those who gained weight during pregnancy gave birth to overweight children. Mothers who are overweight are likely to give birth to big infants, who are then at high risk for the development of other health effects, such as obesity and type 2 diabetes mellitus (Li *et al.*, 2009). These diseases are all associated with increased body fatness (Pulos *et al.*, 2014).

Mothers who tend to over-feed their children may cause a positive energy balance in the child, resulting in the onset of obesity during infancy (Williams *et al.*, 2014). Weight gain and the method used to feed the infant are both predictors of developing obesity later in life (Baird *et al.*, 2005). Breastfed children tend to have normal weight compared to those who are not breastfed, because a breastfed child is likely to stop feeding when full, unlike a bottle-fed child who might be encouraged to continue feeding until the bottle is empty (Pinheiro *et al.*, 2015).

2.3.7 The effect of dietary energy intake on body fatness

Energy intake should equal energy expenditure to ensure a stable body weight (Müller & Geisler, 2017). If energy intake exceeds energy expenditure the individual will gain weight and is said to have a positive energy balance (Lillefosse *et al.*, 2014). If energy expenditure exceeds energy intake the individual will lose weight and is said to have a negative energy balance (Müller & Geisler, 2017). Many factors influence energy intake, but the diet is probably the single most important modifiable risk factor for execessive energy intake. In the human diet carbohydrates, proteins and lipids contribute to energy intake. Lipids conain more energy per gram than carbohydreates

and proteins, thus an increased intake of fat can easily result in a positive energy balance (You & Henneberg, 2016). The typical western diet followed in most developed and increasingly in developing countries is rich in refined carbohydrates, sugar, proteins (especially animal protein), and fat. The portion size also plays an important role in dietary intake, whereby the larger the portion size the higher the energy intake, which puts an individual at high risk of having a positive energy balance and developing overweight or obesity (Chen *et al.*, 2012).

In individuals with a positive energy balance, fat cells increase in number (Whitney & Rolfes, 2002) and this is a major global contributor to the childhood obesity epidemic (You & Henneberg, 2016). A high-calorie diet such as high carbohydrate intake influences the fat storage in adipose tissues (Lillefosse *et al.*, 2014).

2.3.8 The effects of physical activity on body weight

Physical activity is the body movements which causes muscle contraction and an increase in energy expenditure (Moore *et al.*, 2013). The high positive energy balance leads to the accumulation of calories as fat in the adipose tissues which lead to obesity (You & Henneberg, 2016). Even though there are many factors that affect energy expenditure, physical activity is probably the single most important modifiable factor that affects energy expenditure. Physical activity is associated with high benefits such as the reduction of weight gain and obesity (Wiklund, 2016).

Lack of exercise due to few sport activities and lack of awareness about the importance of physical activity at school and at home increases the development of obesity (Davies *et al.*, 1991). Urbanisation is associated with the decrease in the physical activity whereby children in urban areas have access to transport than in rural areas which make them having high positive energy balance (Amadou *et al.*, 2013). In urban environments, physical activities such as aerobics, skipping rope and riding bikes have been replaced by a sedentary lifestyle, characterised by a large amount of time being spent watching television, playing video games and being busy on cellular phones (Anrig, 2003). Anderson *et al.* (2008) reported that children between the ages of four to twelve years have a low physical activity due to living a sedentary lifestyle.

Physical activity is one of the major factors that lowers the risks of diabetes, high blood pressure, heart diseases, osteoporosis and certain related cancers (Luke and Cooper, 2013). Staying physically active increases the energy expenditure which helps in losing weight and maitaining a healthy body weight (Borodulin *et al.*, 2015).

2.3.9 The effect of leptin on the development of obesity

Leptin is a hormone that regulates food intake, energy expenditure and energy balance in humans through interactions with the hypothalamus (Crujeiras *et al.*, 2015). Leptin interacts with specific leptin receptors to inhibit excess feeding (Sainz *et al.*, 2015). Obesity is therefore often thought to indicate a leptin resistant state (Shapiro *et al.*, 2011). Leptin resistance in obese individuals is thought to arise due to a blunted response of leptin to caloric intake (Koch *et al.*, 2013).

2.4 The causal mechanism associated with increased body fatness

Obesity is a neuroendocrine dysfunction characterized by abnormal adiposity (Kwon & Pessin, 2013). It has been implicated in the aetiological mechanisms of metabolic disorders such as diabetes mellitus, insulin resistance, dyslipidaemia, hypertension and atherosclerosis, mainly due to hypersecretion of intermediary signalling mediators, such as adipocytokines, by the adipocytes (Finkelstein et al., 2012). Adipocytokines are extensively involved in hormonal communication, and are not only limited to fuel metabolism. Consequently, dysregulation of these molecules results in organ dysfunction of the cardiac, pulmonary, muscular, gastrointestinal, immune and reproductive systems, in addition to glucose and lipid metabolism (Refer to Figure 2.1). Furthermore, apart from the previously studied endocrine functions, adipocytokines, especially the thrombotic-promoting and pro-inflammatory type, are strongly associated with inflammatory response of the adipocytes (Kang et al., 2016). As the adipocytes become progressively subdued with nutrient reserves, tissue hyperplasia and hypertrophy initiates the self-induced inflammation of the adipocytes and increases levels of reactive oxygen species (ROS) (Li et al., 2009). Exaggerated local and systemic inflammation triggers a cascade of related events which lead to excessive production of antherogenic pro-inflammatory cytokines including tumor necrosis factor- α (TNF- α), interleukins (IL-6 and 1 β) (but not limited to) and chemokines (Li et al., 2009). These inflammatory modulators have deleterious effects on both fatty liver and pancreatic functioning that chronically cause the metabolic

pathologies that are central to insulin resistance (Aouadi *et al.*, 2013). Accummulation of adipocyte macrophages and adipokines synergistically contribute to the pathogenesis of various malignant tumours such as hepatocellular, oesophageal and colon cancers (Olesfsky & Glass, 2010). Therefore, insulin-induced obesity follows when the compensatory mechanisms from leptin and adiponectin, both of which regulate feeding and energy utilization, are surpassed by overproduction of adipokines (Yadav *et al.*, 2012).



Figure 2.1: The causal mechanism associated with increased body fatness. (Constructed based on Kwon & Pessin, 2013, Finkelsten et al., 2012,

Li et al., 2009, Aoudi et al., 2013, Olesfsky & Glass, 2010 and Yadav et al., 2012)

2.5 The complications associated with overweight/obesity

The Word Health Organization (WHO) classified childhood obesity as a major public health problem in both developed and developing countries globally (Anrig, 2003). Obesity is one of the leading causes of mortality and morbidity (Musaiger, 2011). Childhood obesity causes physical and psychological complications.

Physical complications include diabetes mellitus, ischaemic heart disease, and some types of cancer. Over 3.4 million deaths occur annually, of which overweight/obesity are the fifth major cause of mortality (Pulos *et al.*, 2014). It is estimated that in 2020 a quarter of the deaths will be caused by CDLs associated with overweight/obesity (WHO, 2015). It is reported that globally 44% of deaths are caused by diabetes, 23% caused by ischaemic heart diseases and 41% caused by related cancers; these diseases are all associated with increased body fatness (Pulos *et al.*, 2014).

Psychological complications include low quality of life, depression, and low selfesteem (Pulos et al., 2014). Childhood is a stage where an individual is going through a lot of physical, behavioural and psychological changes which lead to the preparation of adulthood. In rural areas, obese individuals are often regarded as being affluent. On the other hand, many individuals, including children, regard being obese is as being less attractive (Zametkin et al., 2004). As a result, obese children tend to isolate themselves from their peers (Mc Cullough et al., 2009). One out of every three obese children is antisocial and has few or no friends at all (Zeller et al., 2008). Bosch et al. (2004) indicated that some children call obese people inappropriate names such as lazy and ugly. Children as young as five years old engage in such stereotypical behaviour (Brylinsky et al., 1994). These negative remarks can affect a child's selfesteem and confidence (Cornette, 2008). Within societies, it appears that only females suffer from image problems due to increased body fatness, as compared to males (O'Dea, 2005). However, limited research on males and body dissatisfaction suggests that males also experience body dissatisfaction, and that this trend is on the rise (Mitch, 2014).

3 Undernutrition

3.1 Overview of stunting

3.1.1 The prevalence of stunting globally

The global prevalence of stunting has been declining in the past decade (Black *et al*, 2013). Globally, one in four children is stunted (de Onis *et al*., 2012). However, there seems to be an increase in the prevalence of stunting in Africa due to a population increase in the past decade (Black *et al*., 2013). It has been estimated that with the rise in population outcomes (increased births) the prevalence of stunting in Africa will rise from 56 to 61 million (Black *et al*., 2013). Africa is still encountering challenges in reducing stunting by 40% in 2025; the reductions have been lagging behind, with reductions predicting 20% in 2025 with the current trend (Black *et al*., 2013). Factors that may contribute to the development of stunting in developing countries include maternal education, accessibility to clean household water and sanitation facilities, and household income (Willey *et al*., 2009). Sub-Saharan Africa is one of the regions with the highest prevalence of stunting at 40%. Approximately 42% of children in sub–Saharan Africa are stunted (Kravdal & Kodzi, 2011). Malawi has the highest prevalence rates, ranging from 41–48% (Huey & Mehta, 2016).

3.1.2 The prevalence of stunting in South Africa

Stunting is a common disorder affecting children in SA (Berry *et al.*, 2010). The prevalence of stunting in SA is currently at an estimated 20% (Kimani-Murage, 2010), and it is constantly increasing (Jinabhai *et al.*, 2003). However, there are disparities in the prevalence of stunting in provinces, with rural Limpopo (34%) reported as the highest in the country and urbanised Gauteng (12%) with the lowest rate (Kleynhans *et al.*, 2006).

3.1.3 The prevalence of stunting in urban and rural areas

Stunting is consistently higher in rural than urban areas (Jiang *et al.*, 2015). Patil *et al.* (2016) reported that the prevalence of stunting in rural and urban India was 37% and 22% respectively. A study conducted in eastern India found that the prevalence of stunting was higher in boys (56%) compared to girls (47%), preterm born boys and those with low birthweight are more prone to be stunted than girls (Krishna *et al.*,

2016). The study revealed that high birth order and educational status were independently associated with stunting (Biswas & Bose, 2010); thus it is possible that these factors are the mediators of stunting in rural areas. Low income level, poor socioeconomic status and low education levels are associated with stunting (Biswas & Bose, 2010). A study conducted by Herrador *et al.* (2014) in rural and urban Ethopia found the prevalence of stunting to be 42.7% and 29.2% respectively. The THUSA BANA Study in the North-West province of SA reported that stunting is more prevalent in the rural areas (girls 23.7% and boys 26.7%) as compared with the urban areas (girls 11.6% and boys 17.1%) (Mukuddem-Petersen & Kruger, 2004).

3.2 Overview of wasting

3.2.1 The prevalence of wasting globally

The highest prevalence of children with severe wasting is found in central Africa and south-central Asia (Black *et al.*, 2008). Globally in 2013, 8% of children suffered from moderate wasting and 3% of children were severely wasted, with the highest prevalence rates in Africa (28%) and Asia (71%).

3.2.2 The prevalence of wasting in South Africa

The prevalence of wasting in the South African provinces is high in the Northern Cape (19%), followed by the Western Cape (12%) (Berry *et al.*, 2010). This might be because these provinces have large numbers of people living under poor socioeconomic conditions, with increased risk of food insecurity. In Limpopo's Vhembe district, the prevalence of wasting is 0.4% in children (Motadi *et al.*, 2015). There is not enough information about the prevalence of wasting in children in the Limpopo Province.

3.2.3 The prevalence of wasting in urban and rural areas

The prevalence of wasting is reported to be common in the rural areas (Berry *et al.*, 2010). A study conducted by Herrador *et al.* (2014) reported a higher prevalence of wasting in rural area than urban area settings of Fogera and Libo Kemkem districts, Ethiopia.

3.3 Overview of underweight

3.3.1 The prevalence of underweight globally

The prevalence of underweight in Indonesian school aged children was reported to be 14.5% (Syahrul *et al.*, 2016). A study conducted by Hasan *et al.* (2013) in higher primary schools in Azad Nagar and its surrounding area, Bangalore (Bengaluru), found that the prevalence of underweight amongst primary school children was 58.3%.

3.3.2 The prevalence of underweight in South Africa

Generally, the prevalence of underweight in SA is low (Berry *et al.*, 2010), but reported prevalences vary between areas and studies. The reported prevalences may also be affected by the lack of studies in some areas. The prevalence of underweight in children between the ages of one and nine years was 9.3% (Berry *et al.*, 2010) in 2005. A study conducted by Puckree *et al.* (2011) in Kwazulu-Natal amongst primary school children reported that the prevalence of underweight was 66%. A study conducted by Moselakgomo *et al.* (2015) in the Limpopo and Mpumalanga provinces of SA amongst primary school children reported that the prevalence of underweight in Limpopo's Vhembe district is 0.3% in children (Motadi *et al.*, 2015). This low prevalence of underweight in Vhembe district might be because of following a prudent diet due to agricultural practices which alleviates poverty.

3.3.3 The prevalence of underweight in urban and rural areas

The prevalence of underweight is common in rural areas (Herrador *et al.*, 2014). A study conducted by Meshram *et al.* (2016) reported that the prevalence of underweight in urban and rural areas were 32% and 46% respectively. Monyeki *et al.* (2015) reported that the prevalence of underweight in rural and urban areas was 0.7%–66% and 3.1%–32.4% respectively.

3.4 The risk factors of undernutrition in children

3.4.1 The socio economic factors on undernutrition in children

Socio-economic factors associated with undernutrition include amongst others residing in an over-populated area, spending less money on food, and sharing a one roomed apartment (Kleynhans *et al.*, 2006). Traditionally in Africa, child welfare is the responsibility of the mother; therefore, factors such as maternal employment may have
both negative and positive nutrition outcomes in a child (Mesfin *et al.*, 2015). In SA, Limpopo is one of the provinces with the highest rates of poverty and unemployment (STATSSA, 2016). Low socio-economic class is associated with undernutrition because of the few resources available to this society (Kleynhans *et al.*, 2006).

3.4.2 The link between undernutrition and family history

Hereditary factotors play an important role in the predisposition of being underweight; however, environment in terms of physical activity, energy intake and socio-economic status also plays a role in family history as a risk factor for undernutrition (Kleynhans *et al.*, 2006). A study conducted by (Fernald & Neufeld, 2007) found that European children with short mothers, were stunted. It is possible that the mothers themselves were stunted and the children may then also be stunted, both because of poor socio-economic conditions.

3.4.3 Gestation and maternal health

Van Stuijvenberg *et al.* (2005) suggested that factors associated with gestation and maternal nutrition are implicated in the prevalence of stunting. Gestation is one of the important processes that determine health and development outcomes of the child later in life. Nutritional challenges during pregnancy and failure to gain maternal weight may have long-term effects on child development (Popkin *et al.*, 1996). Poor maternal nutrition results in inadequate foetal nutrition and thus in sub-optimal foetal development (Darton-Hill & Mkparu, 2015). Age and the nutritional status of the mother are associated with childhood undernutrition (Asafaw, 2015). Shrivastava and Shrivastava (2013) found that the prevalence of Indian children born with low birth weight was 45.6% among children born of teenage mothers. Factors such as short maternal stature, low pre-pregnancy weight, poor gestational nutrition outcomes and infections indirectly lead to higher risk of stunting (Krishna *et al.*, 2016). A study conducted by Fernald and Neufeld (2007) found that European children with short mothers, were stunted.

3.4.4 Birthweight

There are four major stages of growth in children: (i) growth during the third trimester of pregnancy, (ii) during early infancy, (iii) between the ages of five to seven years (adipose rebound) and (iv) during adolescence (Popkin *et al.*, 1996). The markers of

stunting are reflected within the first 1000 days of life. Low birth weight is one of the markers associated with poor foetal conditions (Krishna *et al.*, 2016). A study conducted by Kleynhans *et al.* (2006) in poor areas of rural Limpopo and urban Gauteng found that there was a significant correlation between low birth weight and the child's current HFA.

3.4.5 Breastfeeding

Breast milk is a bio-active liquid with a multifaceted composition that varies within and between feedings to suit the unique nutritional and immunological requirements of an infant. It consists of macronutrients and micronutrients, with fat being the most varying nutrient; however, it provides half of the energy content in the milk (Lutter & Morrow, 2013). The hind milk is richer in fat as compared to the foremilk, and in essence the feeding process can occur eight to twelve times or more per day (Lutter & Morrow, 2013). Breast milk also constitutes bio-active substances such as immunoglobulins, hormones, growth factors and enzymes (Guo & Ahmad, 2014).

The nutritional requirements of an adult diet necessitate that it be composed of 55–60% carbohydrates; 10% proteins and 30% fats, varied according to polyunsaturated and saturated fats (Franz, 2003). The nutritional requirements of infants during the first six months ranges between 95 and 115 kcal/kg/day, with 40–60%, 8–12% and 50% derived from carbohydrates, proteins and fats respectively (Guo & Ahmad, 2014). In 100 ml of breast milk there is approximately 7 g of carbohydrates (with lactose being the main source), 0.9 g proteins, and 3.5 g fat. With the limited amount of protein in breast milk, it can easily be decreased during the process of pasteurisation, and therefore caution should be taken when giving infants donated breast milk to ensure that it is tailor-made for every infant's needs (Lutter & Morrow, 2013). On average an infant will gain approximately 25–40 g per day in the first three months and 15–20 g per day in the second three months (Guo & Ahmad, 2014).

Exclusive breastfeeding involves the feeding of an infant with only breast milk or expressed breast milk, with the exception of drops or syrups consisting of vitamins, mineral supplements, or medications, and therefore no other liquids or solids are given (Guo & Ahmad, 2014). It is recommended that with infants who cannot be exclusively breastfed by their own mothers, the next best alternative will be to receive donated

pasteurized breastmilk (WHO, 2003). A study conducted in Limpopo's Vhembe district reported that breast milk decreases the risk of infectious diseases such as bacterial meningitis and diarrhoea, as well as stunted growth (Kyei *et al*, 2014). Children who are breastfed for more than one month have a lower risk of becoming overweight (Harder *et al.*, 2005).

3.4.6 The effect of diet on undernutrition diet

Minerals are inorganic substances that are found in the food that the body needs for growth and health (van Stuijvenberg et al., 2015). Minerals are found in foods such as meat, cerelas, fish, milk and diary foods, vegetables, fruit and nuts (Motadi et al., 2015). Protein energy malnutrition is caused by inadequate intake of energy and protein, resulting in body muscles wasting (Voortman et al., 2017). Deficiencies in mineral nutrients causes diseases (Rooze et al., 2016). Deficiencies in Fe, iodine (I), Ca, vitamin A (Vit A), vitamin D (Vit D) and Zinc (Zn) are the most common micronutrient deficiencies in developing countries (Rooze et al., 2016). A study by van Stuijvenberg et al. (2015) focused on the dietary intake of South African children using a 24-hour recall test and found that deficiencies in Ca and Vit D were significantly associated with stunting. Since Ca is one of the main mineral components of bone tissue, it is essential for adequate bone formation, and considering that Vit D plays an important role in Ca metabolism, a diet with insufficient quantities of these nutrients can impact on the formation of the skeleton and on the process of growth and development (Rooze et al., 2016). Another study conducted in the Limpopo Province among pre-school children aged three to five years found that the Fe and Zn deficiencies were associated with poor growth and increased susceptibility to various infections (Motadi et al., 2015). Zn deficiency is prevalent in low income countries causing growth restriction, diarrhea, and other health issues (Shekhar, 2013). Stunting prevalence correlates with high risk of zinc insufficiency in young children leading to pneumonia, diarrhea and malaria (Shekhar, 2013). The prevalence of anemia is high in 18 months old children and drops as Fe requirements decreases (Black et al, 2008).

3.4.7 Infections

Poor socio-economic and environmentalfactors increase the risk of predisposition of children to infections and increase the risk for the development of stunting (Biswas & Bose, 2010). Complications associated with undernutrition include increased risk of

mortality from infectious diseases such as diarrhoea, pneumonia, and measles (Black *et al.*, 2013). South Africa has been listed among eight countries with a history of high occurrence of diarrhoea and malnutrition (Dillingham & Guerrant, 2004). Diarrhoea is a consequence of malnutrition (Prendergast & Humphreys, 2014). Infections increase the susceptibility of children to malnutrition, with Fe deficiency identified as a confounding factor associated with infection (Oliveira *et al.*, 2015). There are two common pathogens involved in the pathogenesis of diarrhoea, *Cryptosporidium parvum* and *Giardia lamblia*, which affect the growth and nutritional status of children (Berkman *et al.*, 2002). The high burden of diarrhoea and enteric infections correlates with impaired growth in children (Dillingham & Guerrant, 2004).

3.5 Causal mechanisms associated with undernutrition (stunting, wasting and underweight)

Causal pathways associated with stunting, wasting and underweight are still not well understood (Huey & Mehta, 2016). However, factors such as socio-economic and environmental factors, maternal, foetal and early nutrition, as well as physiological changes might lead to the development of undernutrion (Biswas & Bose, 2010). Improved socio-economic status of the mother, such as employment, might cause disruptions in the process of exclusive breastfeeding (refer to **Figure 2.2**). This will increase the risk of predisposition of children to infections and increase the risk for the development of stunting (Biswas & Bose, 2010).

When the body is invaded with infections, non-specific chronic activation of the immune system becomes activated. The immune system will respond by causing inflammation. The gradual inflammation will result in the accumulation of ROS, which will overwhelm the free radical scavenger activity of the body. This will be signified by deficiency such as Vit A, Vitamin E (Vit E), and essential minerals such as copper (Cu), Zn and some of the important fats such as glutatathione carotene and polyunsaturated fatty acid, and lead to the development of wasting (Aly *et al.*, 2014).

Improved socio-economic status can further increase livelihood of the mother due to increased income and this may positively influence availability of food and food quality (NHANES, 2010). This will increase consumption of macronutrients and micronutrients. A high consumption of macronutrients and decreased physical activity

25

levels will result in excess energy and increased adiposity, which causes the development of overweight/obesity (Chen *et al.*, 2012).

The relationship between stunting and overweight is due to individual adaptability wherein stunted children did not have sustained better living conditions (Said-Mohamed *et al.*, 2015). With emergence of the nutritional transition, it became apparent that children do not have control mechanisms towards adapting to the rapid change from a prudent to a Western diet; therefore, increased body fatness develops (Popkin *et al.*, 1996).

Poor socio-economic status affects foetal and maternal nutrition (Darton-Hill & Mkparu, 2015). Poor feeding practices affect foetal development and result in lack of dietary control mechanisms. This will affect the sub-normal intake of macronutrients and micronutrients which will result in undernutrition or overnutrition (Emery, 2005).

During the gestation stage, poor feeding practices cause maternal and foetal hormonal changes and physiological responses (Darton-Hill & Mkparu, 2015). Breastfed children tend to have normal weight compared to those who are not breastfed, because a breastfed child is likely to stop feeding when full, unlike a bottle-fed child who might be encouraged to continue feeding until the bottle is empty (Pinheiro *et al.*, 2015). Foetal hypothalamic programming alters appetite as a result of early chronic malnutrition. Chronic malnutrition results in slow growth, a shift in the BMR and changes in the gastrointestinal hormones such as ghrelin, Cholecystokinin (CCK) and peptide YY (PYY₃₋₃₆) (Emery, 2005). Slow growth, changes in BMR and changes in gastrointestinal tract (GIT) hormones cause a decreased intake of macronutrient and micronutrients, which results in the development of underweight (Emery, 2005). Inverserly, an increase or decrease in macronutrient and micronutrient intake causes changes in BMR (Emery, 2005).

26





KEY: BMR = Basal metabolic rate, CCK = Cholecystokinin, PUFA = Polyunsaturated fatty acids, PYY₃₋ $_{36}$ = Peptide YY, ROS = Reactive oxygen species, Cu = Copper, Se = Selenium, Zn = Zinc

3.6 Complications associated with undernutrition

Complications associated with stunting, wasting and underweight include increased risk of mortality from infectious diseases such as diarrhoea, pneumonia, and measles (Black *et al.*, 2013). Optimal growth and development in an infant is achieved between zero and six months through adequate dietary intake (Prendergast & Humphreys, 2014). Undernutrition is associated with increased likelihood of slow rate of developmental milestones and increases the risk of being overweight later in life (Bove

et al., 2012). Complications associated with stunting before two to three years include poor cognitive and educational outcomes later in childhood and adolescence (Black *et al.*, 2013). Stunting has been reported to be associated with poor education outcomes, especially in Africa and Asia, where it was reported that stunted children enrolled in school later than children of the same age (Murungi, 2012).

4 Summary

Malnutrition is an important public health concern in developing countries such as SA and is increasing alarmingly. The prevalence of malnutrition conditions (such as stunting, wasting, underweight, overweight/obesity) is increasing with urbanisation and modernisation. South African children are under a nutritional siege that stretches from stunting to obesity. However, the effect of urbanisation and modernisation on the development of these conditions and the co-occurance of stunting and overweight in South-African children, especially those residing in relatively poorer and rural provinces are not known.

Chapter Three: Research methodology

1 Introduction

Chapter Three outlines the methodology used to select the subjects, collect data and analyse the data for this cross-sectional study. The cut-off points for the different categories of malnutrition used during the analysis of data are also included.

2 Study area

This study included children from four primary schools in rural areas and four primary schools in urban areas of the PLM, Limpopo Province, SA. The selection of schools was done by the Limpopo Department of Basic Education and was based on the geographical areas in which the schools were located.

3 Study design

This cross-sectional study was conducted over a period of five months starting from May 2016. Anthropometric measurements and the questionnaire were only conducted once, but the 24-hour recall dietary assessment was conducted twice to include a week day and a weekend day.

4 Exclusion and inclusion criteria

The children with the following conditions were not included in the study population of the research:

- Children with cardiovascular conditions
- Children with respiratory complications such as asthma
- Children taking medications
- Children with some disabilities
- Children with injuries

Only children that met the following inclusion criteria were included in the study population of the research:

• Male and female children between the ages of six and nine years

- Children who brought back signed informed consent forms and completed questionnaires
- Children who gave ascent to participate in the study

5 Sample size and technique

The sample size for this study was calculated with the help of a qualified bio-statistician using Slovin's formula: $n = N / 1 + N(e)^2$ (Johnson & Christensen, 2012), where n is the sample size, N is the population size, and e is the level of precision (0.05). The population size of the primary school children was 433 and the sample size is 208, therefore $n = 208 / 1 + 208(0.05)^2$. Sampling was done by the researcher at the respective schools during school hours. A total of 208 randomly selected subjects were expected in this study, as described in **Table 3.1**. Four rural schools and four urban schools were included in the study. However, at one urban school the children between the ages of six to nine years did not bring back signed consent forms, therefore one urban school was excluded from the study. Furthermore, of all the seven schools that participated, only 174 consent forms came back signed by parents. Only children with signed consent forms were included in this study.

Urban schools			Rural schools				Total	
Gender	Age	Per school	Total: 4 schools	Gender	Age	Per school	Total: 4 schools	8 schools
Male	6-7	6	24	Male	6-7	6	24	48
	8-9	7	28		8-9	7	28	56
Female	6-7	6	24	Female	6-7	6	24	48
	8-9	7	28		8-9	7	28	56
Total		26	104			26	104	208

Table 3.1: Composition of the intended study population

One urban school did not participate in the study.

6 Study population

Primary school children between the ages of six and nine were included in the study. Every effort was made to include an equal number of boys and girls across the study population, but it was not always possible. This is due to the fact that grade one learners include six and seven-year-old children, while puberty tends to start as early as the age of 10 years, therefore, children older than nine years were excluded to eliminate the effect of puberty-related hormone changes on the body composition of the children (Solorzano & Mc Cartney, 2010). Every effort was made to include 104 urban and 104 rural children in the study, but it was not possible. Only children with signed informed consent forms were allowed to participate in the study.

7 Research methodology

A cross-sectional study was conducted over a period of five months starting in May 2016. Anthropometric measurements and the questionnaire were only conducted once, but the 24-hour recall dietary assessment was conducted twice to include a week day and a weekend day. The sample size for this study was calculated with the help of a qualified bio-statistician using Slovin's formula: $n = N / 1 + N(e)^2$ (Johnson & Christensen, 2012), where n is the sample size, N is the population size, and e is the level of precision (0.05). The population size of the primary school children was 433 and the sample size is 208, therefore $n = 208 / 1 + 208(0.05)^2$. A total of 174 randomly selected subjects were included in this study. Sampling was done by the researcher at the respective schools during school hours. Weight and height measurements were used to calculate BMI, weight-for-height (WFH) and height-for-age (HFA) ratios (Mei et al., 2002). Skinfolds were used to calculate the body fat percentage (BF%) (Mei et al., 2002). The World Health Organisation's (WHO) 2007 cut-off points for BMI-for-age were used to classify wasting (thinness), overweight/obesity of the children. Criteria suggested by WHO (2007) were used to classify stunting and underweight. Z-scores were used to calculate undernutrition. All statistical analyses were performed using IBM SPSS Statistics version 23. Frequency distributions were calculated and checked for normal distribution. Any obviously incorrect outliers were checked against the guestionnaires and data record sheets for correctness. Descriptive statistics were performed for all variables. Student's T-tests were used to compare rural and urban populations. Regression analyses were used to determine relationships between different variables, body fatness and nutritional status. Pearson correlatons were done to assess the relationship between BF% and nutrients, and a p-value of less than 0.05 was considered statistically significant.

8 Data collection

8.1 Organisational procedure

A formal letter for permission to conduct the study and a consent form was sent to the principals of the selected schools to give permission for the study to be conducted at the school (Addendum D). An appointment was made with the school to explain the study to the learners. Information letters and consent forms were sent to the parents to sign (Addendum E). The researcher distributed more consent forms than the number of participants wanted, to compensate for forms that may be lost or come back without the parent's signature. The researcher visited the school on a second occasion to collect the consent forms, conduct the first 24-hour recall assessment and take the anthropometric measurements. Each participating child received a questionnaire (Addendum F) to take home for the parents to complete as well as a feedback chart (Addendum G) with his/her results and their meaning. The researcher went to the school on a third occasion to collect the questionnaires and conduct the second 24-hour recall assessment. On some occasions, the researcher had to telephonically contact the primary care-giver of the child to clear up any uncertainties with regard to the dietary intake.

8.2 Questionnaire

Each participating child received a questionnaire (Addendum F) to take home with a request for the parents to complete. Parents were allowed to leave some questions unanswered, if they did not feel comfortable answering all the questions, especially those concerning family income and health status of the parents. The questionnaire also contained the contact details of the parents/guardians so that they could be contacted to verify the dietary intakes reported by the children. The questionnaire has been translated from English into Afrikaans and Sepedi (the native languages of most of the children in the geographic area) and backtranslated into English to ensure that the meaning of the questions remained the same throughout the translation process.

The questionnaire consists of the following sections:

- Section A: Demographic information
- Section B: Medical history
- Section C: Dietary information

• Section D: Physical activity information

8.3 Measuring malnutrition in children

8.3.1 Criteria for determining malnutrition in children

There are several anthropometric methods that can be used to measure malnutrition in children. In cases of a nutritional emergency, nutritional status is best measured with the following variables: gender, age, height, and weight. These variables are then used to calculate the different anthropometric indices, such as BMI-for-age, heightfor-age (HFA), weight-for-height (WFH) and weight-for-age (WFA). In this study HFA, WFH, WFA and BMI for age were used to classify malnutrition. These methods are mostly used in literature because they are accurate and cost effective. Deficits in HFA are a sign of stunting and is usually a linear growth delay due to chronic insufficient food and nutrient intake and frequent infections (Das & Bose 2011); WFH index helps to identify children suffering from wasting and is an indicator of acute nutrient depreviation due to recent food shortage (famine) or illness (Corsi et al., 2015); and WFA index helps to identify the condition of being underweight for a specific age and is a result of short-term and long-term malnutrition (Faber & Wenhold, 2007). The index BMI for age is used to classify underweight, lean, overweight/obesity (Chung, 2015). Skinfolds are also used to calculate BF%, which can be used to classify overweight or obesity in children. Obesity is the excessive accumulation of fat in the body, resulting in adverse health effects such as the development of metabolic diseases (Osuji et al., 2010). The World Health Organisation (WHO) reference standards (WHO, 2007) were used to classify malnutrition in this study population. These criteria are summarised in **Table 3.2**.

Table 3.2: The criteria used to classify malnutrition in children aged six to nine years(WHO, 2007)

BMI for age	SD-Z scores
Severely thin and wasted	<-3 SD
Thin and wasted	≥-3 SD and < -2 SD
Normal weight	≥-2 SD and <+1 SD
Risk of overweight	≥+1 SD and <+2 SD
Overweight	+2 SD and < +3 SD
Obese	+3 SD
Height-for-age	SD-Z scores
Severely stunted	<-3 SD
Stunted	≥-3 SD and <-2 SD
Weight-for-age	SD-Z scores
Severely underweight	<-3 SD
Underweight	≥-3 SD and <-2 SD

8.3.2 Criteria for determining the recommended daily allowance of macronutrients and micronutrients in children.

The dietary reference intakes: recommended dietary allowance and adequates of micronutrients and macronutrients were used to classify malnutrition in this study population. These criteria are summarised in **Table 3.3**.

 Table 3.3:
 The criteria used for determining recommended dietary allowance of micronutrients and macronutrients requirements for

Group	Vit A (µg/d)	Vit D (µg/d)	Ca (mg/d)	Fe (mg/d)	Zn (mg/d)	P (mg/d)	Protein (mg/d)	Carbohydrates (g/d)	Energy (kcal/d)
Children 4–8 years	400	15	1000	10	5	500	19	130	
Males 3–8 years									1.742
Females 3–8 years									1.642
Males 9–13 years	600	15	1300	8	8	1250	34	130	2.279
Females 9–13 years	600	15	1300	8	8	1250	34	130	2.071

children between 4-13 years of age according to http://www.nap.edu/openbook.php?isbn=0309085373

Vit A = Vitamin A; Vit D = Vitamin D; Ca = Calcium; Fe = Iron; Zn = Zinc; P = Phosphorus;

8.3.3 Anthropometric measurements

Anthropometric measurements were taken by a trained staff using the standardized techniques of the ISAK (Marfell-Jones *et al.*, 2006). The researcher in this study was trained to do anthropometric measurements. The following anthropometric measurements were taken: weight (kg) and height (cm), and the following skinfolds: triceps, subscapular, abdominal and suprailliac crest (all in mm). Weight was measured using an electronic scale, with subjects wearing minimal clothing. Height was taken using a portable stadiometer, with the head of the subject positioned in the Frankfort plane with the body standing upright. The children presented in minimal clothing for the skinfolds measurements. Prior to measuring the skin folds, landmarks were identified and marked, using a Lufkin Executive Thin line steel tape to measure the midpoints. A calibrated HaB Direct skinfold calliper was used to measure the skinfolds.

8.3.3.1 Anthropometric and nutritional status calculations

Body mass index was calculated by dividing mass (kg) into and height (m) squared (kg/m²) and then used to calculate BMI-for-age, an indication of overweight/obesity (Chung, 2015). WFA index was calculated for each child to classify children as being underweight, overweight and obese for a specific age. HFA was calculated for each child to determine the prevalence of stunting. WFH was also calculated for each child as an indication of wasting. The equation of Brook (1971) was used to calculate and assess BF% from skinfold thickness, using the four skinfolds measured in this study. These two Brook, (1971) equations were used to calculate density for males and females.

F: D = 1.2063–0.0999 (LOG sum of 4 skinfolds)

M: D = 1.1690–0.0788 (LOG sum of 4 skinfolds).

The following Lohman *et al* (1984) equation was used to calculate BF% for both males and females, the children were classified as as overweight or obese, M and F: Fat (%) = 530/D - 489.

8.4 Measuring dietary intake in children

There are different methods of assessing diet, such as diet records, diet history, weekly consumption of foods, 24-hour recall test, and food frequency questionnaires (Shim *et al.*, 2014). It is extremely difficult to get accurate dietary information from

children (Pérez-Rodrigo *et al.*, 2015). Children tend to under- or over-report their nutrient intakes, the under-reporting of food intakes appears to be more common than over-reporting (Raina, 2013). However, parents can be contacted to verify food consumption.

The 24-hour recall test was used in this study in an effort to capture exactly what the children consumed immediately prior to a study visit, because it is cost effective and time efficient (Shim *et al.*, 2014). Advantages of the 24-hour recall test are that it is quantifiable and does not alter eating behaviour (Shim *et al.*, 2014). Compared to the food frequency questionnaire, the 24-hour recall test only requires children to recall a shorter time (one day) and it is therefore more accurate than the food frequency questionnaire (Shim *et al.*, 2014). Accuracy can be further improved by contacting parents and/or caregivers to verify if the food reported by the child is correct. The disadvantages of the 24-hour recall test are that it relies on subject recall, requires multiple recalls, and it is a time consuming task to capture the data (Raina, 2013).

8.5 24-hour recall test

A 24-hour recall test was used to analyse the intake of macronutrients and micronutrients (minerals, vitamins and trace elements) in an individual. It requires subjects to report all foods and beverages consumed in the past 24 hours. This can be done via telephone or face-to-face interview. The test was done according to the guidelines in Wolmarans *et al.*, (2009). It is best to repeat the 24-hour recall test at least twice to include one week-day and one week-end day. This is what was done in this study, each child was interviewed twice to report on a week day and a week-end day. Trained staff must conduct the interview to prompt for details, such as cooking methods and portion sizes. This researcher was trained to do a 24-hour recall. The data from the 24-hour recall test is usually coded and analysed using software such as the FoodFinder developed by the South Africa Medical Research Council (SAMRC) (Wolmarans *et al.*, 2009).

Recommended energy and nutrient intakes vary according to gender and age. Information from the 24-hour recall test was used to calculate total energy intake as well as the percentage energy that was contributed by some of the macronutrient groups and was compared to the RDAs for each individual. Similarly, all macronutrient and micronutrient intakes were calculated for each individual and compared to RDAs for that age and gender group. The test was done according to the guidelines in Wolmarans *et al.*, (2009) and the results from both the week day and week-end day were coded and analysed using the FoodFinder software developed by the SAMRC (Wolmarans *et al.*, 2009).

9 Statistical analysis

All measurements have been expressed statistically using IBM SPSS Statistics version 23. The distribution of data was tested for normality using histograms and line graphs. Descriptive statistics were performed for the analyses of demographic and health information. Frequencies and percentages were used to interpret demographic and health information of the children, and to indicate the prevalence of malnutrition. The Student's T-test was used to compare rural and urban populations, and chi-square tests were performed and frequencies and percentages were presented for the comparison of the prevalence of stunting, wasting, underweight, overweight, and obesity between rural and urban children. A *p*-value of less than 0.05 was considered statistically significant.

To assess the prevalence of risk factors for the development of stunting, wasting, underweight, overweight/obesity in children, descriptive statistics was presented as means and standard deviation (\pm SD) and frequencies and percentages. Chi-square test was performed and a *p*-value of less than 0.05 was considered statistically significant.

Pearson correlatons were done to assess the relationship between BF% and nutrients, and a *p*-value of less than 0.05 was considered statistically significant.

10 Ethical considerations

This study was ethically approved by the University of Limpopo Ethics committee, project number (TREC/103/2017: PG) (Addendum A). Permission to collect data at the selected schools was granted by the Limpopo Department of Education and the Mankweng circuit offices respectively. The purpose of the study, the extent of their involvement, and what will happen to their private information was explained to the

participants and their parens/guardians in the letter that accompanied the informed consent form. Thus both the participants and parents knew what the procedures involved before they gave consent to participate in the study. Parents of participating children gave informed consent on behalf of the child to participate in the study. Before data collection on any child commenced, the child was asked for ascent. The participants were also informed that the research was voluntary and if any individual chose to withdraw from the research he/she could do so at any time. In case some questions to the participants might sound offensive and uncomfortable to respond to, they were informed not to respond.

11 Reliability, viability and objectivity

For all anthropometric measurements, two values were taken and a third value was taken if the first two measurements differed too much the median was used. A BMI value larger than 60, Z-scores higher than 9.0 and Z-scores less than 0.9 were excluded as invalid.

12 Bias

The fact that the study population was not randomly selected could have introduced selection bias to the study. The Department of Basic Education selected the schools that participated in the study and the first children in each age and gender group who brought back signed consent forms was selected to participate in the study. The number of participants from the rural area was higher than the number of participants from the rural area was higher than the number of participate in the study because no signed informed consent forms were returned) and this could also have led to some bias. The same structured questionnaire was used for all the respondents to ensure consistency. For accurate results, two measurements were taken in each test and the average was used. There are many factors that affect and influence skinfold measurements such as tight skin, too much body fat, body lotion, and injuries. These factors may have affected the accuracy of the results.

Chapter Four: Presentation and interpretation of findings

1 Introduction

Chapter Three outlined the methodology used in this study. This chapter presents and interprets the results of the study. Tables and graphs are used to explain the results. The results are presented according to the sections of the questionnaire. Furthermore, the chapter is subdivided according to the objectives of the study, with the exception that the first section of the chapter is devoted to demographic and descriptive statistics to give an overview of the study population.

2 Demographic profile and medical history of the study participants

One hundred and seventy-four children participated in this study, giving a response rate of 84% (174/208). Every effort was made to include 104 urban and 104 rural children in the study, but only 174 children returned signed informed consent forms and this is dispite the fact that more than 208 forms were distributed. Of the 174 children in this study, 61% (106 children) were from the rural area and 39% (68 children) from the urban area. This is because there was a higher response rate from participants in rural areas than in urban areas.

2.1 Age distribution of the children

All the children were between the ages of six and nine years. The age distribution of the children is shown in **Figure 4.1**. The majority (40%) of the children were eight years old, followed by 28% aged seven years and 23% aged nine years. This was because this age group had the highest response rate and included children from grade 1–4. Few of the children were six years old (9%) because the majority of six-year-old children were still in pre-schools and have thus not been included in this study.



Figure 4.1: The age distribution of the children who participated in this study

2.1.1 The age distribution of the children in the rural versus the urban areas

The age distribution of children in the rural areas versus the urban areas is given in **Table 4.1**. From the table it is clear that for ages six and eight years there were more respondents from the rural areas, but for ages seven and nine years, there were more respondents from the urban areas. The reasons for this discrepancy are not clear, the response rate from the schools in the urban area was generally poor.

Area	Rura	l (n=106)	Urban (n=68)		
	Age (years) Frequency (%)		Age (years)	Frequency (%)	
	6	14.2	6	1.5	
	7	23.6	7	33.8	
	8	44.3	8	32.4	
	9	17.9	9	32.4	

Table 4.1: The age distribution of children from urban versus rural areas

2.1.2 The age distribution of the children between males and females

Table 4.2 below shows the age distribution of the children between males and females. Females of all ages had higher response rates than males. This is due to the fact that the response rate for females in the total population was higher (57%) than for males (43%).

Age	Gender					
(years)	Male	Female				
	Frequency (%)	Frequency (%)				
6	3.5	5.8				
7	10.3	17.2				
8	18.9	20.7				
9	9.8	13.8				

Table 4.2: The age distribution of the children between males and females

2.2 Gender distribution of the children

Every effort was made to include an equal number of females and males in the study; however, the response rate showed that more than half (57%) of the children were females, and only 43% were males. The rural population had 42.4% males and 57.6% females while the urban population had 42.6% males and 57.4% females. Thus, although they were not exactly equal, the gender proportions of the rural and urban populations compared well.

2.3 Age distribution of the parents/guardians

Mothers who are young have an increased risk of giving birth to children who have low birthweight (Aras, 2013). The age distribution of the participants' parents/guardians is shown in **Figure 4.2**. The largest proportion (40%) of the parents/guardians was in the age group 30–39 years, followed by those in the age group 40–49 years (21%). This shows that the majority of the children stayed with their parents or a guardian between the ages of 30 and 49 years. However, a few of the children clearly stayed with their grandparents (over the age of 60 years). In some cases, the parent/guardian was not willing to disclose his/her age, resulting in the "unspecified" group. Thus the majority of the children who participated in this study lived with parents who were clearly above the age of 20 years while being pregnant with these children.



Figure 4.2: Age distribution of the parents/guardians

2.4 Monthly household income

Household monthy income is shown in **Table 4.3** below. Monthly income was grouped into three categories, namely: low-income (less than R1000), middle-income (R1000–R5000) and high-income (R5100–R10 000+) households. More than 20% of the families were in the low-income group with an income of less than R1000 per month and this means that the parents of the children may have been unemployed and depended on social grants or worked unskilled jobs which pay less. Nearly two-thirds (61%) of the parents reported that their monthly income was R5000 or less and only 39% reported an income of more than R5000. Thus the majority of participants in this study were from the middle-income group.

Table 4.3: Monthly household income

Income (per month)	Frequency (%)
< R1000	24.06
R1000–R5000	36.84
R5100–R10 000	13.53
R10 000+	25.56

2.5 Monthly household income between urban and rural population

A comparison of the monthy household income of the parents between urban and rural area is given in **Figure 4.3.** In the rural area 54.14% of respondents reported a monthly income of R5000 or less, compared to 6.77% respondents from urban areas. Approximately 1.5% rural parents reported that their monthly income was more than

R5000, while 37.29% urban parents reported that their monthly income was more than R5000. Thus, as expected the incomes in rural areas were less than that of families residing in urban areas. This will have an impact on the socio-economic status of the children who participated in the study.



Figure 4.3: Monthy household income of urban and rural population

2.6 Availability of motor vehicle

Thirty-six percent of the parents/guardians reported that they own a motor vehicle, while 64% had no motor vehicle. This may be because the majority (54.14%) of parents were from rural areas with a low monthly income (R5000 or less) and cannot afford to buy a motor vehicle.

2.7 Parents medical history

Due to genetics and shared environmental risk factors, many CDL, including obesity runs in families (Chamieh *et al.*, 2015) and this might increase the risk of the children to develop CDLs later in life. The medical history of the parents in this study is shown in **Figure 4.4.** Twenty per cent of the parents reported that they had diabetes mellitus, 16% hypertension, 9% stroke and 7% reported that they had high cholesterol. The majority (78%) of parents/guardians were between 20–49 years, this might be why they were still relatively healthy. In a few years' time this profile may look significantly

different, because the prevalence of these diseases is known to increase with increasing age.



Figure 4.4: Medical history of selected non-communicable diseases in parents/guardians

3 Prevalence of malnutrition (stunting, wasting, underweight, overweight/obesity) in children between urban and rural areas

Table 4.4 below shows the descriptive statistics of anthropometric measurements between urban and rural population. Children in urban areas had significantly higher mean values of anthropometric measurements as compared to those in rural areas.

From **Table 4.4** it is clear that the HFA, WFH, WFA and BF% of the rural children were significantly lower than that of the urban children. This clearly indicates that the markers of undernutrition are significantly more pronounced in the rural compared to the urban population. On the other hand, since the mean BF% of the urban population was significantly higher than that of the rural population it indicates that the children living in urban environments have significantly more body fat than those living in rural environments.

Total P value Area Indices Rural Urban Mean (±SD) Mean (±SD) Mean(±SD) 27.98(8.84) Weight 24.78(5.28) 32.14(11.35) 0.000 Height 1.24(0.08) 1.29(0.09) 1.26(0.09) 0.001 7.78(0.91) 7.66(0.94) 7.96(0.85) 0.037 Age Height for 124.55(8.13) 128.69(8.92) 146(8.66) 0.002 age (stunting) Weight for height 15.87(2.06) 18.89(4.87) 17.03(3.71) 0.000 (wasting) Weight for age 24.78(5.28) 32.14(11.35) 27.98(8.84) 0.000 (underweight) 7.52(3.47) 12.89(8.23) 9.57(6.30) 0.000 Triceps Subscapular 5.22(2.29) 10.92(9.72) 7.39(6.82) 0.000 Abdominal 13.93(12.53) 8.91(9.17) 5.82(3.89) 0.000 Suprailliac crest 5.72(4.02) 13.61(12.56) 8.72(9.17) 0.000 Bodv fat 12.13(5.02) 18.27(9.23) 14.49(7.54) 0.000 percentage (overweight/obese)

Table 4.4: Descriptive statistics of anthropometric measurements and calculations

 used to determine nutritional status in the urban and rural population

BMI = body mass index

Table 4.5 below shows the measurements/calculations used to measure the prevalence of stunting, wasting, underweight, overweight/obesity as an indication of malnutrition in children from urban and rural areas. Cut-off values used to classify children into each of the malnutrition categories were given in **Table 3.2** in Chapter 3. Overweight/obesity were considered as one entity because there were insufficient results of obese children.

In this study population, prevalence of stunting was 14%, wasting 6% and underweight 20% **(Table 4.5)**. A higher proportion of rural children were stunted, wasted and underweight as compared to urban children, however, the results for stunting and underweight were not statistically significant (p > 0.05). Thus despite the fact that the mean values for these indicators of undernutrition were significantly lower in the rural than in the urban population, the prevalence of two of the markers for under nutrion

(stunting and underweight) were not significantly lower in the urban than in the rural population.

According to the WHO's BMI-for-age criteria, the prevalence of overweight/obesity was 26% in this study population. A significantly greater proportion of urban children were overweight and obese as compared to those in rural areas (p < 0.05). This is in accordance with the fact that the urban population had a significantly higher mean percentage body fat compared to the rural population.

	n - 174	Urban (n = 68)	Rural (n = 106)	nyaluaa
	n = 174	Number (%)	Number (%)	<i>p</i> -values
Height for age:				
Stunting	24 (14%)	8 (12%)	16 (15%)	0.534
Weight for height:				
Wasting	11 (6%)	1 (2%)	10 (9%)	0.035
Weight for age:				
Underweight	35 (20%)	11 (16%)	24 (23%)	0.299
BMI for age:				
Overweight/obesity	45 (26%)	27 (40%)	18 (17%)	0.001

Table 4.5: Prevalence of stunting, wasting, underweight, overweight/obesity

4 Prevalence of risk factors for stunting, wasting, underweight, overweight/obesity

Tables 4.6 to 4.9 display the same risk factors for the various types of malnutrition; these risk factors are: gender, child's age, birth weight, mother's age, gestational age at birth, monthly household income, area, sedentary time, mode of transport to school, breastfeeding, consumption of sweets, biscuits and /or cake, takeaways, cold drinks and fresh fruit.

4.1 Stunting

Table 4.6 displays the risk factors for stunting. It is clear that there was a significant relationship (p=0.0030) between gender and stunting, with males having had a higher prevalence of stunting than females. Furthermore, there was a significant relationship (p=0.0158) between stunting and birth weight, with the stunted children having lower birth weights compared to those who were not stunted. However, the mean birthweight

of stunted children was within normal ranges, but at the lower limit of normal (normal being considered as 2.99kg). There were no significant relationships between any of the other risk factors and stunting.

Risk factors	- 474	Stur	nted	n velue		
	n = 174	Yes, n (%)	No, n (%)	<i>p</i> -value		
Gender						
Male	74	17 (23)	57 (77)	0 0030		
Female	100	7 (7)	93 (93)	0.0030		
Child's age						
6	16	1 (6)	15 (94)			
7	48	10 (21)	38 (79)	0.2710		
8	69	8 (12)	61 (88)	0.3710		
9	41	5 (12)	36 (88)			
Birth weight (mean ± sd)	120	2.9 ± 0.6	3.3 ± 0.6	0.0158		
Mother's age at pregnancy	146	28.1 ± 6.9	26.7 ± 5.9	0.3531		
Gestational age (mean ± sd)	116	37.6 ± 2.6	37.9 ± 2.8	0.6313		
Monthly income						
< R1000	32	5 (16)	27 (84)			
R1000–R5000	49	7 (14)	42 (86)	0.0500		
R5100–R10 000	18	3 (17)	15 (83)	0.9590		
R10 000+	34	4 (12)	30 (88)			
Area						
Rural	106	16 (15)	90 (85)	0.5240		
Urban	68	8 (12)	60 (88)	0.5540		
Sedentary hours (mean ± sd)		5.4 ± 6.6	7.3 ± 7.2	0.3146		
Mode of transport to school						
Walk/run	79	8 (10)	71 (90)	0 7220		
Bus/car	76	9 (12)	67 (88)	0.7330		
Breastfeeding						
Yes	137	14 (10)	123 (90)			
No	14	2 (14)	12 (86)	0.7200		
Don't know	5	1 (20)	4 (80)			
Times per week that sweets,						
biscuits and /or cake are		2.9 ± 1.3	3.1 ± 1.9	0.7234		
consumed (mean ± sd)						
Times per week that take away						
meals are consumed (mean ±		1.6 ± 1.8	1.6 ± 1.5	0.9910		
sd)						
Times per week that cold						
drinks are consumed (mean ±		4.0 ± 1.9	4.5 ± 4.8	0.6531		
sd)						
Times per week that fresh fruit		4.5 ± 2.2	4.3 ± 2.7	0.7258		
are consumed (mean ± sd)				0.1.200		

Table 4.6: Risk factors for stunting

4.2 Wasting

Table 4.7 displays the risk factors for wasting in this study population.

Risk factors	n =	Wasted				
	174	Yes, n (%)	No, n (%)	<i>p</i> -value		
Gender		· · · · ·				
Male	74	20 (27)	54 (73)	0.0500		
Female	100	15 (15)	85 (85)	0.0500		
Child's age						
6	16	3 (19)	13 (81)			
7	48	10 (21)	38 (79)	0 0700		
8	69	13 (19)	56 (81)	0.9790		
9	41	9 (22)	32 (78)			
Birth weight (mean ± sd)	120	3.0 ± 0.44	3.2 ± 0.59	0.0734		
Mother's age at pregnancy	146	26.6 ± 7.7	26.9 ± 5.8	0.8006		
Gestational age (mean ± sd)	116	38.6 ± 2.2	37.8 ± 2.8	0.2300		
Monthly income						
< R1000	32	8 (25)	24 (75)			
R1000–R5000	49	13 (27)	36 (73)	0 2350		
R5100–R10 000	18	4 (22)	14 (78)			
R10 000+	34	3 (9)	31 (91)			
Area		1	1			
Rural	106	24 (23)	82 (77)	0 2000		
Urban	68	11 (16)	57 (84)	0.2000		
Sedentary hours (mean ± sd)		4.1 ± 2.2	7.7 ± 7.7	0.0199		
Mode of transport to school		Γ	Γ			
Walk/run	79	22 (28)	54 (68)	0 4000		
Bus/car	76	6 (8)	70 (92)	0.1000		
Breastfeeding		Γ	Γ			
Yes	137	26 (19)	111 (81)			
No	14	2 (14)	12 (86)	0.4460		
Don't know	5	2 (40)	3 (60)			
Times per week that sweets,						
biscuits and /or cake are		3.3 ± 1.4	3.0 ± 1.9	0.4536		
consumed (mean ± sd)						
Times per week that take away		1.8 ± 2.1	1.6 ± 1.2	0.4452		
meals are consumed (mean ± sd)						
Times per week that cold drinks		4.9 ± 3.5	4.3 ± 4.7	0.5536		
are consumed (mean ± sd)						
Times per week that fresh fruit		4.5 ± 3.7	4.2 ± 2.4	0.5306		
are consumed (mean ± sd)						

Table 4.7: Risk factors for wasting

As for stunting there was a significant relationship (p=0.0500) between wasting and gender, with males having had a higher prevalence of wasting than females. There

was also a significant relationship (p=0.0199) between wasting and time spent on sedentary activities. The mean time spent on sedentary activities per day was less for the wasted children than for those who were not wasted (4.1 hours and 7.7 hours respectively). Thus, wasted children may have been more active, resulting in a lower body weight than those who were not wasted. There was no significant relationship between any of the other risk factors and wasting. However, as in the case of stunting the mean birth weight of wasted children was less than that of the children who were not wasted (3.0kg and 3.2kg respectively), but in contrast with stunting, the relationship was not significant (p=0.0734).

4.3 Underweight

Table 4.8 displays the risk factors for underweight. There was a significant difference (p=0.0500) between gender and underweight, with the prevalence of underweight being higher in males than in females (27% and 15% respectively). The same was observed for stunting and wasting. There was no significant relationship between underweight and any of the other risk factors. However once again the mean birth weight for the stunted children was lower than that of the children who were not underweight (3.0kg and 3.2kg respectively). As was the case with wasting, this relationship was not significant.

Risk factors	n - 174	Underv	n volue			
	n = 174	Yes, n (%)	No, n (%)	<i>p</i> -value		
Gender						
Male	74	20 (27)	54 (73)	0.0500		
Female	100	15 (15)	85 (85)	0.0500		
Child's age						
6	16	3 (19)	13 (81)			
7	48	10 (21)	38 (79)	0.0700		
8	69	13 (19)	56 (81)	0.9790		
9	41	9 (22)	32 (78)			
Birth weight (mean ± sd)	120	3.0 ± 0.45	3.2 ± 0.59	0.0734		
Mother's age at pregnancy	146	26.5 ± 7.7	26.9 ± 5.8	0.8006		
Gestational age (mean ± sd)	116	38.6 ± 2.2	37.8 ± 2.8	0.2300		
Monthly income						
< R1000	32	8 (25)	24 (75)	0 2350		
R1000-R5000	49	13 (27)	36 (73)	0.2350		

Table	4.8:	Risk	factors	for	underv	weiaht
IUNIC	T.U .	1,101	1001010	101	under	Noigin

Risk factors	n - 174	Underw	n voluo	
	n = 174	Yes, n (%)	No, n (%)	<i>p</i> -value
R5100–R10 000	18	4 (22)	14 (78)	
R10 000+	34	3 (8.8)	31 (91.2)	
Area				
Rural	106	24 (23)	82 (77)	0 2000
Urban	68	11 (16)	57 (84)	0.2990
Sedentary hours (mean ± sd)	7.1±7.1	5.4 ± 6.5	7.2 ± 7.1	0.3146
Mode of transport to school				
Walk/run	79	17 (22)	62 (78)	0.2540
Bus/car	76	11 (14)	65 (85)	0.2540
Breastfeeding				
Yes	137	26 (19)	111 (81)	
No	14	2 (14)	12 (86)	0.4460
Don't know	5	2 (40)	3 (60)	
Times per week that sweets, biscuits and /or cake are consumed (mean \pm sd)	3.1 ± 1.8	3.3 ± 1.4	3.0 ± 1.9	0.4536
Times per week that take away meals are consumed (mean ± sd)	1.6 ± 1.5	1.8 ± 2.1	1.6 ± 1.2	0.4452
Times per week that cold drinks are consumed (mean ± sd)	4.5 ± 3.5	4.9 ± 3.5	4.3 ± 4.7	0.5535
Times per week that fresh fruit are consumed (mean ± sd)	4.3 ± 2.7	4.6 ± 3.7	4.2 ± 2.4	0.5306

4.4 Overweight/obesity

Table 4.9 displays the risk factors for overweight/obesity. There was a significant association between overweight/obesity and monthly income (p=0.0210), area (p=0.0001), time spent on sedentary activities per day (p=0.0056), mode of transport to school (p=0.0320), as well as breastfeeding (p=0.0560). From the Table it is clear that the overweight/obese population tend to have a higher monthly household income than those who were not overweight/obese. Only 17% of the rural population were overweight/obese, while 40% of the urban population were overweight/obese, thus living in an urban environment seems to increase the risk for overweight/obesity in this population. The mean number of hours per day spent on sedentary activities was 9.6 hours for the overweight/obese population. Of the children who walked or ran to school only 19% were overweight/obese, but 34% of those who were driven to school in a car or bus, were overweight/obese. Of the children who were breastfed only 26% were overweight/obese, while 50% of those who were not breastfed were overweight/obese.

Risk factors	n - 174	Overweight/Obese		n valuo
	n = 174	Yes, n (%)	No, n (%)	<i>p</i> -value
Gender				
Male	74	17 (23)	57 (77)	0 4540
Female	100	28 (28)	72 (72)	0.4340
Child's age				
6	16	2 (13)	14 (87)	
7	48	12 (25)	36 (75)	0 5080
8	69	20 (29)	49 (71)	0.3900
9	41	11 (27)	30 (73)	
Birth weight (mean ± sd)	120	3.4 ± 0.7	3.2 ± 0.5	0.1010
Mother's age at pregnancy	146	26.9 ± 5.3	26.8 ± 6.4	0.9377
Gestational age (mean ± sd)	116	38.1 ± 2.2	37.9 ± 2.9	0.7471
Monthly income				
< R1000	32	7 (22)	25 (78)	
R1000–R5000	49	10 (20)	39 (80)	0.0240
R5100–R10 000	18	5 (28)	13 (72)	0.0210
R10 000+	34	17 (50)	17 (50)	
Area		· · · ·	· · · ·	
Urban	68	27 (40)	41 (60)	0.0001
Rural	106	18 (17)	88 (83)	0.0001
Sedentary hours (mean ± sd)		9.6 ± 9.2	6.1 ± 5.9	0.0056
Mode of transport to school				
Walk/run	79	15 (19)	64 (81)	0.0220
Bus/car	76	26 (34)	50 (66)	0.0320
Breastfeeding		· · · ·	· · · ·	
Yes	137	35 (26)	102 (74)	
No	14	7 (50)	7 (50)	0.0560
Don't know	5	-	5 (100)	
Times per week that sweets,				
biscuits and /or cake are		3.2 ± 2.1	3.1 ± 1.7	9.7189
consumed (mean ± sd)				
Times per week that take away				
meals are consumed (mean ±		1.6 ± 1.2	1.6 ± 1.6	0.9182
sd)				
Times per week that cold				
drinks are consumed (mean ±		5.3 ± 7.6	4.1 ± 2.6	0.1515
sd)				
Times per week that fresh fruit		44+30	42+26	0 6670
are consumed (mean ± sd)		т.т <u>т</u> Э.О	T.2 I 2.0	0.0070

Table 4.9: Risk factors for overweight/obesit	y
---	---

Thus it seems as if belonging to a lower socio-economic group, living in an urban area, spending more time on sedentary activities, being driven to school and not being breastfed as a baby, were associated with overweight/obesity in this study population. With regards to income, the middle and high income groups seem to have been more

likely to be overweight/obese. Being breast fed was associated with not being overweight/obese, but despite this fact, 50% of children who were not breastfed were not overweight/obese. This may be because there are so many different risk factors that play a role in the development of overweight/obesity, and if a baby was not breastfed, but were not exposed to many of the other risk factors for overweight/obesity he/she will not necessarily develop the condition.

Table 4.10 below shows the descriptive statistics of the food consumption per week in the urban and rural population. It is clear that urban children consumed significantly more fresh fruit (p=0.027) and take away meals (p=0.017) than the rural children. The consumption of sweets, biscuits, cakes and cold drinks did not differ significantly between the two populations, but consumption of these products were higher in the urban than in the rural population. This is an indication that the urban population is following a more western, energy dense diet than the rural population. Take away meals are known to be rich in fat and refined carbohydrates and are therefore energy dense (You & Henneburg, 2016).

Table 4.10: Descriptive stat	istics for food consum	ned per week in the urb	an and rural
population			

Variables	Area		P value
	Rural	Urban	
	Mean(±SD)	Mean (±SD)	
Times per week that sweets, biscuits and	3.01(1.78)	3.25(1.91)	0.284
/or cake are consumed			
Times per week that cold drinks are	4.00(1.41)	5.73(6.09)	0.583
consumed			
Times per week that fresh fruit are	4.00(2.65)	4.78(2.67)	0.027
consumed			
Times per week that take away meals are	2.84(1.90)	4.33(0.71)	0.017
consumed			

5 Pearson correlation between weekday and weekend dietary intake and body fat percentage of the study population

Dietary analysis was done for 56 (32%) of the total study population, to investigate a possible correlation between weekday and weekend dietary intake and BF% in this study. Food finder was only able to calculate the micronutrients and macronutrients for children shown in **Table 4.11** and **Table 4.12** below.

	Area		Total	P value
Variables	Rural	Urban		
	Mean(±SD)	Mean (±SD)	Mean (±SD)	-
Total energy	43.13(17.13)	43.99(19.73)	43.59(18.41)	0.863
Total protein	101.30(49.29)	86.29(37.91)	93.26(43.81)	0.204
Са	38.33(31.38)	28.17(23.81)	32.89(27.84)	0.176
Fe	49.08(27.62)	48.83(26.02)	48.95(26.53)	0.973
Р	63.51(37.20)	56.50(28.91)	59.76(32.91)	0.432
Zn	31.35(15.13)	32.45(15.32)	31.94(15.11)	0.788
Vit A	48.96(48.56)	34.01(48.09)	40.95(48.45)	0.253
Vit D	58.60(52.67)	16.41(22.63)	35.91(44.52)	0.000

 Table 4.11: Descriptive statistics of weekday macronutrients and micronutrients

 between urban and rural population

 Table 4.12: Descriptive statistics of weekend macronutrients and micronutrients

 between urban and rural population

	Area		Total	P value
Variables	Rural	Urban		
	Mean(±SD)	Mean (±SD)	Mean (±SD)	
Total energy	32.27(12.28)	40.81(27.24)	38.99(22.35)	0.035
Total protein	80.23(47.51)	85.44(38.93)	83.02(42.84)	0.654
Са	12.41(8.04)	24.00(24.68)	18.66(16.90)	0.027
Fe	32.00(18.88)	45.20(41.84)	39.07(33.60)	0.144
Р	37.35(14.75)	48.61(30.07)	43.43(24.66)	0.086
Zn	28.41(21.05)	28.90(17.94)	28.71(19.27)	0.937
Vit A	14.09(22.76)	27.82(48.55)	21.44(39.06)	0.192
Vit D	10.88(10.71)	17.74(17.19)	14.56(14.83)	0.085

All the children attending school in rural areas had the privilege of getting lunch at school as part of the government feeding scheme, while only one of the urban schools had a feeding scheme for some of the children. From **Table 4.11** it can be seen that the nutrient consumption of children in rural and urban areas did not differ significantly, except for Vit D consumption which was significantly higher in the rural population. The reason for the higher Vit D intake in the rural population is not clear, but may perhaps be explained by the fact that all rural children ate lunch prepared at school as part of the feeding scheme, while only a small number of urban children received lunch at school. According to **Table 4.12** it is clear that during weekends when rural children do not have the benefit of the feeding scheme, the total energy intake and Ca intakes of the rural children were significantly lower than that of the urban children. The Vit D intake of the rural children over weekends was not significantly higher than that of the urban children (as during the week), in fact over weekends rural children consumed less Ca than urban children. These results clearly indicate the benefit of general school feeding schemes in rural areas.

percentage of the study population			
	Pearson correlation factor (r)	Significance (p)	
Energy	-0.091	0.507	
Protein	-0.185	0.171	
Са	-0.216	0.110	
Fe	0.060	0.659	
Р	-0.169	0.214	
Zn	-0.025	0.854	
Vit A	-0.301	0.024	

Table 4.13: Pearson correlation between weekday dietary intake and body fat

-0.344 Ca = Calcium, Fe = Iron, P = Phosphorus, Zn = Zinc, Vit A = Vitamin A; Vit D = Vitamin D

0.009

Vit D

 Table 4.13 displays the results for the Pearson correlation between weekday dietary
 intake and BF%. There were statistically significant, but negative relationships between Vit A (p=0.024; r=-0.301) and Vit D (p=0.009; r=-0.344) intake and BF% however, there were no statistically significant associations between total energy intake, consumption of proteins, consumption of micronutrients (Ca, Fe, P, Zn) and BF%.

 Table 4.14: Pearson correlation between weekend dietary intake and body fat

 percentage of the study population

	Pearson correlation factor (r)	Significance (p)
Energy	0.011	0.934
Protein	0.019	0.890
Ca	0.069	0.616
Fe	0.089	0.514
Р	0.047	0.731
Zn	0.008	0.951
Vit A	0.033	0.809
Vit D	-0.119	0.383

Ca = Calcium, Fe = Iron, P = Phosphorus, Zn = Zinc, Vitamin A = Vit A; Vitamin D = Vit D

Table 4.14 displays the results for the Pearson correlation between weekend dietary intake and BF%. There were no statistically significant associations between total energy intake, consumption of proteins, consumption of micronutrients (Ca, Fe, P, Zn, Vit D and A) (p > 0.05) and BF%.

The absence of significant associations between energy intake and BF% may indicate that the nutritional data collected were not sufficiently accurate, but it may also indicate that there are numberous other factors that influence BF%, and mask the effect of energy intake.

Chapter Five: Discussion

1 Introduction

In Chapter Four, the results of the study were presented and interpreted. This chapter discusses the implication of the major findings as presented in Chapter Four. Furthermore, findings of the current study were compared with, and discussed in the light of previous studies. The objectives of this study were to: compare the prevalence of stunting, wasting, underweight, overweight/obesity between rural and urban children; compare the prevalence of risk factors associated with stunting, wasting, underweight, overweight, and to determine correlations between dietary intake and body fatness.

2 Prevalence of stunting, wasting, underweight, overweight/obesity in children between urban and rural areas.

2.1 Stunting

Reports on the prevalence of stunting in children living in SA vary considerably. The National Food Consumption Survey (NFCS) that was conducted in SA in 1999 and focused on children between the ages of one to nine years found that 23% of children were stunted (Labadarios *et al.*, 2005). In the North-West province of SA, a study found that the prevalence of stunting was 4.2% among children in Grade 1 (Kruger *et al.*, 2014). Said-Mohamed *et al.* (2015) reported that the prevalence of stunting in South African children has increased between 2005 and 2008, from 23.2% to 30%. Other African countries also report a considerable variation in the prevalence of stunting. A study conducted in Nigeria amongst children in the age group five to nineteen years reported that the prevalence of stunting was 17.4% (Senbanjo *et al.*, 2011).

Finding studies on stunting in children between specifically six and nine years is difficult and as a result the studies used to compare the current study to included children younger than six years and/or children older than nine years. This fact could also in part explain the differences in stunting between this study population and the other study populations. In the present study, the prevalence of stunting in the total population (14%) was lower as compared to the Nigerian study and the NFCS, and higher as compared to the South African study in the North-West province. The higher
prevalence of stunting in this study as compared to the North-West province study might be due to the delay of linear growth caused by chronic poor health and poor nutrition from areas which are more rural (de Onis & Branca, 2016). The low prevalence of stunting in this study, as compared to that in the Nigerian study, might be due to the different socio-economic groups. The study in Nigeria was conducted in an urban area with low socio-economic group where children attended public schools, while the current study was conducted not only in an urban area, but also in rural areas and subjects were not selected on the basis of socio-economic groups, and as can be seen from **Figure 4.3** the urban population in this study reported higher household income than the rural population.

Labadarios *et al.* (2005) found that the prevalence of stunting amongst rural and urban South African children between the ages of one and nine was 27% and 17% respectively. Furthermore, the THUSA BANA Study in the North-West province of SA also reported that stunting is more prevalent in the rural areas (girls 23.7% and boys 26.7%) than in the urban areas (girls 11.6% and boys 17.1%) and the difference was not significant (Mukuddem-Petersen & Kruger, 2004). The same is true for other African countries. Herrador *et al.* (2014) indicated that the prevalence of stunting among rural and urban children in Ethopia was 42.7% and 29.2% respectively.

In agreement with the studies mentioned above, in the current study a greater proportion (15%) of rural children were stunted as compared to urban children (12%) and the result was not significant and although this was not significantly different the mean HFA in the rural population was significantly lower than in the urban population. This might be due to poor socio economic background in rural and low-middle income in rural families (Said-Mohamed *et al.*, 2015).

The majority of the children in the current study are from low-middle income households in rural areas, as shown in **Figure 4.3**. Parents/guardians of the children in rural areas often do not have tertiary education, therefore, they are often unemployed, or they work on farms, as domestic workers and other forms of unskilled labour (Govender *et al.*, 2017). Moreover, the parents of children in such areas often rely on child support grants, which do not allow them to afford nutritionally adequate

diet, because food prices in SA have been steadily increasing (van Wyk & Dlamini, 2018). Therefore, children from poor families are more likely to be stunted than children from wealthy families.

The majority of the children stay with their parents or guardians who are 30–39 years old, as depicted by **Figure 4.2**. The women in rural areas are unable to meet their nutritional requirements during pregnancy due to poverty and high unemployment rates (Govender *et al.*, 2017). These women are more likely to give birth to low birthweight infants who are at risk of illness and grow up to be stunted (Krishna *et al.*, 2016). Unfortunately, the weight and dietary information of the mothers of the children in the current study were not available to researchers of the current study.

2.2 Wasting

The prevalence of wasting in children between the ages of one and nine was reported to be 19% in a study conducted by Berry *et al.* (2010) in Cape Town, SA. A study conducted in Burkina Faso amongst children in the age group seven to fourteen years found the prevalence of wasting to be 13.7% (Dabone *et al.*, 2011). In contrast, the prevalence of wasting in this study was lower (6%) as compared to these two studies. This might be due to the implementation of the feeding schemes in South African primary rural school children (van Stuvjenberg *et al.*, 2005). Moreover, the introduction of a child support grant with the aim of supporting children in the form of cash might also play a role in decreased prevalence of wasting (Triegaardt, 2005). In five of the schools that participated in this study there were feeding schemes, three schools did not have any feeding schemes.

In this study, the prevalence of wasting was significantly higher in rural children (9%) than urban children (2%), (p = 0.035) and the mean WFH in the rural population was significantly lower than in the urban population. Labadarios *et al.* (2005) found that the prevalence of wasting amongst rural and urban children between the ages of one and nine years in SA was 5% and 17% respectively. However, an Ethopian study found a higher prevalence rate of wasting in rural areas than in urban areas (Herrador *et al.*, 2014). The higher prevalence of wasting in rural areas than in urban areas (Herrador *et al.*, and the higher prevalence in this study being from rural families (61%) with low-to-middle-incomes (R5000 or less). Furthermore, there is availability of food in SA, however,

food prices have been steadily increasing, making it less accessible to low and middleincome families (STATS SA, 2016). The feeding schemes introduced in South African rural primary schools 20 years ago might not contain food which have insufficient micronutrients and macronutrients (van Stuijvenberg, 2005).

2.3 Underweight

Twenty per cent of the children in the present study were underweight. The results of this study are in agreement with studies which found a prevalence rate of 14.5% amongst Indonesian children aged between 6 and 13 years (Syahru *et al.*, 2016). In contrast, a study by Chirita-Emandi *et al.* (2016) of children between the ages of six and nine years, from eight Romanian countries, residing in urban and rural areas found the prevalence of underweight to be 5%. The high prevalence of underweight in this current study might be due to low income levels and high poverty rates (Govender *et al.*, 2017)

Labadarios *et al.* (2005) found that the prevalence of underweight amongst rural and urban children between the ages of one and nine years in SA was 5% and 17% respectively. In contrast a study in India reported that the prevalence of underweight was 46% and 32% in rural and urban areas, respectively (Meshram *et al.*, 2016). Similarly, in the present study the prevalence of underweight was higher in rural than urban areas and although this was not significantly different the mean weight for age in the rural population was significantly lower than in the urban population. This may be due to a high percentage of parents residing in rural areas where there is low income level, and high poverty rates (Govender *et al.*, 2017).

2.4 Overweight/obesity

Labadarios *et al.* (2005) also found that the prevalence of overweight/obesity in children between the ages of one and nine years in SA was 10% and 4% respectively. Furthermore, the prevalence of overweight/obesity amongst 3-10-year-old children from Gauteng and Mpumalanga provinces was 12.0% and 3.7% respectively in a study conducted by Symington *et al*, (2016). In contrast, the prevalence of overweight/obesity amongst six to thirteen-year-old primary school children in SA was 31.1% and 8.1% respectively (Armstrong *et al.*, 2006). The prevalence of overweight/obesity amongst six-to-nine-year-old children was 5.6% and 6.3%

respectively in a study conducted by Mosha and Fungo (2010) of Dodoma and Kinondoni Municipalities, Tanzania respectively. The combined prevalence of overweight/obesity in the current study was 26% which confirms findings of the study conducted by Armstrong *et al.* (2006). Worldwide there is a trend of increasing overweight/obesity in children that is caused by lifestyle changes associated with urbanisation and modernisation (van Zyl *et al.*, 2012). SA, even the more rural parts such as the Limpopo province, is clearly no exception.

A South African study conducted by Labadarios *et al.* (2005) in children between the ages of one and nine years found that the prevalence rates amongst rural and urban were 12% and 13% respectively for overweight, and 4% and 6% respectively for obesity. In agreement with the previous study, Ahmed *et al.* (2016) also found higher prevalence rates of overweight/obesity in urban areas than rural. In China, a higher prevalence of overweight was found in urban areas than in rural areas among children in the age group seven to eighteen years (Zhang *et al.*, 2016). The findings of the current study also confirm the results of previous studies. In the present study, the combined prevalence of overweight/obesity was also higher in urban children (40%) as compared to rural children (17%) and the mean BF% in the rural population was significantly lower than in the urban population. National Health and Nutrition Examination (NHANES), (2010) stated that overweight/obesity was more prevalent in high-income households for the past 32 years. Chen *et al.* (2012) reported that children in high-income households were more likely to follow a Western diet, which is high in fat and carbohydrates, and thus increase the body fat accumulation.

In the present study, monthly income was grouped into three categories, namely: lowincome (less than R1000), middle-income (R1000-R5000) and high-income (more than R5100) households. It is clearly seen from **Figure 4.3** that the majority of the children belong to the low-middle-income households in the rural setting. This might be due to the fact that majority of the parents in rural setting works in Gauteng and send money to their families on a monthly basis (Govender *et al.*, 2017). Based on our findings, overweight/obesity is more prevalent in the middle-income households compared to the low-income households. This might be because the middle-income households can afford to buy foods often found in Western diets, such as proteins (e.g. meat) and soft drinks compared to the low-income households (Chen *et al.*,

61

2012). However, a few children belong to the high-income households in the rural setting, which might be due to the high unemployment rate of the area. Furthermore, the majority of the parents in the rural setting have poor education; therefore, they are usually only eligible to work on farms and as domestic workers (Govender *et al.*, 2017).

The high combined prevalence of overweight/obesity in urban children might be due to children consuming large amounts of energy-dense food and decreased physical activity, whereas low prevalence of overweight in rural areas may be due to high levels of physical activity and following a prudent diet (Friedman, 2003). In the current study urban children consumed significantly more take away meals than rural children (**Table 4. 10**; p=0.017) while the children residing in rural areas consumed significantly more fruit than those residing in urban areas (**Table 4.10**; p=0.027). Take away meals are known to be energy dense as it contains large amounts of fat, refined carbohydrates and sugar (Lillefosse *et al.*, 2015).

3 The prevalence of risk factors of stunting, wasting, underweight, overweight/obesity

3.1 Stunting

Monyeki *et al.* (2000) conducted a study in Lephalale (formerly known as Ellisras) in the Limpopo province of SA on children between the ages of seven and eight years. Their results indicated that 5.5% of males and 3.5% of females were stunted. Similarly, in this study 23% of the males and 7% of the females were stunted and there was a significant association between gender and stunting (p = 0.0030). This suggests that males in both these studies might be more vulnerable to frequent inappropriate feeding practices and poverty than female counterparts of the same age. In Tanzania, Abubakar *et al.* (2012) reported that child age (p = 0.001) and breastfeeding (p < 0.001) were significantly associated with stunting. The present study reported that child age and breastfeeding were not statistically associated with stunting (p = 0.3710 and p = 0.7200 respectively). In another study, child age was a risk factor associated with stunting in children (p < 0.001) living in rural Eastern Cape and Kwazulu-Natal, SA (Lesiapeto *et al.*, 2010). A study conducted in rural Limpopo and urban Gauteng found a statistically significant association between low birth weight and stunting (Kleynhans *et al.* (2006). Kleynhans *et al.* (2006) conducted a study amongst children

in rural Limpopo and urban Gauteng and also reported that low birth weight was a risk factor associated with stunting (p = 0.0156). In the present study, stunted children had a significantly lower birth weight than non-stunted children (p = 0.0158), however, the average birth weight of the stunted children was within normal ranges (2.9 kg ± 0.6). Thus it seems as if low birth weight is a risk factor for stunting in this population, similar to what has been reported in other studies. This is clearly demonstrated by the results of the studies, where birth weight was significantly associated with stunting (Kleynhans *et al.*, 2006).

3.2 Wasting

Panigrahi and Das (2014) reported that monthly income and the mother's age at pregnancy were not statistically associated with wasting in a study conducted on children aged three to nine years residing in India. Similarly, in this study monthly income (p = 0.207) and the mother's age at pregnancy (p = 0.9016) were not associated with wasting. Children who reside in rural areas are more likely to be wasted because of poor environmental and socio-economic, factors, poverty and lack of proper feeding practices (Wolde et al., 2015). In the present study there was no significant association between area and wasting, thus living in a rural area does not seem to be a risk factor for wasting in this population. In fact, wasting was more prevalent in the urban population (23%) as compared to the rural population (16%). Similarly, a study conducted by Oninla *et al.* (2006) in Nigerian children reported that the prevalence of wasting was also higher in urban areas than in rural areas. This might be due to urbanisation which results in poor sanitation and a lower household income (Buttenheim, 2009). A study conducted by Harding et al. (2018) reported that mothers who reside in urban areas and gave birth more than twice were more likely to give birth to a wasted child than women who had given birth less frequently. Furthermore, working mothers were also likely to give birth to wasted children than mothers who were not working. In the present study, neither the number of births prior to the subject being born, nor whether the mother is working or not, has been measured.

Feeding practices and wasting in children were found to be statistically significant (p < 0.05) (Islam *et al.*, 2015). In the present study, wasted children reported a higher frequency of consuming takeaways than children who were not wasted, but this was

not significant. Takeaways might be deficient in nutrients that are needed by the body and thus result in wasting. Thus the result from the present study is in agreement with suggestions in literature that poor feeding practices and inadequate food (or nutrient) intake play a role in determining weight and height of a child (Angood *et al.*, 2016).

3.3 Underweight

Underweight was statistically associated with male children (p = 0.021) in a study conducted by Lesiapeto et al. (2010) in rural districts of the Eastern Cape and KwaZulu-Natal provinces of SA. This is in accordance with the results of the present study where there was also a significant relationship (p=0.0500) between gender and underweight, with males having a higher prevalence of underweight than females (27% and 15% respectively). A possible explanation for this trend may be that males were more active than females (Telford et al., 2016). An increase in physical activity is often associated with a decrease in BMI (Curtner-Smith et al., 2007). A study conducted by Abubakar et al. (2012) in Tanzania among children reported that low birth weight was not statistically associated with underweight (p = 0.07). In agreement, the present study also showed that underweight and birth weight were not significantly associated (p = 0.0734). However, the current study revealed that for all markers for undernutrition (stunding, wasting and underweight), the undernourisched population had lower birth weights than the population who was not undernourished. This relationship was only significant in the case of stunting (p=0.0158), and not in the case of wasting (p=0.0734) and underweight (p=0.0734). Thus in this study population there is a trend that children who were undernourished also had lower birth weights.

In the present study, the prevalence of underweight was higher in rural than in the urban population (23% and 16% respectively). This is in contrast with the results from a study conducted by Oninla *et al.* (2006) in Nigerian children, where the prevalence of underweight was higher in an urban area than a rural area. The easy access and availability of food lacking proper nutrients, that are not approved by food and drug administration (FDA), might lead to development of underweight (Khan *et al.*, 2018). Furthermore, children in this age group tend to under-feed themselves (Bryant-Waugh *et al.*, 2010).

3.4 Overweight/obesity

A study conducted by Cort et al. (2013) in KwaZulu-Natal, SA, reported that socioeconomic factors such as high monthly income have an influence on the accumulation of body fat. Cawley (2010) reported that high income promotes obesity, by enabling people to eat more calories and exert low physical activity due to sedentary pursuits. Therefore, households with high monthly income are more prone to overweight/obesity than households with low income. In this study, similar findings were reported, whereby the children from high monthly income households had a significantly higher prevalence of overweight/obesity (p = 0.0210) than children from low monthly income households. Speculations about these findings are that children from a household with a high monthly income engage in more sedentary activities, resulting in less energy expenditure. These findings are supported by Table 4.9, where there is an increase in sedentary time. Furthermore, accessibility of cars in a household with high income which decreases physical activity may result in a more sedentary lifestyle. In contrast, Jiang et al. (2006) found no association between overweight and monthly income (p =0.56) in children two to six years residing in China. The results of the study conducted by Jiang et al. (2006) further indicated that there was no association between time spent playing video games and obesity. In contrast, the results of the current study reported a highly significant association between sedentary activities and overweight/obesity (p = 0.0056). Kyallo et al. (2013) conducted a study amongst public and private primary school children in Nairobi, Kenya, and reported an association between residential area and overweight/obesity (p = 0.028). Similarly, in the present study there was a highly significant association between area and overweight/obesity (p = 0.0001). In this study the majority of overweight/obese children lived in the urban area (40%) and not in the rural area (17%). Thus, in the present study, children residing in an urban area were at significantly higher risk for being overweight/obese. Chen et al. (2012) reported that children in high-income households were more likely to follow a Western diet, which is high in fat and carbohydrates, and thus increase the body fat accumulation. Furthermore, possible reasons for these findings might be that urban areas are modernised due to urbanisation (Monyeki et al., 2015), whereby children use a bus/car as a mode of transport, which decreases their physical activity levels; this is supported by the results in **Table 4.9** where there is a high prevalence of overweight and obese children who use car/bus as mode of transport to school and the association is significant (p = 0.0320). The children engage in more sedentary time

and consume a diet rich in food containing saturated fats and carbohydrates (You & Hennesburg, 2016).

Yan et al. (2014) reported that breastfeeding was associated with a significantly reduced risk of overweight/obesity in children (p < 0.05). In contrast, Gopinath *et al.* (2012) reported that breastfeeding was significantly associated with the development of overweight/obesity in pre-school children (p = 0.002). In agreement, breastfeeding was significantly associated with overweight/obesity (p = 0.0560) in this study, However, of the children who were breastfed only 26% were overweight/obese, while 50% of those who were not breastfed were overweight/obese. Jing et al. (2014) further indicated that children who were breastfed for more than eleven months were more likely to be overweight and obese than those who were breastfed for only six to eleven months of their lives. Furthermore, Mamabolo et al. (2004) reported that mothers tend to introduce supplementary feeds and formula milk at an early age which causes excess feeding. In agreement, Stuebe (2009) reported that formula fed children are more likely to become obese later in life. However, Jing et al. (2014) reported that breastfeeding has beneficial effects, such as reducing the development of type 2 diabetes mellitus and obesity. This could be a possible explanation for the low prevalence of overweight and obese children who were both breastfed and formula fed in this study.

A South African study conducted by Labadarios *et al.* (2005) in children between the ages of one and nine years found that the prevalence rates amongst rural and urban were 12% and 13% respectively for overweight, and 4% and 6% respectively for obesity. In contrast, the prevalence of overweight/obesity in the present study was higher in rural areas than in urban areas. Modernisation and urbanisation might be the cause of this high prevalence of overweight/obesity in the rural areas. In the present study, overweight/obesity was more prevalent in the middle and higher income groups than in the lower income groups. This might result in easy access and availability of food that are not approved by FDA from shops. Furthermore, this might result in decreased physical activity levels, due to the fact that, in rural areas, the availability of excerise and sport facilities is low (Hoekman *et al.*, 2013). Thus there is little opportunity for children to be active after school in sports and events, while unsafe conditions (due to crime) makes it impossible for the children to play outside (Senda,

66

2015). Furthermore, the use of mobile transport and the consumption of food rich in saturated fats also promote a positive energy balance (Lillefosse *et al.*, 2014).

4 Pearson correlation between weekday and weekend dietary intake and BF% of the study population

Pearson correlations were conducted to investigate probable correlations between dietary intake, both during the week and over weekends, and BF%. The data analysis was done separately for week days and for weekend days to investigate probable differences in dietary intake between week days and weekend days.

A study in school children in Western Algeria showed that BF% was significantly associated with percentage of total energy intake (Saker *et al.*, 2011). In contrast, the present study reported that there was no significant correlation between BF% and percentage of energy intake (p > 0.05) during either the week or over weekends. However, the correlation was negative for week days and positive over weekends. A possible reason could be that an increase in physical activity levels during childhood may have an influence on body fat levels whereby there is an increase in energy expenditure (Atkin & Davies, 2000). It is also possible that the collection of dietary intake data was not accurate or that there are various other factors that are masking the relationship.

You and Henneberg (2016) reported that protein is highly correlated with prevalence of obesity (r = 0.666, p < 0.001). However, the present study shows that there was a weak negative but not significant correlation between protein and BF% during weekdays (r = -0.185, p = 0.171) and a weak positive but not significant correlation over weekends (r = 0.019, p = 0.890). A similar result was found among five-to-twelveyear old Mexican obese children (r = -0.07, p = 0.07) (Moschonis *et al.*, 2012). The negative association between protein intake and BF% during weekdays may be due to the fact that children conume a disproportionate amount of energy from fats and carbohydrates and do not consume sufficient proteins. The government feeding schemes probably provide food which are not rich in proteins, parents pack lunch boxes with less proteins, the portion size of proteins is not sufficient, or the children do not like consuming food rich in protein (Arsenault & Brown, 2017). Furthermore, in this study a large proportion (37%) of the households have a monthly income of R1000– R5000, as depicted in **Table 4.3**. This might result in purchasing food which is poor in protein as it is known that food rich in proteins is expensive. In populations where there is a low monthly income, a typical meal contains mostly starch-rich products, such as maize, as this is more affordable (Michaelsen *et al.*, 2009). This low protein intake might therefore lead to increased body fat which might lead to the development of obesity. These results are supported by **Table 4.9**, where children who consume a high amount of sweets, biscuits and /or cake and cold drink per week are more likely to be overweight and obese, but the association is not significant (p > 0.05). Over weekends it is possible that the consumption of proteins increase, causing the positive relationship between protein intake and BF%.

A high intake of Ca was associated with low body fat in a study conducted by Carruth and Skinner (2009) in the United States. Similarly, Dixon *et al.* (2005) found an inverse correlation between Ca and BF% in US children. In contrast, the current study reported no significant correlation between Ca and BF% for weekdays (r = -0.216, p = 0.110) or weekend days (r = 0.069, p = 0.616). There was a significant but negative correlation of low Fe levels with BMI in a study conducted by Pinhas-Hamiel *et al.* (2000) in Israel. However, the current study shows that there was no significant correlation between Fe and BF% during the week (r = 0.060, p = 0.659) or over weekends (r = 0.089, p = 0.514). The present study found that Zn was not significantly associated with BF% during the week (r = -0.025, p = 0.854) or over weekends (r =0.008, p = 0.951). The fortification of maize meal and wheat flour (used for bread) may have increased the consumption of micronutrients in children and therefore there may not have been correlations between BF% and some micronutrients. Furthermore, the school feeding schemes may also have significantly contributed to the consumption of micronutrients such as Ca and Fe.

Garcia *et al.* (2013) reported that Vit A in Mexican children was positively associated with BF%. In agreement, there was a significant but negative association between BF% and Vit A in the current study during the week (r = -0.301, p = 0.024); thus, the lower the Vit A intake, the higher the chances for a child to be overweight or obese (Yang *et al.*, 2015). However, the association was positive but not significant over weekends (r = 0.033, p = 0.809). There was a significant but negative correlation

between BF% and Vit D during the week (r = -0.301, p = 0.09) in the current study. Over weekends the association was negative but not significant (r = -0.119, p = 0.383) Furthermore, Lee *et al.* (2013) found that Vit D was also inversely associated with BF% in preadolescent children residing in Korea. Buyukinan *et al.* (2012) reported that deficiency in Vit D causes children to have excess body fat.

Thus, in this study population there were differences in food intake during the week compared to over weekends, as was expected. In general, the associations (whether significant or not) were negative during the week and positive over weekends. The school feeding schemes may have had an impact on the consumption of micronutrients by childer and have improved the micronutrient nutrition of the rural children, which were also generally in the lower socio-economic groups compared to the urban children.

Chapter Six: Conclusion and recommendations

1 Conclusion

The first objective of this study was to compare the prevalence of stunting, wasting, underweight (as indicators of undernutrition), and overweight/obesity (as an indicator of overnutrition) between rural and urban children. In this population, the indicators of undernutrion were more prevalent in rural than in urban areas, however this was only significant for wasting (p=0.035). On the other hand, the indicator of overnutrition was significantly more prevalent in the urban compared to the rural population (p=0.001). these results are in accordance with similar studies done in SA, Africa and in other countries. These results imply that the prevalence of malnutrition is increasing alarmingly and it indicates that this population is still facing a double burden of malnutrition (undernutrition as well as overnutrition). Socioeconomic factors, urbanisation, modernisation (FAO, 2015) and mothers' lack of knowledge about malnutrition are the most important causes of malnutrition among children (Saaka, 2014). With such a poor start in life, these children are likely to never reach their full potential as they might develop short- and long-term health consequences later on in life (Sahoo et al., 2015). Childhood obesity tends to continue and even increase into adulthood (Sahoo et al., 2015). Thus if 26% of these primary school children are already overweight or obese, the prevalence is likely to increase as the children grow older. This may have a far-reaching impact on their future health status, and this needs to be taken into account by health care services to ensure that there will be proper medical treatment available for them. It is also of the utmost importance to design effective prevention programmes in order to prevent further development of overweight/obesity in this population.

The second objective of this study was to investigate the association between malnutrition and some of its known risk factors. In the current study, children from both urban and rural areas suffered from malnutrition. This is in accordance with other studies done in SA, Africa and other countries. Malnutrition is not only due to lack of good nutrition, but are also often worsen by adverse socio-economic conditions, poor education, poverty, and infections (Leddy *et al.*, 2008). Overweight/obesity seems to

be increasing in the rural areas due to urbanisation and modernisation (Monyeki *et al.*, 2015). In the current study gender was consistenly associated with undernutrition (males being more undernourished than females). This was not true for overnutrition. Birth weight tended to be lower in the undernourished children, with a significant association between birth weight and stunting (p=0.0030). Even though the association between birth weight and wasting as well as underweight was not significant (p=0.0734 in both cases) it shows a trend towards an association between birth weight and undernutrion. This is in accordance with literature and may be due to the poor socio-economic conditions of a large percentage of this study population. Mothers who are poor, often do not have proper nutrition during pregnancy resulting in low birth weight babies, who also will not have proper nutrition growing up, resulting in stunting, wasting and underweight (Aras, 2013).

In terms of overnutrition, the current study shows that there are a number of factors that increase the risk for overnutrition in this study population. These risk factors are middle to high household income, living in an urban area, having a sedentary lifestyle, being driven to school and not being breastfed. From these risk factors one can see that a western-type lifestyle, characterised by sedentary activities, motor vehicle transport, and a higher household income was associated with the development of overweight and obesity. Breastfeeding has been reported to protect against the development of overweight and obesity and this is in accordance with the results of the present study. Children who were breastfed had a lower prevalence of obesity compared to those who were not breastfed. However, many children who were not breastfed, were also not overweight/obese.

Table 6.1 below summarizes the risk factors associated with malnutrition in this population. It is clear that being a male is a risk factor for having a nutritional status below normal, while there are more risk factors for having a nutritional status above normal.

Malnutrition category	Risk factors for this study population
Stunting	Male gender
	Low birth weight
Wasting	Male gender
	Not having a sedentary lifestyle
Underweight	Male gender
Overweight/obesity	Middle to high-income household
	Urban area
	Sedentary lifestyle
	Being driven to school
	Not being breastfed

Table 6.1: The risk factors associated with malnutrition in this study population

In this study, low protein, Ca, Vit A and Vit D intake were associated with increased BF% (although not significant in all cases). Often these nutrients are found in food containing protiens. Proteins take a longer time to digest compared to carbohydrates and thus if children eat more protein they are likely to feel full for longer, making them less susceptible to snacking. Children replacing proteins with carbohydrates in their diets, may feel hungry relatively soon after a meal and will therefore be more likely to snack between meals. In low socio-economic settings such snacks often consists of refined carbohydrates and significant amounts of fat, which may, at least partly explain the association of these nutrients and BF%.

2 Limitations

- This was a cross-sectional baseline study; therefore, causality could not be determined.
- This research project was limited to the rural and urban primary schools of PLM, Capricorn District, Limpopo Province which were selected by the Senior District manager of the Capricorn District.
- The rural primary schools selected consisted of only black children, whereas the urban primary schools had white, coloured, Indian and black children.
- The study only included children from six to nine years of age.
- Although parents/gaurdians were requested to complete the questionnaire as truthfully as possible, there is no way of determining how "true" the answers to the questions were.
- Some of the questions were left unanswered, which may have affected the veracity of the responses.

- Schools in rural areas gave us a higher response rate compared to urban schools, which might have skewed the results.
- Despite a high response rate in our study (84%), children who were six years old showed a low response rate in comparison with other age groups.
- The fact that the study population was not randomly selected could have introduced bias to the study. The Department of Basic Education selected the schools that participated in the study and the first children in each age and gender group who brought back signed consent forms was selected to participate in the study.
- Recall bias may be reported due to the subjective nature of the dietary preferences in children.
- A 24-hour recall questionnaire tends to underestimate the amount of an individual's intake because of day to day variation; however, the selection of this method was justified based on a review of the existing literature, particularly on national studies, with consideration of the unique characteristics of the South African population, and on the basis that it is the best method to use on children (Oldewage-Theron *et al.*, 2011).

The young children between the ages of six and nine found it difficult to accurately recall the food that had been consumed in the previous 24 hours. In this regard, the parents were interviewed telephonically to verify intakes if the children were unsure about what they ate.

3 Recommendations

- Urbanisation in Limpopo is occurring rapidly. Designing context-specific interventions for these changing demographics is increasingly necessary, especially in terms of malnutrition.
- South Africa needs to focus on programs targeting the first 1000 days of life, especially taking into consideration the pre-conception period and expecting mothers with low BMIs, as this may improve the nutritional status of children.
- Policy makers and program designers should broaden their perspectives on nutrition interventions to address and change the environmental and behavioural factors. This can include promoting vegetable gardens at schools and in communities, introducing three meals per day in schools as a standard,

including a Physical Education time slot, as well as fun physical activities for children to partake in during school hours.

- Increasing education (of parents, children and educators), alleviating poverty, improving feeding practices, and preventing and treating infections are all required to assist in combating malnutrition in this province.
- Local communities must develop partnerships with schools, healthcare providers, local businesses and community groups, and strengthen the mission to create opportunities for healthy living and reducing malnutrition.
- Further investigation about feeding schemes should be done to check if it is reducing malnutrition and wether it should be extended to include weekends and holidays.

References

- Abubakar, A., Uriyo, J., Msuya, S.E., Swai, M. and Stray-Pederson, B. 2012. Prevalence and risk factors for poor nutritional status among children in the Kilimanjaro region of Tanzania. *International Research and Public Health.* 9: 3506 – 3518.
- Ahmed, M., Shah, K. and Kshirsagar, V.Y. 2016. Prevalence and risk factor obesity in urban and rural school going children of Karad taluka, Maharashtra, India. *International Journal of Contemporary Paediatrics*. **3:** 2349 – 2391.
- Aly, G.S., Shaalan, A.H., Mattar, M.K., Ahmed, H.H., Zaki, M.E. and Abdallah, H.R.
 2014. Oxidative stress status in nutritionally stunted children. *Egyptian Paediatric Association Gazette*. 62: 28 – 33.
- Amadou, A., Ferrari, P., Muwonge, R., Moskal, A., Biessy, C., Romieu, I., et al, 2013.
 Overweight, obesity and risk of premenopausal breast cancer according to ethnicity. A systematic review and dose-response meta-analysis, Obesity Reviews. 97:135 146.
- An, R., Patel, D., Segal, D. and Sturm, R. 2013. Eating better for less: A national discount program for healthy food purchases in South Africa. *American Journal* of Health Behaviour. **37**: 56 – 61.
- Anderson, S.E., Economos, C.D. and Must, A. 2008. Active play and screen time in US children aged 4 to 11 years in relation to socio-demographic and weight status characteristics: A nationally representative cross-sectional analysis. BMC *Public Health.* 8: DOI: 10.1186/1471-2458-8-366.
- Angood, C., Khara, T., Dolan, C., Berkley, J.A. and Wast technical interest group.
 2016. Research priorities on the relationship between wasting and stunting.
 PLos ONE. 11: DOI: 10.1371/journal.pone.0153221.

Anrig, C.D.C. 2003. The obese child. *Dynamic Chiropractic*. **21**: 27 – 31.

Aouadi, M., Tencerova, M., Vangala, P., Yawe, J.C., Nicoloro, S.M., Amano, S.U., Cohen, J.L. and Czech, M.P. 2013. Gene silencing in adipose tissue macrophages regulates whole-body metabolism in obese mice. *Proceedings of the National Academy of Sciences of the United States of America*. **110**: 8278 – 8283.

- Aras, R.Y. 2013. Is maternal age risk factor for low birth weight? *Archives of Medicine* and Health Sciences. **1:** 33 – 37.
- Arifeen, A.E., Black, R.E., Caulfield, L.E., Antelman, G., Baquie, A.H. and Nahar, Q. 2000. Infant growth patterns in the slums of Dhaka in relation to birth weight, intrauterine growth retardation, and prematurity. *American Journal of Clinical Nutrition*. **72**: 1010 – 1017.
- Armstrong, M.E.G., Lambert, M.I. and Lambert, E.V. 2006. Obesity and overweight in South African primary school children. The Health of the Nation Study. *South African Medical Journal*. **96:** 439 – 444.
- Arndt, C., Davies, R. and Thurlow, J. 2018. Urbanization, Structural Transformation and Rural-Urban Linkages in South Africa. 1 – 14.
- Arsenault, J.E. and Brown, K.H. 2017. Dietary protein intake in young children in selected low-income countries is generally adequate in relation to estimated requirements for healthy children, except when complementary food intake is low. *The Journal of Nutrition*. **147:** 932 – 939.
- Asfaw, M., Wondaferash, M., Taha, M. and Dube, L. 2015. Prevalence of under nutrition and associated factors among children aged between six to fifty nine months in Bule Hora district, South Ethiopia. *BMC Public Health.* 15: DOI.org/10.1186/s12889-015-1370-9.
- Atkin, L.M. and Davies, P.S.W. 2000. Diet composition and body composition in preschool children. *American Journal of Clinical Nutrition*. **72:** 15 21.
- Atsu, B.K., Guure, C.and Laar, A.K. 2017. Determinants of overweight with concurrent stunting among Ghanaian children. *BMC Pediatrics*. **17:** 1 – 12 DOI 10.1186/s12887-017-0928-3
- Badawi, N.E., Barakat, A.A., Sherbini, S.A.E., and Fawzy, H.M. 2013. Prevalence of overweight and obesity in primary school children in Port Said city. Egyptian Pediatric Association Gazette. 61(1): 31 – 36.
- Bahreynian, M., Qorbani, M., Khaniabadi, B.M., Motlagh, M.E., Safari, O., Asayesh, H. and Kelishadi, R. 2017. Association between Obesity and Parental Weight Status in Children and Adolescents. *Journal of Clinical Research in Pediatric Endocrinology*. 9(2): 111 117 DOI: 10.4274/jcrpe.3790.
- Baird, J., Fisher, D., Lucas, P., Kleijnen, J., Roberts, H. and Law, C. 2005. Being big or growing fast: systematic review of size and growth in infancy and later

obesity. *The British Medical Journal*. **331(7522):** 929. doi: 10.1136/bmj.38586.411273.E0.

- Berkman, D.S., Lescano, A.G., Gilman, R.H., Lopez, S.L. and Black, M.M. 2002.
 Effects of stunting, diarrhoeal disease, and parasitic infection during infancy on cognition in late childhood: A follow-up study. *Lancet.* 359: 564 571.
- Berry, L., Hall K, Hendricks M. Child Health: Nutrition. In: South African Child Gauge 2009/2010, Kibel M, Lake L, Pendlebury S, Smith C (eds.) (2010). Cape Town: Children's Institute, University of Cape Town ISBN: 978-0-9814320-4-5
- Berry, L., Hall, K. and Hendricks, M. 2010. Child health: Nutrition. South African Child Gauge. 120 123.
- Biswas, S. and Bose, K. 2010. Sex differences in the effect of birth order and parents educational status on stunting: A study on Bengalee preschool from eastern India. *Journal of Comparative Human Biology*. **61:** 272 – 276.
- Black, R.E., Allen, L.H., Bhutta, Z.A., Caulfield, L.E., de Onis, M., Ezzati, M., Mathers, C., Rivera, J. and Maternal and Child Undernutrition Study Group. 2008.
 Maternal and child undernutrition: Global and regional exposures and health consequences. US National Library of Medicine National Institute of Health.
 371: 243 260.
- Black, R.E., Victora, C.G., Walker, S.P., Bhutta, Z.A., Christian, P., de Onis, M., Ezzati, M., Grantham-McGregor, S., Katz, J., Mortorell, R., Uauy, R. and the Maternal and Nutrition Study Group. 2013. Maternal and child undernutrition and overweight in low-income and middle-income countries. *Lancet.* 382: 427 451.
- Borodulin, K., Harald, K., Jousilahti, P., Laatikainen, T., Mannisto, S., Vartiainen, E.,
 2016. Time trends in physical activity from 1982 to 2012 in Finland.
 Scandinavian Journal of Medicine & Science Sports. 26(1): 93 100.
- Bosch, J., Stradmeijer, M., and Seidell, J. 2004. Psychosocial characteristics of obese children/youngsters and their families: Implications for preventive and curative interventions. *Patient Education and Counselling*. **55**: 353 362.
- Bove, I., Miranda, T., Campoy, C., Uauy, R. and Napol, M. 2012. Stunting, overweight and child development impairment go hand in hand as key problems of early infancy: Uruguayan case. *Early Human Development*. **88:** 747 – 751.

- Brabin, B.J. and Coulter, J.B.S. 2003. Nutrition-associated disease. In: Cook GC, Zumla AI, editors. Manson's tropical diseases. London. UK: Saunders. 561 – 580.
- Bray, G.A. and Bellanger, T. 2006. Eidemiology, trends, and morbidities of obesity and the metabolic syndrome. *Endocrine*. **29:** 109 117.
- Brook CGD (1971): Determination of body composition of children from skinfold measurements. *Archies of Disease in Childhood*. **46:** 182 184.
- Bryant-Waugh, R., Markham, L., Kreipe, R.E. and Walsh, B.T. 2010. Feeding and eating disorders in childhood. *International Journal of Eating Disorders*. 43: 98 111.
- Brylinsky, J., Jody, A. and Moore, J.C. 1994. The Identification of Body Build Stereotypes in Young Children. *Journal of Research in Personality*. **28(2)**:170 – 181 DOI: 10.1006/jrpe.1994.1014.
- Buttenheim, A.M. 2009. The sanitation environment in urban slums: Implications for child health. *Population and Environment*. **30:** 26 47.
- Buyukinan, M., Ozen, S., Kokkun, S. and Saz, E.U. 2012. The relation of vitamin D deficiency with puberty and insulin resistance in obese children and adolescents. *Journal of Paediatric Endocrinology and Metabolism*. **25:** 83 87.
- Carruth, B.R. and Skinner, J.D. 2001. The role of dietary calcium and other nutrients in moderating body fat in preschool children. *International Journal of Obesity*.
 25: 559 566.
- Caulfield, L.E., Richard, S.A., Rivera, J.A., Musgrove, P. and Black, R.E. 2004. "Stunting, wasting, and micronutrient deficiency disorders". *Disease Control Priorities in Developing Countries*. 2nd edition. World Bank Editor. 551 – 569.
- Cawley, J. 2010. The Economics of Childhood Obesity. *Health Affairs*. 29: 364 371.
- Chamieh, M.C., Moore, H.J., Summerbell, C., Tamim, H., Sibai, A.M. and Nahla Hwalla, N. 2015. Diet, physical activity and socio-economic disparities of obesity in Lebanese adults: Findings from a national study. *BMC Public Health*. 15: 279. DOI: 10.1186/s12889-015-1605-9.
- Chen, H., Wang, Y., Ma, L., Zhao, J., Li, Y., Li, M. 2012. Long-term high animal protein diet reduces body weight gain and insulin secretion in diet-induced obese rats. *Journal of the Science of Food and Agriculture*. **92**: 38 – 43.
- Chirita-Emandi, A., Barbu, C.G., Cinteza, E.E., Chesaru, B.I., Gafencu, M., Mocanu, V., Pascanu, I.M., Tatar, S.A., Balgradean, M., Dobre, M., Fica, S.V., Ichim,

G.E., Pop, R. and Puiu, M. 2016. Overweight and underweight prevalence trends in children from Romania – pooled analysis of cross sectional studies between 2006 and 2016. *Obesity Facts*. **9:** 206 – 220.

- Christopher, D., John, B., Watkins, W. and Walker, A. 2008. Nutrition in pediatrics: basic science, clinical application. 4th edition. Decker, B. C., Hamilton, Canada.
- Chung, S. 2015. Body mass index and body composition scaling to height in children and adolescent. *Annals of Pediatric Endocrinology Metabolism*. **20(3):** 125 – 129 DOI: 10.6065/apem.2015.20.3.125.
- Cole, T.M., Bellizzi, M.C., Flegal, K.M. and Dietz, W.H. 2000. Establishing a standard definition for child overweight and obesity worldwide: International survey. *British Medical Journal.* **320**: 1 – 6.
- Cornette, R. 2008. The Emotional Impact of Obesity on Children. World Views on Evidence Based Nursing. 5: 136 141. DOI.org/10.1111/j.1741-6787.2008.00127.x.
- Corsi, D.J., Mejía-Guevara, I. and Subramanian, S.V. 2015. Risk factors for chronic undernutrition among children in India: Estimating relative importance, population attributable risk and fractions. *Social Science and Medicine*. DOI: 10.1016/j.socscimed.2015.11.014.
- Cort, M.A., Gwebub, E.T., Tull, E.S., Cox, N.A., Modise, T. 2013. The differential gender effect of internalized racism on abdominal obesity in KwaZulu-Natal, South Africa. *The Social Science Journal*. **50:** 557 – 564.
- Crowther, N.J. and Norris, S.A. 2012. The current waist circumference cut point used for the diagnosis of metabolic syndrome in sub-Saharan African women is not appropriate. *PLoS One*. **7(11):** DOI: 10.1371/journal.pone.0048883.
- Crujeiras, A.B., Carreira, M.C., Cabia, B., Andrade, S., Amil, M. and Casanueva, F.F.
 2015. Leptin resistance in obesity: An epigenetic landscape. *Life Science*. 140: 57 63.
- Cummings, D.E. and Schwartz, M.W. 2003.Genetics and pathophysiology of human obesity. *Annual Review of Medicine*. **54:** 453 471. DOI: 10.1146/annurev.med.54.101601.152403.
- Curtner-Smith, M., Sofo, S., Chouinard, J. and Wallace, S. 2007. Health-promoting physical activity and extra-curricular sport. *European Physical Education Review*. **13(2):**131 144.

- Dabone, C., Delisle, H.F. and Receveur, O. 2011. Poor nutritional status of school children in urban and peri-urban areas of Ougadou (Burkina Faso). *Nutrition Journal*. **10:** DOI: 10.1186/1475-2891-10-34.
- Darnton-Hill, I. and Mkparu, U.C. 2015. Micronutrients in pregnancy in low- and middle-income countries. *Nutrients*. **7**: 1744 1768.
- Das, S. and Bose, K. 2011. Prevalence of thinness among Santal preschool children using new body mass index cut-off points. *Journal of Anthropology*. 2011: DOI.org/10.1155/2011/602782.
- Davies, P.S., Day, J.M. and Lucas, A. 1991. Energy expenditure in early infancy and later body fatness. *International Journal of Obesity*. **15:** 727 731.
- de Onis, M. and Branca, F. 2016. Childhood stunting: a global perspective. *Maternal* & *Child Nutrition*. **12:** 12 26.
- de Onis, M., Blössner, M. and Borghi, E. 2012. Prevalence and trends of stunting among pre-school children, 1990-2020. *Public Health Nutrition*. 15(1):142 148. DOI: 10.1017/S1368980011001315.
- Dietz, W.H. 1994. Critical periods in childhood for the development of obesity. *American Journal of Clinical Nutrition*. **59(5):**955 – 959 DOI: 10.1093/ajcn/59.5.955.
- Dillingham, R. and Guerrant, R.L. 2004. Childhood stunting: Measuring and stemming the staggering costs of inadequate water and sanitation. *The Lancet.* 363: 94 – 95.
- Dinsa, G.D., Goryakin, Y., Fumogalli, E. and Suhrcke, M. 2012. Obesity and socio economic status in developing countries. *Systematic Review*. **13**:1067 1079.
- Dixon, L.B., Pellizzon, M.A., Jawad, A.F. and Tershakovec, A.M. 2005. Calcium and diary intake and measures of obesity in hyper- and normo-cholesterolemic children. *Obesity Research*. **13**: 1727 – 1738.
- Doak, C.M., Visscher, T.L., Renders, C.M. and Seidell, J.C. 2006. The prevention of overweight and obesity in children and adolescents: a review of interventions and programmes. *Obesity review*. 7(1):111 – 136.
- Driul, L., Cacciaguerra, G., Citossi, A., Martina, M.D., Peressini, L. and Marchesoni,
 D. 2008. Prepregnancy body mass index and adverse pregnancy outcomes.
 Archives of Gynecology and Obstetrics. 278(1):23 26.

- Dubern, B., Bisbis, S., Talbaoui, H., Le Beyec, J., Tounian, P., Lacorte, J.M. and Clément, K. 2007. Homozygous null mutation of the melanocortin-4 receptor and severe early-onset obesity. *Journal of Pediatrics*.**150**: 613 – 617.
- Duh, H.I., Grubliauskiene, A. and Dewitte, S. 2016. Pre-exposure to food temptation reduces subsequent consumption: A test of the procedure with a South-African sample. *Appetite.* **96**: 636 – 641.
- Emery, P.W. 2005. Metabolic changes in malnutrition. *Malnutrition*. **9:**1029 1034.
- Faber, M. and Wenhold, F. 2007. Nutrition in contemporary South Africa. *Water SA*. **33:** 393 400.
- FAO (Food and Agriculture Organization); International Fund for Agricultural Development, World Food Program. 2015. The State of Food Insecurity in the World 2015. Strengthening the enabling environment for food security and nutrition. Rome (Italy): FAO - Food and Agriculture". 2015. www.fao.org/hunger/en/.
- Farooqi, I.S. 2011. Genetic, molecular and physiological insights into human obesity. *European Journal of Clinical Investigation*. **41(4):** 451-455. DOI: 10.1111/j.1365-2362.2010.02468.x.
- Farooqi, I.S. and O'Rahilly, S. 2005. Monogenic obesity in humans. *Annual Review of Medicine*. **56:** 443 458. DOI: 10.1146/annurev.med.56.062904.144924.
- Fernald, L.C. and Neufeld, L.M. 2007. Overweight with concurrent stunting in very young children from rural Mexico: Prevalence and associated factors. *European Journal of Clinical Nutrition*. **61:** 623 – 632.
- Finkelstein, E.A., Khavjou, O.A., Thompson, H., Trogdon, J.G., Pan, L., Sherry, B., and Dietz, W. 2012. Obesity and severe obesity forecasts through 2030. *American Journal of Preventive Medicine*. **42:** 563 – 570.
- Flegal, K.M., Caroll, M.D., Kit, B.K. and Ogden, C.L. 2012. Prevalence of obesity and trends in the distribution of body mass index among US adults, 1990 – 2010. US National Library of Institute of Health. 1,301: 491 – 497
- Flegal, K.M., Tabak, C.J. and Ogden, C.L. 2006. Overweight in children: definitions and interpretation. *Health Education Research*. **21(6)**: 755 – 760. DOI.org/10.1093/her/cyl128.
- Franz, M.J. 2003. So many nutrition recommendations-contradictory or compatible? *Diabetes Spectrum*. **16:** 56 – 63.
- Friedman, J.M. 2003. A war on obesity, not the obese. Science. 299: 856 858.

- Galtier-Dereure, F., Boegner, C. and Bringer, J. 2000. Obesity and pregnancy: complications and cost. *The American Journal of Clininical Nutrition*. **71(5 Suppl)**:1242S – 8S. DOI: 10.1093/ajcn/71.5.1242s.
- Garcia, O.P., Ronquillo, D., Caamano, M.C., Martinez, G., Camacho, M., Lopez, V. and Rosado, J.L. 2013. Zinc, iron and vitamins A, C and E are associated with obesity, inflammation, lipid profile and insulin resistance in Mexican school aged children. *Nutrients*. **5**: 5012 5030.
- Gopinath, B., Subramanian, I., Flood, V.M., Baur, L.A., Pfund, N., Burlutsky, G. and Mitchell, P. 2012. Relationship between breast-feeding and adiposity in infants and pre-school children. *Public Health Nutrition*. **15:** 1639 – 1644.
- Govender, L., Pillay, K., Siwela, M., Modi, A. and Mabhaudhi, T. 2017. Food and nutrition insecurity in selected rural communities of KwaZulu-Natal, South Africa linking human nutrition and agriculture. *International Journal of Environmental Research and Public Health.* 14: DOI: 10.3390/ijerph14010017.
- Griffiths, P.L., Johnson, W., Cameron, N., Pettifor, J.M. and Norris, S.A. 2013. In urban South Africa, 16 year old adolescents experience greater health equality than children. *Economics and Human Biology*. **11**: 502 – 514.
- Guo, H., Zeng, X., Zhuang, Q., Zheng, Y. and Chen, S. 2015. Intervention of childhood and adolescents' obesity in Shantou city. *Obesity Research and Clinical Practice.* **9**: 357 – 364.
- Guo, M. and Ahmad, S. 2014. Human Milk Biochemistry and Infant Formula Manufacturing Technology. Food Science, Technology and Nutrition. 141 – 171.
- Gupta, N., Goel, K., Shah, P. and Misra, A. 2013. Childhood obesity in developing countries: Epidemiology, determinants, and prevention. *Endocrine Reviews*.
 33: 48 70.
- Han, E. and Powell, L.M., 2013. Consumption patterns of sugar-sweetened beverages in the United States. *Journal of Academy of Nutrition and Dietetics*. **113**: 43 – 53.
- Harder, T., Bergmann, R., Kallischnigg, G. and Plagemann, A. 2005. Duration of breastfeeding and risk of overweight: A meta-analysis. *America Journal of Epidemiology.* **162**: 397 – 403.

- Harding, K.L., Aguayo, V.M. and Webb, P. 2018. Factors associated with wasting among children under five years old in South Asia: Implications for action. *PLos ONE*. **13 (7):** e0198749. https://doi.org/10.1371/journal.pone.0198749.
- Hasan, I., Zulkifle, M. and Ansari, A.H. 2013. Prevalence of stunting among school children of government Urdu higher primary schools in Azad Nagar and its surrounding area, Bangslore. *International Journal of Medicine and Medical Sciences.* **3**: 304 – 310.
- He, L., Ren, X., Chen, Y., Jin, Y., Pan, R., Wei, N., Qiu, S., Wei Lu, W., Ding, L., Guo,
 D. Wang, L., Nie, Z and Yao, Y. 2014. Prevalence of overweight and obesity among primary school children aged 5 to 14 years in Wannan area, China. *Nutrición hospitalaria*. **30(4):**776 – 781. DOI:10.3305/nh.2014.30.4.7693.
- Herrador, Z., Sordo, L., Gadisa, E., Moreno, J., Nieto, J., Benito, A., Aseffa, A., Canavate, C. and Custodio, E. 2014. Cross-sectional study on malnutrition and associated factors among school aged children in rural and urban settings of Fogera and Libo Kemkem districts, Ethiopia. *PLos ONE*. **9:** DOI: 10. 1371/journal.pone.0105880.
- Hoekman, R., Hoenderkamp, K. and van der Poel, H. 2013. Sport facilities in the picture. *Arko Sports Media*. **36:** DOI: 10.1080/02614367.2016.1182201
- Huey, S.L. and Mehta, S. 2016. Stunting: The need for application of advances in technology to understand a complex health problem. *EBioMedicine*. 6: DOI: 10.1016/j.ebiom.2016.03.013.
- Hughes, M.M., Black, R.E. and Katz, J. 2017. 2500-g Low Birth Weight Cutoff: History and Implications for Future Research and Policy. *Maternnal and Child Health Journal.* 21(2): 283 – 289 DOI: 10.1007/s10995-016-2131-9.
- Islam, A. and Biswas, T. 2015. Chronic stunting among under 5 children in Bangladesh: A situation analysis. *Advances of Paediatric Research*. **2:** DOI: 10. 12715/april.2015.2.18.
- Jackson-Leach, R. and Lobstein, T. 2006. Estimated burden of paediatric obesity and co-morbidities in Europe. Part 1. *International Journal of Paediatric Obesity*. **1**: 26 32.
- Jiang, J., Resenqvist, U., Wang, H., Greiner, T., Ma, Y. and Toschke, A.M. 2006. Risk factors for overweight in 2 6 year old children in Beijing, China. *International Journal of Paediatric Obesity.* **1**: 103 108.

- Jiang, Y., Su, X., Wang, C., Zhang, L., Zhang, X., Wang, L. and Cui, Y. 2015. Prevalence and risk factors for stunting and severe stunting among children under three years old in mid-western rural areas of China. *Child Care Health Development.* **41(1):** 45 – 51.DOI: 10.1111/cch.12148.
- Jinabhai, C.C., Taylor, M. and Sullivan, K.R. 2003. Implications of the prevalence of stunting, overweight and obesity amongst South African primary school children: a possible nutritional transition? *European Journal of Clinical Nutrition*. 57: 358 – 365.
- Jinabhai, C.C., Taylor, M., Coutsoudis, A., Coovadia, H.M., Tomkins, A.M and Sullivan, K.R. 2001. A health and nutritional profile of rural school children in KwaZulu-Natal, South Africa. *Annals of Tropical Paediatrics*. **25(1):** 3 15.
- Jing, H., Xu, H., Wan, J., Yang, Y, Ding, H., Chen, M., Li, L., Lv, P., Hu, J, and Yang, J. 2014. Effect of Breastfeeding on Childhood BMI and Obesity. *Medicine*. 93(10): DOI: 10.1097/MD.000000000000055.
- Johnson, B. and Christensen, L. 2012. Educational Research: Quantitative, Qualitative and Mixed Approaches. 4th edition. Washington DC: SAGE Publications.
- Kang, Y.E., Kim, J.M., Joung, K.H., Lee, J.H., You, B.R., Choi, M.J., Ryu, M.J., Ko, Y.B., Lee, M.A., Lee, J., Ku, B.J., Shong, M., Lee, K.H. and Kim, H.J. 2016. The roles of adipokines, proinflammatory cytokines, and adipose tissue macrophages in obesity-associated insulin resistance in modest obesity and early metabolic dysfunction. *PLoS ONE*. 11: DOI:10.1371/journal.pone.0154003.
- Karnik, S. and Kanekar, A. 2012. Childhood obesity: A global public health crisis. *International Journal Prevention Medicine*. **3**: 1 – 7.
- Karpe, F. and Pinnick, N.E. 2015. Biology of upper-body and lower body adipose tissue link to whole body phenotypes. US National Library of Medicine National Institute of Health. 11: 90 – 100.
- Khan, A., Khan, S.U., Khan, S., Zia-ul-islam, S., Baber, N.K. and Khan, M. 2018.
 Nutritional complications and its effects on human health. *Journal of Food Science and Nutrition*. 1: 17 20.
- Kimani-Murage, E.W., Kahn, K., Pettifor, J.M., Tollman, S.M., Dunger, D.B., Gomez-Olive, X.F. and Norris, S.A. 2010. The prevalence of stunting, overweight and

obesity, and metabolic disease risk in rural South African children. *BMC Public Health*. **10**: DOI: 10.1186/1471-2458-10-158.

- Kleynhans, I.C., MacIntyre, U.E. and Albertse, E.C. 2006. Stunting among young black children and the socio-economic and health status of their mothers/caregivers in poor areas of rural Limpopo and urban Gauteng – the NutriGro study. South African Journal of Clinical Nutrition. **19:** 163 – 172.
- Koch, C.E., Lowe, C, Pretz, D., Steger, J., Williams, L.M. and Tups, A. 2013. High-fat diet induces leptin resistance in leptin-deficient mice. *Journal of Neuroendocrinology*. 26: 58 – 67.
- Koirala, M., Khatrib, R.B., Khanal, V., Amatya, A. 2015. Prevalence and factors associated with childhood obesity/overweight of private school children in Nepal. Obesity Research and Clinical Practice. 9: 220 – 227. Demographic Research. 18: 565 - 594.
- Kravdal, Ø. and Kodzi, I. 2011. Children's stunting in sub-Saharan Africa: Is there an externality effect of high fertility? *Demographic research*. **25:** 565 594.–594.
- Krishna, A., Fink, G., Berkman, L.F. and Subramanian. 2016. Short and long run associations between birth weight and children's height. *Economics and Human Biology*. **21**: 156 166.
- Kruger, G., Pienaar, A.E., Coetzee, D. and Kruger, S.H. 2014. Prevalence of stunting, wasting and underweight in grade 1 learners: The NW child study. *Health South Africa*. **19**: DOI: 10.4102/hsag.v19i1.750.
- Kruger, H.S., Puoane, T., Senekal, M. and Van der Merwe, M. 2005. Obesity in South Africa: Challenges for government and health professionals. *Public Health Nutrition.* 8: 491 500.
- Kwon, H. and Pessin, J.E. 2013. Adipokines mediate inflammation and insulin resistance. *Endocrinology*. **4:** doi: 10.3389/fendo.2013.00071.
- Kyallo, F., Mokokha, A. and Mwangi, A.M. 2013. Overweight and obesity among public and private primary school children in Nairobi, Kenya. *Health.* **5:** 85 90.
- Kyei, K.A., Netshikweta, M.L. and Spio, K. 2012. Breastfeeding in the Vhembe district of Limpopo Province, South Africa: Duration and Factors. *Studies on Ethno-Medicine*. 8: 317 – 324.
- Labadarios, D., Steyn, N.P., Maunder, E., MacIntryre, U., Gericke, G., Swart, R., Huskisson, J., Dannhauser, A., Vorster, H.H. and Nesmvuni, A.E. 2005. The

National Food Consumption Survey (NFCS): South Africa, 1999. *Public Health Nutrition*. **8:** 533–543. DOI: 10.1079/PHN2005816.

- Leddy, M.A., Power, M.L. and Schulkin, J. 2008. The impact of maternal obesity on maternal and foetal health. *The Reviews in Obstetrics and Gynaecology*. **1**: 170 178.
- Lee, H.A., Kim, Y.J., Lee, H., Gwak, H.S., Park, E.A., Cho, S.J., Oh, S.Y., Ha, E.H., Kim, H.S. and Park, H. 2013. Association of vitamin D concentrations with adiposity indices among preadolescent children in Korea. *Journal of Paediatric Endocrinology and Metabolism.* 26: 849 – 854.
- Lee, J., Houser, R.F., Must, A. and de Fulladolsa, P.P. 2012. Socioeconomic disparities and the familial coexistence of child stunting and maternal overweight in Guatemala. *Economics and Human Biology*. **10**: 232 241.
- Lesiapeto, M.S., Smuts, C.M., Hanekom, S.M., Du Plessis, J. and Faber, M. 2010. Risk factors of poor anthropometric status in children under five years of age living in rural districts of the Eastern Cape and Kwazulu – Natal provinces, South Africa. South African Journal of Clinical Nutrition. 23: 202 – 207.
- Li, S., Shin, H.J., Ding, E.L. and van Dam, R.M. 2009. Adiponectin levels and risk of type 2 diabetes: A systematic review and meta-analysis. *The Journal of the American Medical Association*. **302:** 179 188.
- Lifshitz, F. 2008. Obesity in Children. *Journal of Clinical Research in Pediatric Endocrinology*. **1(2):** 53 – 60 DOI: 10.4008/jcrpe.v1i2.35.
- Lillefosse, H.H., Clausen, M.R., Yde, C.C., Ditlev, D.B., Zhang, X. and Du, Z.Y. 2014. Urinary loss of tricarboxylic acid cycle intermediates as revealed by metabolomics studies: An underlying mechanism to reduce lipid accretion by whey protein ingestion? *Journal of Proteome Research*.**13**: 60 – 70.
- Lobstein, T. and Frelut, M.L. 2003. Prevalence of overweight among children in Europe. *Obesity Reviews*. **4:** 195 200.
- Lohman, T.G., Slaughter, M.H., Boileau, R.A., Bunt, J and Lussier, L. 1984. Bone mineral measurements and their relation to body density in children, youth and adults. *Human Biology*. **56:** 667 679.
- Łuczyński, W., Głowińska-Olszewska, B. and Bossowski, A. 2016. The influence of clinical and genetic factors on the development of obesity in children with type 1 diabetes. *Diabetes Metabolism Research and Reviews*. 32: 666 671.

- Luke, A and Cooper R.S., 2013. Physical activity does not influence obesity risk: time to clarify the public health message. *Internation Journal of Epidemiology*. 42(6): 1 – 6.
- Lutter, C.K. and Morrow, A.L. 2013. Protection, promotion, and support and global trends in breastfeeding. *Advances in Nutrition*. **4(2):** 213 219. DOI: 10.3945/an.112.003111
- Mamabolo, R.L., Alberts, M., Mbenyane, G.X., Steyn, N.P., Nthangeni, N.G., Delemarre-Van De Waal, H.A. and Levitt, N.S. 2004. Feeding practices and growth of infants from birth to 12 months in the central region of the Limpopo Province of South Africa. *Nutrition.* **20**: 327 – 333.
- Marfell-Jones, M., Olds, T., Stewart, A. and Carter, J.E.L. 2006. International standards for anthropometric assessment. University of South Australia, Australia. 1 37.
- Mascie-Taylor, C.G. and Goto, R. 2007. Human variation and body mass index: A review of the universality of BMI cut-offs, gender and urban-rural differences and secular changes. *Journal of Physiological Anthropology*. **26**: 109 112.
- McCullough, N., Muldoon, O. and Dempster, M. 2009. Self-perception in overweight and obese children: A cross-sectional study. *Child: Care, Health and Development.* **35**. DOI.org/10.1111/j.1365-2214.2008.00924.
- Mchiza, Z.J., Parker, W. Makoae, M., Sewpaul, R., Kupamupindi, T and Labadarios,
 D. 2015. Body image and weight control in South Africans 15 years or older:
 SANHANES-1. BMC Public health. 15: doi.org/10.1186/s12889-015-2324-y.
- Mei, Z., Grummer-Strawn, L.M., Pietrobelli, A., Goulding, A., Goran, M.I. and Dietz, W.H. 2002. Validity of body mass index compared with other body-composition screening indexes for the assessment of body fatness in children and adolescents. *American Journal of Clinical Nutrition*.**75**: 978 – 985.
- Meko, L.N.M, Slabber-Stretch, M., Walsh, C.M., Kruger, S.H.,and Nel, M. 2015. School environment, socioeconomic status and weight of children in Bloemfontein, South Africa. *African Journal of Primary Health Care & Family Medicine*. **7(1):** DOI: 10.4102/phcfm.v7i1.751.
- Mesfin, F., Berhane, Y. and Worku, A. 2015. Prevalence and associated factors of stunting among primary school children in Eastern Ethiopia. *Nutrition and Dietary Supplements.* **7:** 61 – 68.

- Meshram, I.I., Mallikharjun, Reddy, C.G. Ravindranath, M., Kumar, S.S., Sreerama, K.K, Hari, K.R., Venkaiah, K. and Laxmaiah, A. 2016. Prevalence of Under Nutrition and its Predictors among Under 5 Year Children in Surat Region, Gujarat, India. *Journal of Clinical Nutrition & Dietetics*. 2: DOI: 10.4172/2472-1921.100009.
- Michaelsen, K.F, Hoppe, C., Roos, N., Kaestel, P., Stougaard, M., Lauritzen, L., Mølgaard, C., Girma, T and Friis, H. 2009. Choice of foods and ingredients for moderately malnourished children 6 months to 5 years of age. SAGE Journal: Food and Nutrition Bulletin. **30:** DOI.org/10.1177/15648265090303S303
- Mitch, P. 2014. Body dissatisfaction and males: A conceptual model. *Scholarly Horizons: University of Minnesota, Morris Undergraduate Journal.* **1:** 1 – 25.
- Monyeki, K.D., Cameron, N. and Getz, B. 2000, 'Growth and nutritional status of rural South African children 3-10 years old: The Ellisras growth study', *American Journal of Human* Biology.**12(1):** 42 – 49.
- Monyeki, M.A., Awotidebe, A., Strydom, G.L., de Ridder, J.H., Mamabolo., R.L. and Kemper, H.C.G. 2015. The challenges of underweight and overweight in South African children: are we winning or losing the battle? A systematic review. *International Journal of Environmental Research and Public Health.* **12**: DOI:10.3390/ijerph120201156.
- Moore, S.A., Hallsworth, K., Plötz, T., Ford, G.A., Rochester, L. and Trenell, M.I. 2013. Physical Activity, Sedentary Behaviour and Metabolic Control following Stroke:
 - A Cross-Sectional and Longitudinal Study. *PLoS ONE*. DOI.org/10.1371/journal.pone.0055263. **8:** 1 6.
- Moschonis, G., Chrousos, G.P., Lionis, C., Mougios, V., Manios, Y. and Healthy Growth Study group. 2012. Association of total body and visceral fat mass with iron deficiency in preadolescents: The Healthy Growth Study. *British Journal of Nutrition*. **108**: 710 – 719.
- Moselakgomo, V.K., Monyeki, M.A. and Toriola. A.L. 2015. Relationship between physical activity and risk factors of body weight disorders among South African primary school children. *Biomedical research*. **26(4)**: 730 – 738.
- Mosha, T.C and Fungo, S. 2010. Prevalence of overweight and obesity among children aged 6-12 years in Dodoma and Kinondoni Municipalities, Tanzania. *Tanzania Journal of Health Research*. **12:** 1 – 13.

- Motadi, S.A., Mbenyane, X.G., Mbhatsani, H.V., Mabapa, N.S. and Mamabolo, R.L.
 2015. Prevalence of iron and zinc deficiencies among preschool children ages
 3 to 5 y in Vhembe district, Limpopo province, South Africa. *Nutrition.* 31: 452 458.
- Mukuddem-Petersen, J. and Kruger, H.S. 2004. Association between stunting and overweight among 10 – 15-y-old children in the North West Province of South Africa: the THUSA BANA Study. *International Journal of Obesity and Related Metabolic Disorders.* 28: 842 – 851.
- Müller, M.J. and Geisler, C. 2017. From the past to future: from energy expenditure to energy intake to energy expenditure. *European Journal of Clinical Nutrition*. **71**: 358 – 364.
- Murungi, C.A. 2012. Stunting and children's attendance in early childhood education in Kenya. *Prime Journal of Social Sciences*. **1:** 6 10.
- Musaiger, A.O. 2011. Overweight and obesity in Eastern Mediterranean Region: Prevalence and possible causes. *Journal of Obesity*. **2011**: 1 – 183.
- National Health and Nutrition Examination (NHANES), 2010. Public Data General Release file. *National Centre for Health Statistics*.
- Navaneetham, K. and Dharmalingam, A. 2013. Demography and development:
 Preliminary interpretations of the 2011 census. *Economic and Political Weekly*.
 46: 13 16.
- Nordang, G.B.N., Busk, Ø. L., Tveten, K., Hanevik, H.I., Fell, A.K.M., Hjelmesæth, J., Holla, Ø.L. and Hertel, J.K. 2017. Next-generation sequencing of the monogenic obesity genes LEP, LEPR, MC4R, PCSK1 and POMC in a Norwegian cohort of patients with morbid obesity and normal weight controls. *Genetics and Metabolism.* **121**: 51 – 56.
- Ntyintyane, L.M., Panz, V.R., Raal, F.J. and Gill, G.V. 2006. Metabolic syndrome, undiagnosed diabetes mellitus and insulin resistance are highly prevalent in urbanised South African blacks with coronary artery disease. *Cardiovascular Journal of Africa*. **17(2):** 50 – 55.
- O'Dea, J.A. 2005. School based health education strategies for the improvement of body imagine and prevention of eating problems: an overview of safe and successful interventions. *Health Education*. **105**: 11 33.
- Ogden, C.L and Flegal, K.M. 2010. Changes in Terminology for Childhood Overweight and Obesity. National health statistics reports; no 25. 1 – 8.

- Ojiambo, R.M., Easton, C., Casajus, J.A., Konstabel, K., Reilly, J.J. and Pitsiladis, Y. 2012. Effect of urbanisation on objectively measured physical activity levels, sedentary time, and indices of adiposity in Kenyan adolescents. *Journal of Physical Activity and Health*. **9**:115 123.
- Oken, E. and Gillman, M.W. 2003. Fetal origins of obesity. *Obesity Research*. **11(4)**: 496 506 DOI: 10.1038/oby.2003.69.
- Oldewage-Theron, W., Napier, C. and Egal, A. 2011. Dietary fat intake and nutritional status indicators of primary school children in a low-income informal settlement in the Vaal region. *South African Journal of Clinical Nutrition*. **24:** 99 104.
- Olesfsky, J.M. and Glass, C.K. 2010. Macrophages, inflammation, and insulin resistance. *Annual Review of Physiology*. **72**: 219 246.
- Oliveira, D., Ferreira, F.S., Atougia, J., Fortes, F., Guerra, A. and Centeno-Lima, S. 2015. Infection by intestinal parasites, stunting, and anaemia in school-aged children from Southern Angola. *Research article plos one*. 1 10.
- Oninla, S.O., Owa, J.A., Onayade, A.A. and Taiwo, O. 2006. Comparative study of nutritional status of urban and rural Nigerian school children. *Journal of Tropical Paediatrics.* 53: 39 – 43.
- Osuji, C.U. Nzerem, B.A., Meludu, S., Dioka, C.E., Nwobodo, E. and Amilo, G.I. 2010. The prevalence of overweight/obesity and dyslipidemia amongst a group of women attending "August" meeting. *Nigerian Medical Journal*. **4:** 155 -159.
- Padez, C., Fernandes, T., Mauroa, I., Moreira, P. and Rosado, R. 2004. Prevalence of overweight and obesity in 7 – 9 year old Portuguese children: Trends in body mass index from 1970 – 2002. *American Journal of Human Biology*. **16:** 670 – 678.
- Panigrahi, A. and Das, S.C. 2014. Undernutrition and its correlates among children of 3 – 9 years of age residing in slum areas of Bhubaneswar, India. *The Scientific World Journal.* 2014: DOI:10.1155/2014/719673.
- Parizkova, J. and Hills, A. 2001. Childhood Obesity: Prevention and Management. CRC Press, Boca Rotan. 422.
- Pérez-Rodrigo, C., Escauriaza, B.A., Bartrina, J.A. and Allúe, I.P. 2015. Dietary assessment in children and adolescents: Issues and recommendations. *Nutricion Hospitalaria*. **31:** 76 83.
- Pérusse, L., Rankinen, T., Zuberi, A., Chagnon, Y.C., Weisnagel, S.J., Argyropoulos, G., Walts, B., Eric E. Snyder, E.E. and Bouchard, C. 2005. The Human Obesity

Gene Map: The 2004 Update. *Obesity A Research Journal*. **12:** 381-490 DOI.org/10.1038/oby.2005.50.

- Pinhas-Hamiel, O., Newfield, R.S., Koren I., Agmon, A., Lilos, A.P. and Phillip, M. 2003. Greater prevalence of iron deficiency in overweight and obese children and adolescents. *International Journal of Obesity*. **27**: 416 – 418.
- Pinheiro, E., Bogen, D.L., Hoxha, D., Ciolino, J.D. and Wisner, K.L. 2015. Sertraline and breastfeeding: review and meta-analysis. *Archives in Womens Mental Health.* **18(2):**139 – 146. DOI: 10.1007/s00737-015-0499-y.
- Popkin, B.M., Richards, M.K. and Montiero, C.A. 1996. Stunting is associated with overweight in children of four nations that are undergoing the nutrition transition. *The Journal of Nutrition*. **126(12)**:3009 3016.
- Prendergast, A.J. and Humphrey, J.H. 2014. The stunting syndrome in developing countries. *Paediatrics and International Child Health*. **34:** 250 265.
- Pulos, N., McGraw, M.H., Courtney, P.M. and Lee, G.C. 2014. Revision THA in obese patients is associated with high re-operation rates at short-term follow-up. *Arthroplasty.* 29: 209.
- Puoane, T., Steyn, K., Bradshaw, D., Laubscher, R., Fourie, J., Lambert, V. and Mbananga, N. 2002.Obesity in South Africa: The South African demographic and health survey. *Obese Research*. **10**: 1038 – 1048.
- Puoane, T., Tsolekile, L. and Steyn, N. 2010. Perceptions about body image and sizes among black African girls living in Cape Town. *Ethnicity & Disease*. **20**: 29 – 34.
- Rahman, S., Islam, M.T., Alam, D.S. 2014. Obesity and overweight in Bangladeshi children and adolescents: a scoping review. *BMC Public Health.* 14: DOI: 10.1186/1471-2458-14-70.
- Raina, S.K. 2013. Limitations of 24-hour recall method: Micronutrient intake and the presence of the metabolic syndrome. North American Journal of Medical Sciences. 5: DOI:10.4103/1947-2714.117329.
- Reilly, J.J., Armstrong, J., Dorosty, A.R., Emmett, P.M., Ness, A., Rogers, I., Steer, C. and Sherriff, A. 2005. Early life risk factors for obesity in childhood: cohort study. *The British Medical Journal.* **330(7504):** 1357. DOI: 10.1136/bmj.38470.670903.E0.
- Richard, S.A., Black, R.E., Gilman, R.H., Guerrant, R.L., Kang, G., Lanata, C.F., Molbak, K., Rasmussen, Z.A., Sack, R.S., Valentiner-Branth, P. and Checkley,

W. 2012. Wasting is associated with stunting in early childhood. *The Journal of Nutrition*. **142**: 1291 – 1296.

- Rocco, L., Fumagalli, E., Suhrcke, M., 2014. From social capital to health and back. *Health Economics.* **5**: 586 – 605.
- Rooze, S., Mathieu, F., Claus, W., Yangzom, T., Yangzom, D., Goyens, P. and de Maertelaer, V. 2016. Effect of calcium and vitamin D on growth, rickets and Kashin–Beck disease in 0- to 5-year-old children in a rural area of central Tibet. *Tropical Medicine and International Health.* 21: 768 – 775.
- Saaka, M. 2014. Relationship between mother's nutritional knowledge in childcare practices and the growth of children living in impoverished rural communities. *Journal of Health, Population and Nutrition.* **32:** 237 248.
- Sahoo, K., Sahoo, B., Choudhury, A.K., Sofi, N.Y., Kumar, R. and Bhadoria, A.S. 2015. Childhood obesity: Causes and consequences. *African Journal of Family Medicine and Primary Care.* **4**: 187 – 192.
- Said-Mohamed, R., Micklesfield, L.K., Pettifor, J.M. and Norris, S.A. 2015. Has the prevalence of stunting in South African children changed in 40 years? A systematic review. *BMC Public Health*. **15:** 534 DOI: 10. 1186/s12889-015-1844-9.
- Sáinz, N., Barrenetxe, J., Moreno-Aliagaa, M.J. and Martínez, J.A. 2015. Leptin resistance and diet-induced obesity: Central and peripheral actions of leptin. *Metabolism Clinical and Experimental.* 64: 35 – 46.
- Saker, M., Merzouk, H., Merzouk, S.A., Ahmed, S.B. and Narce, M. 2011. Predictive Factors of Obesity and their Relationships to Dietary Intake in Schoolchildren in Western Algeria. *Maedica (Buchar) a Journal of Clinical Medicine*. 6(2): 90 99.
- Senbanjo, I.O., Oshikoya, K.A., Odusanya, O.O. and Njokanma, O.F. 2011. Prevalence of and risk factors for stunting among school children and adolescents in Abeokuta, Southwest Nigeria. *Journal of Health, Population and Nutrition.* 29: 364 – 370.
- Senda, M. 2015. Safety in public spaces for children's play and learning. *International* Association of Traffic and Safety Science. **38**: DOI.org/10.1016/j.iatssr.2015.02.001

- Shapiro, A., Tumer, N., Gao, Y., Cheng, K.Y. and Scarpace, P.J. 2011. Prevention and reversal of diet-induced leptin resistance with a sugar-free diet despite high fat content. *British Journal of Nutrition.* **106:** 390 – 397.
- Sheikh, A.B., Narullah, A., Haq, S., Akhtar, A., Ghazanfar, H., Nasir, A., Afzal, R.M., Bukhari, M.M., Chaudhary, A.Y. and Naqvi, S.W. 2017. The Interplay of genetics and environmental factors in the development of obesity. *Cureus*. 9: DOI: 10.7759/cureus.1435.
- Shekhar, C. 2013. Hidden hunger: Addressing micronutrient deficiencies using improved crop varieties. *Chemistry and Biology*. **11**: 1305 1306.
- Shim, J., Oh, K. and Kim, H.C. 2014. Dietary assessment methods in epidemiologic studies. *Epidemiology and Health.* **36:** DOI: 10.4178/epih/e2014009.
- Shrivastava, S.R. and Shrivastava, P.S. 2013. A longitudinal study of material and socioeconomic factors influencing neonatal birth weight in pregnant women attending an urban health center. *Saudi Journal for Health Science*. **2**: 87 – 92
- Solorzano, C.M.B. and McCartney, C.R. 2010. Obesity and the pubertal transition in girls and boys. *Reproduction*. **140**: 399 410.
- Statistics South Africa. 2016. Consumer Price Index. http://www.statssa.gov.za/publications/P0141/P0141March2016.pdf
- Statistics South Africa. 2018. Mid-year population estimates. https://www.statssa.gov.za/publications/P0302/P03022018.pdf.
- Stuebe, A. 2009. The risk of not breastfeeding for mothers and infants. *Reviews in Obstetrics & Gynaecology*. **2:** 222 231.
- Sunguya, B.F., Ong, K.I.C., Dhakal, S., Mlunde, L.B., Shibanuma, A., Yasuoka, J. and Jimba, M. 2014. Strong nutrition governance is a key to addressing nutrition transition in low and middle-income countries: Review of countries' nutrition policies. *Nutrition Journal.* **13**: DOI: 10.1186/1475-2891-13-65.
- Sunguya, B.F., Ong, K.I.C., Dhakal, S., Mlunde, L.B., Shibanuma, A., Yasuoka, J.,
 Syahrul, S., Kimura, R., Tsuda, A., Susanto, T., Saito, R. and Ahmad, F. 2016.
 Prevalence of underweight and overweight among school-aged children and its association with children's' sociodemographic and lifestyle in Indonesia.
 International Journal of Nursing Sciences. 3: DOI: 10.1016/j.ijnss.2016.04.004.
- Sutharsan, R., O'Callaghan, M.J., Williams, G., Najman, J.M and Mamun, A.A. 2015. Rapid growth in early childhood associated with young adult overweight and
obesity – evidence from a community based cohort study. *Journal of Health, Population and Nutrition.* **33:** DOI 10.1186/s41043-015-0012-2.

- Syahrul, S., Kimura, R., Tsuda, A., Susanto, T., Saito, R. and Ahmad, F. 2016. Prevalence of underweight and overweight among school-aged children and it's association with children's sociodemographic and lifestyle in Indonesia. *International Journal of Nursing Sciences.* **3:** 169 – 177.
- Symington, E.A., Gericke, G.J., Nel, J.H. and Labadarios, D. 2016. The relationship between stunting and overweight among children from South Africa: Secondary analysis of the National Food Consumption Survey - Fortification Baseline I. *South African Medical Journal*. **106:** 65 – 68.
- Tathiah, N., Moodley, I., Mubaiwa, V., Denny, L. and Taylor, M. 2013. Overweight, obesity, underweight and stunting in female primary school learners in rural KwaZulu-Natal, South Africa. South African Medical Journal. 103(10): 718 723. DOI:10.7196/SAMJ.6922.
- Telford, R.M., Telford, R.D., Olive, L.S., Cochrane, T. and Davey, R. 2016. Why are girls less physically active than boys? Findings from the LOOK longitudinal study. *PLoS One*. **11(3)**: DOI: [10.1371/journal.pone.0150041]
- Triegaardt, J.D. 2005. The child support grant in South Africa: A social policy for poverty alleviation? *International Journal of Social Welfare*. **14:** 249 255.
- Tydeman-Edwards, R., Van Rooyen, F.C. and Walsha, C.M. 2018. Obesity, undernutrition and the double burden of malnutrition in the urban and rural southern Free State, South Africa. *Heliyon.* 4(12): 1 – 6 DOI: 10.1016/j.heliyon.2018.e00983
- UNICEF. 2009. State of the World's Children. www.unicef.org/sowc09/report.php
- UNICEF. 2012. The State of the World's Children. www.unicef.org/sowc2012/fullreport.php.
- Van Stuijvenberg, M.E. 2005. Using the school feeding system as a vehicle for micronutrient fortification: Experience from South Africa. *Food Nutrition Bulletin*.
 26: 213 219.
- Van Stuijvenberg, M.E., Nel, J., Schoeman, S.E., Lombard, C.J., du Plessis, L.M. and Dhansay, M.A. 2015. Low intake of calcium and vitamin D, but not zinc, iron or vitamin A, is associated with stunting in 2- to 5- year- old children. *Nutrition*. **31**: 841 – 846.

- Van Wyk, R.B. and Dlamini. C.S. 2018. The impact of food prices on the welfare of households in South Africa. South African Journal of Economic and Management Sciences. 21: DOI.org/10.4102/sajems.v21i1.1979.
- Van Zyl, S., Van der Merwe, L.I., Walsh, C.M., Groenewald, A.J. and van Rooyen,
 F.C. 2012. Risk factor profiles for chronic diseases of lifestyle and metabolic syndrome in an urban and rural setting in South Africa. *African Journal of Primary Health Care and Family Medicine*. 4: DOI.org/10.4102/phcfm.v4i1.346.
- Voortman, T., Kim, K.V.E., Vincent W.V., Jaddoe, V.W.V. and Franco, O.H. 2017. Macronutrient composition of early childhood diet is associated with growth and adiposity. *The Official Journal of the Federation of American Societies for Experimental Biology.* **31**: 1530-6860 DOI: 10.1096/fj.1530-6860.
- Wang, Y. and Lim, H. 2012. The global childhood obesity epidemic and the association between socio-economic status and childhood obesity. *International Review of Psychiatry*. 24(3): 176–188. DOI: 10.3109/09540261.2012.688195.
- Whitaker, R.C., Wright, J.A., Pepe, M.S., Seidel, K.D. and Dietz, W.H. 1997. Predicting obesity in young adulthood from childhood and parental obesity. *The New England Journal of Medicine*. **337(13):** 869 873 DOI: 10.1056/NEJM199709253371301.
- Whitney, N. and Rolfes, S.R. 2011. Understanding nutrition. 12th edition. Thomson Wadsworth, Belmont. 377 385.
- Wiklund, P, 2016. The role of physical activity and exercise in obesity and weight management: Time for critical appraisal. *Journal of Sport and Health Science*.1 – 4.
- Willey, B.A., Cameron, N., Norris, S.A., Pettifor, J.M. and Griffiths, P.L. 2009. Socioeconomic predictors of stunting in preschool children – a population-based study from Johannesburg and Soweto. *South African Medical Journal*. **99:** 450 – 456.
- Williams, C.B., Mackenzie, K.C. and Gahagan, S. 2014. The effect of maternal obesity on the offspring. *Clinical Obstetrics and Gynaecology*. **57:** 508 515.
- Wolde, M., Berhan, Y. and Chala, A. 2015. Determinants of underweight, stunting and wasting among school children. *BMC Public Health.* 15: 8 DOI: 10.1186/s12889-014-1337-2.

- Wolmarans, P., Kunneke, E. and Laubscher, R. 2009. Use of South African food composition database system (SAFODS) and its products in assessing dietary intake data: Part II. South African Journal of Clinical Nutrition. 22: 59 – 67.
- World Bank. 2015. "Population Estimates and Projections." World Bank, Washington, DC. http://data.worldbank.org/data-catalog/population-projection-tables.
- World Health Organisation (WHO), 2003. Global Strategy for Infant and Young Child Feeding. *World Declaration and Plan of Action for Nutrition*.
- World Health Organisation (WHO), 2010. Country profile Indicators. Nutrition Landscape Information System (NLIS). Nutrition Landscape Information System (NLIS) country profile indicators: interpretation guide. 1 2.
- World Health Organisation (WHO), 2015. Obesity and Overweight. WHO Media Centre. **90**: 609-619. 1 38.

World Health Organisation (WHO). 2007. Growth reference data for 5-19 years. 1 - 7. World Health Organisation (WHO). 2008. Interpreting Growth Indicators. 1 - 13.

- Yadav, A., Kataria, M.A., Saini, V. and Yadav, A. 2012. Role of leptin and adiponectin in insulin resistance. *Clinica Chimica Acta*. **2013**: 80 – 84.
- Yamauchi, T., Umekazi, M. and Ohtsuka, R. 2001. Influence of urbanisation on physical and dietary changes in speaking populations in a comparative study of village dwellers and migrants in urban settlements. *British Journal of Nutrition.* **85**: 65 – 73
- Yan, J., Liu, L., Zhu, Y., Huang, G. and Wang, P.P. 2014. The association between breastfeeding and childhood obesity: A meta-analysis. *BMC Public Health*. 14: 1267 DOI: 10.1186/1471-2458-14-1267.
- Yang, C., Chen, J., Liu, Z., Yun, C., Li, Y. Piao, J. and Yang, X. 2015. Association of vitamin A status with over nutrition in children and adolescents. *International Journal of Environmental Research and Public Health.* **12:** 15531–15539.
- You, W. and Henneberg, M. 2016. Meat consumption providing a surplus energy in modern diet contributes to obesity prevalence: An ecological analysis. *BMC Nutrition*. **2**: 22 DOI 10.1186/s40795-016-0063-9.
- Zametkin, A.J., Zoon, C.K., Klein, H.W. and Munson, S. 2004. Psychiatric aspects of child and adolescent obesity: *A Review of the Psychiatry*. **43**: 134 150.
- Zeller, M.H., Reiter-Purtill, J., and Ramey, C. 2008. Negative peer perceptions of obese children in the classroom environment. *Obesity*.**16**: 755 762.

Zhang, T., Cai, L., Ma, L., Jing, J., Chen, Y. and Ma, J. 2016. The prevalence of obesity and influence of early life and behavioural factors on obesity in Chinese Children in Guangzhou. *BMC Public Health.* **16:** 954 DOI 10.1186/s12889-016-3599-3.

ADDENDUM A: TREC CERTIFICATE



University of Limpopo Department of Research Administration and Development Private Bag X1106, Sovenga, 0727, South Africa Tel: (015) 268 2212, Fax: (015) 268 2306, Email:noko.monene@ul.ac.za

TURFLOOP RESEARCH ETHICS COMMITTEE CLEARANCE CERTIFICATE

MEETING:

04 July 2017

PROJECT NUMBER:

R: TREC/103/2017: PG

PROJECT:

Title:

Researcher: Supervisor: Co-Supervisor: School: Degree: Differences in body fatness and nutritional status of urban and rural Primary school children between the ages of six and nine years in the Polokwane Local Municipality, Limpopo Province, South Africa Ms MN Mokabane Dr M van Staden Mr H Malan Molecular and Life Sciences Masters in Physiology

D PROF TAB MASHEGO

CHAIRPERSON: TURFLOOP RESEARCH ETHICS COMMITTEE

The Turfloop Research Ethics Committee (TREC) is registered with the National Health Research Ethics Council, Registration Number: **REC-0310111-031**

Note: i)

- Should any departure be contemplated from the research procedure as approved, the
 - researcher(s) must re-submit the protocol to the committee.
- ii) The budget for the research will be considered separately from the protocol. PLEASE QUOTE THE PROTOCOL NUMBER IN ALL ENQUIRIES.

Finding solutions for Africa

ADDENDUM B: LETTER FROM THE CIRCUIT MANAGER

	DEPARTMENT OF
	CAPRICORN DISTRICT
	DIMAMO CIRCUIT
Eng 1	(rona M S
CONT	AFT: 0000170000
CONT	1. 1. 02201/0805
	he Principal 19 Jehn any 2016
Kakan	a Dilmala Drimanz Coheal
NOROH	
Diman	
PERMI	ISSION TO CONDUCT RESEARCH IN DIMAMO CIRCUIT: UNIVERSITY OF LIMPOPO, ITMENT OF PHYSIOLOGY AND ENVIRONMENTAL HEALTH AT YOUR SCHOOL
1.	When the service is a state of the service of the s
7	The above matter refers.
2.	The above matter refers. Kindly provide the necessary support and cooperation to the team as they conduct research at your institution.
2. 3.	The above matter refers. Kindly provide the necessary support and cooperation to the team as they conduct research at your institution. Sufficient consultation to the Department has been done for this study to be conducted.
2. 3. 4.	The above matter refers. Kindly provide the necessary support and cooperation to the team as they conduct research at your institution. Sufficient consultation to the Department has been done for this study to be conducted. Herewith find documents that pennit them to operate in the circuit.
2. 3. 4. 5.	The above matter refers. Kindly provide the necessary support and cooperation to the team as they conduct research at your institution. Sufficient consultation to the Department has been done for this study to be conducted. Herewith find documents that pennit them to operate in the circuit. The Department wishes you well and appreciates your interest on matters that would uplift
2. 3. 4. 5.	The above matter refers. Kindly provide the necessary support and cooperation to the team as they conduct research at your institution. Sufficient consultation to the Department has been done for this study to be conducted. Herewith find documents that pennit them to operate in the circuit. The Department wishes you well and appreciates your interest on matters that would uplift the health standard in Education.
2. 3. 4. 5.	The above matter refers. Kindly provide the necessary support and cooperation to the team as they conduct research at your institution. Sufficient consultation to the Department has been done for this study to be conducted. Herewith find documents that pennit them to operate in the circuit. The Department wishes you well and appreciates your interest on matters that would uplift the health standard in Education. CIRCUIT MANAGER
2. 4. 5.	The above matter refers. Kindly provide the necessary support and cooperation to the team as they conduct research at your institution. Sufficient consultation to the Department has been done for this study to be conducted. Herewith find documents that pennit them to operate in the circuit. The Department wishes you well and appreciates your interest on matters that would uplift the health standard in Education. CIRCUIT MANAGER

ADDENDUM C: A LETTER FROM DEPARTMENT OF EDUCATION



Private Bag X 9711 PCLOKWANE 0703 Tel: 015 285 7300 Fax: 015 285 7499

EDUCATION CAPRICORN POLOKWANE DISTRICT

CONFIDENTIAL

Eng :	Mphaphuli AJ
Tel No.:	015 285 7410
Email :	MphaphuliAJ@edu.limpopo.gov.za
Date:	06 November 2015

To : Dr M van Staden

University of Limpopo

Department of Physiology and Environmental Health

SUBJECT: APPLICATION TO CONDUCT RESEARCH. PIETERSBURG CIRCUIT , MANKWENG CIRCUIT , MAMABOLO CIRCUIT, KGAKOTLOU CIRCUIT AND DIMAMO CIRCUIT .

Title: A non –invasive investigation into the prevalence and risk factor for malnutrition and its associated complications in children and adolescents in Limpopo Province, South Africa.

- 1. The about matter refers.
- The Department wishes to inform you that your request to conduct a research has been approved.
- 3. The following conditions should be considered
 - 3.1 The research should not have any financial implication for Limpopo Department of Education.
 - 3.2 Arrangements should be made with both the circuit offices and school concerned.
 - 3.3 The conduct of research should not anyhow disrupt the academic programs at schools.



- 3.4 The research should not be conducted during the time examinations especially the fourth term.
- 3.5 During the study , the research ethics should be practiced, in particular the principle voluntary participation (the people involved should be respected)
- 3.6 Upon completion of research study, the researcher shall share the final product of the research with Department.
- Furthermore you are expected to produce this letter at schools/offices where you intend conduct your research as evidence that you are permitted to conduct the research.
- The department appreciates the contribution that you wish to make and wish you success in your research.

Name of Circuit	Name of Primary Schools	Name of Secondary Schools
Pietersburg Circuit	 Pietersburg Comprehensive School Greenside Primary School Laerskool Pietersburg Ivy Park Primary School 	 Westenburg Secondary School Capricorn High School Noorderland High School Taxilla High School
Mankweng Circuit	(m)	• Mphetsebe Secondary School
Mamabolo Circuit	Mamabolo Primary School	Motlakaro Secondary School
Dimamo Circuit	Kokona Dikgale Primary School	Moruleng Secondary School
Kgakotlou Circuit	Ntji Mothapho Primary SchoolItireleng Primary School	Mothimako Secondary School

Best wishes MR MOTHEMANE KD ACTING DISTRICT SENIOR MANAGER 2015 -11-09 Cnr Blaauwberg & Yster Street, Ladarna PRIVATE BAG X9711 "We Belong, We Care, We Serve"

ADDENDUM D: A FORMAL LETTER FOR PERMISSION TO CONDUCT THE STUDY

PROJECT TITLE: Differences in body fatness and nutritional status of urban and rural primary school children between the ages of six and nine years in the Polokwane Local Municipality, Limpopo province, South Africa.

PROJECT LEADERS:Dr M van Staden (Promoter)Mr H Malan (Co-promoter)

INFORMATION FOR PARTICIPANTS

- Your school is invited to participate in the following research project/experiment: Differences in body fatness and nutritional status of urban and rural primary school children between the ages of six and nine years in the Polokwane Local Municipality, Limpopo province, South Africa.
- 2. Participation in the project is completely voluntary and you are free to withdraw from the project/experiment (without providing any reasons) at any time. You are, however, requested not to withdraw without careful consideration since such action might negatively affect the project/experiment.
- 3. It is possible that you might not personally experience any advantages during the experiment/project, although the knowledge that may be accumulated through the project/experiment might prove advantageous to others.
- 4. You are encouraged to ask any questions that you might have in connection with this project/experiment at any stage. The project leader and her/his staff will gladly answer your question. They will also discuss the project/experiment in detail with you.
- 5. Your involvement in the project:

The study activities will include that you give the researcher permission to address learners in your school to inform them about the project and to invite them to participate, collect data from 50 children who return informed consent forms signed by the parents. The researcher will need to visit the school about three times to gather all information.

If you have any questions or queries regarding this research project, either now or in the future, please do not hesitate to contact either:

Dr. Marlise van Staden, Department of Physiology and Environmental Health, University of Limpopo, Q-Block, 1st floor, Phone Number: 015 268 2817, E-mail address: marlise.vanstaden@ul.ac.za,

OR

Mr H Malan, Department of Physiology and Environmental Health, University of Limpopo, Q-Block, 1st floor, Phone Number: 015 268 3005, E-mail address: henk.malan@ul.ac.za,

OR

Ms MN Mokabane, Department of Physiology and Environmental Health, University of Limpopo, Q-Block, 1st floor, Phone Number: 015 268 2209, E-mail address: donnah.mokabane@gmail.com.

PROCEDURES TO BE FOLLOWED

The researcher will explain the study to children and invite them to participate. All children will receive an information letter and consent form to take home. Only children who return signed consent forms may participate in the study. The researcher will visit participating children at the school during school hours (08:00 - 14:00). Each participating child will receive a questionnaire for parents/guardians to complete. The questionnaire contains questions on demography, medical history, and snacking behaviour. The researcher will weigh the child and measure his or her height, waist circumference, hip circumference and some skin folds. Twenty percent of the children will be interviewed to determine dietary intake. The researcher will return to the school on a second occasion to do another dietary intake interview with these children.

NATURE OF DISCOMFORT

To weigh the child all shoes and heavy clothing are removed and the child is asked to step onto a scale and to stand still and look straight ahead for the few seconds it takes to record the weight. Height is then also determined by asking the child (who is still barefoot) to stand with his/her back against a stadiometer. The child's back is against the ruler part and the slider is pressed lightly onto the head of the child so that the correct height can be recorded. Both of these measurements are done fast and it should not be uncomfortable for the child. Skin folds are measured with a device called a "skin fold calliper" and it functions like a plier to determine the thickness of a skinfold. The researcher carefully pinches the skin of the child and pulls it away from the body to allow the calliper to fit over the skin fold. The calliper is then released and it measures the skin fold. This procedure is only slightly uncomfortable, but the measurement only takes a few seconds and is not painful.

ADVANTAGES THAT MAY BE EXPECTED FROM THE RESULTS OF THE EXPERIMENT

According to our knowledge data on body fatness, among urban and rural children in the Limpopo Province, South Africa is limited. Therefore, the importance of this research study is that its findings will shed more light on the current body fatness of children and some of the factors affecting their body fatness. It will also increase community awareness about the risks involved in being obese, overweight or underweight. Furthermore, detailed knowledge of these variables of children will assist the children's parents, the school and the policy holders in further understanding, monitoring and minimizing the likelihood of serious health problems that continue in adulthood due to increased or decreased body fatness during childhood. The results of this study will also inform the need for intervention studies to prevent underweight and overweight or obesity in children.

Each participating child will receive a feedback form with his/her measurements as well as the normal values to take home. This will be accompanied by an information leaflet on healthy lifestyle. Each participating school will receive a short report on the body fatness of the participating children. No child's name will be mentioned in this report so you can rest assure that all personal information will remain confidential at all times. The school will also receive some information on healthy lifestyle with suggestions to improve physical activity and healthy eating on the school grounds.

Yours sincerely, Dr M van Staden

ADDENDUM E: A FORMAL LETTER TO REQUEST PERMISSION TO COLLECT DATA

PROJECT TITLE: Differences in body fatness and nutritional status of urban and rural primary school children between the ages of six and nine years in the Polokwane Local Municipality, Limpopo province, South Africa.

PROJECT LEADERS: Dr M van Staden Mr H Malan

INFORMATION FOR PARTICIPANTS

- Your are invited to participate in the following research project/experiment: Differences in body fatness and nutritional status of urban and rural primary school children between the ages of six and nine years in the Polokwane Local Municipality, Limpopo province, South Africa.
- 2. Participation in the project is completely voluntary and you are free to withdraw from the project/experiment (without providing any reasons) at any time. You are, however, requested not to withdraw without careful consideration since such action might negatively affect the project/experiment.
- 3. It is possible that you might not personally experience any advantages during the experiment/project, although the knowledge that may be accumulated through the project/experiment might prove advantageous to others.
- 4. You are encouraged to ask any questions that you might have in connection with this project/experiment at any stage. The project leader and her/his staff will gladly answer your question. They will also discuss the project/experiment in detail with you.
- 5. Your involvement in the project:

The study activities will include that you fill in a questionnaire to give some demographic and medical information about you and your child. Although we will prefer it if you complete the entire questionnaire, you may leave questions blank if you do not feel comfortable answering them. Furthermore, the researcher will assess your child's body fatness by weighing and measuring him/her. Ten percent of the participating children.will be interviewed to determine dietary intake. The researcher

may telephone you to clear up some uncertainties with regard to the dietary information provided by your child.

If you have any questions or queries regarding this research project, either now or in the future, please do not hesitate to contact either:

Ms MN Mokabane, Department of Physiology and Environmental Health, University of Limpopo, Q-Block, 1st floor, Phone Number: 015 268 2209, E-mail address: donnah.mokabane@gmail.com,

OR

Dr. Marlise van Staden, Department of Physiology and Environmental Health, University of Limpopo, Q-Block, 1st floor, Phone Number: 015 268 2817, E-mail address: marlise.vanstaden@ul.ac.za.

PROCEDURES TO BE FOLLOWED

The researcher will explain the study to children and invite them to participate. All children will receive an information letter and consent form to take home. Only children who return signed consent forms may participate in the study. The researcher will visit participating children at the school during school hours (08:00 - 14:00). Each participating child will receive a questionnaire for parents/guardians to complete. The questionnaire contains questions on demography, medical history, and snacking behaviour. The researcher will weigh the child and measure his or her height, waist circumference, hip circumference and some skin folds. Ten percent of the children will be interviewed to determine dietary intake. The researcher will return to the school on a second occasion to do another dietary intake interview with these children and to collect the questionnaires.

NATURE OF DISCOMFORT

To weigh the child all shoes and heavy clothing are removed and the child is asked to step onto a scale and to stand still and look straight ahead for the few seconds it takes to record the weight. Height is then also determined by asking the child (who is still barefoot) to stand with his/her back against a stadiometer. The child's back is against the ruler part and the slider is pressed lightly onto the head of the child so that the correct height can be recorded. Both of these measurements are done fast and it should not be uncomfortable for the child. Skin folds are measured with a device called a "skin fold calliper" and it functions like a plier to determine the thickness of a skinfold. The researcher carefully pinches the skin of the child and pulls it away from the body to allow the calliper to fit over the skin fold. The calliper is then released and it measures the skin fold. This procedure is only slightly uncomfortable, but the measurement only takes a few seconds and is not painful.

ADVANTAGES THAT MAY BE EXPECTED FROM THE RESULTS OF THE EXPERIMENT

According to our knowledge data on body fatness, among urban and rural children in the Limpopo Province, South Africa is limited. Therefore, the importance of this research study is that its findings will shed more light on the current body fatness of children and the some of the factors affecting their body fatness. It will also increase community awareness about the risks involved in being obese, overweight or underweight. Furthermore, detailed knowledge of these variables of children will assist the children's parents, the school and the policy holders in further understanding, monitoring and minimizing the likelihood of serious health problems that continue in adulthood due to increased or decreased body fatness during childhood. The results of this study will also inform the need for intervention studies to prevent underweight and overweight or obesity in children.

Each participating child will receive a feedback form with his/her measurements as well as the normal values to take home. This will be accompanied by an information leaflet on healthy lifestyle. Each participating school will receive a short report on the body fatness of the participating children. No child's name will be mentioned in this report so you can rest assure that all personal information will remain confidential at all times. The school will also receive some information on healthy lifestyle with suggestions to improve physical activity and healthy eating on the school grounds.

Yours sincerely, Dr M van Staden

ADDENDUM F: CONSENT FORMS (ENGLISH, SEPEDI AND AFRIKAANS)

UNIVERSITY OF LIMPOPO ETHICS COMMITTEE

PROJECT TITLE: Differences in body fatness and nutritional status of urban and rural primary school children between the ages of six and nine years in the Polokwane Local Municipality, Limpopo province, South Africa.

PROJECT LEADERS: Dr M van Staden Mr H Malan

CONSENT FORM

I,

hereby

voluntarily consent to participate in the following project: Differences in body fatness and nutritional status of urban and rural primary school children between the ages of six and nine years in the Polokwane Local Municipality, Limpopo province, South Africa.

I realise that:

- 1. The study deals with the assessment of children's body fatness and nutritional status of urban and rural primary school children between the ages of six and nine years in the Polokwane Local Municipality, Limpopo province, South Africa.
- 2. The procedure or treatment envisaged may hold some risk for me that cannot be foreseen at this stage;
- The Ethics Committee of the University of Limpopo has approved that individuals may be approached to participate in the study;
- 4. The experimental protocol, i.e., the extent, aims and methods of the research, has been explained to me;
- 5. The protocol sets out the risks that can be reasonably expected as well as possible discomfort for persons participating in the research, an explanation of the anticipated advantages for myself or others that are reasonably expected from the research and alternative procedures that may be to my advantage;
- 6. I will be informed of any new information that may become available during the research that may influence my willingness to continue my participation;

- Access to the records that pertain to my participation in the study will be restricted to persons directly involved in the research;
- 8. Any questions that I may have regarding the research, or related matters, will be answered by the researchers;
- 9. If I have any questions about, or problems regarding the study, or experience any undesirable effects, I may contact a member of the research team;
- 10. Participation in this research is voluntary and I can withdraw my participation at any stage;
- 11. If any medical problem is identified at any stage during the research, or when I am vetted for participation, such condition will be discussed with me in confidence by a qualified person and/or I will be referred to my doctor;
- 12.I indemnify the University of Limpopo and all persons involved with the above project from any liability that may arise from my participation in the above project or that may be related to it, for whatever reasons, including negligence on the part of the mentioned persons.

Signature of researched person		Signatu	ure of witness
Signature of person that informed		Signature of parent	/guardian
Signed at	this	day of	2016

UNIVERSITY OF LIMPOPO KOMITI YA MELAO

PROJECT TITLE: A non-invasive investigation into the prevalence of and risk factors for malnutrition and its associated complications in children and adolescents in Limpopo Province, South Africa

PROJECT LEADERS:Dr M van Staden (PhD Physiology)Prof MJ Potgieter (PhD Botany)

CONSENT FORM

Nna,______ ke dumela go ithaopa go tšea karolo mo projekeng ye e latelago: A non-invasive investigation into the prevalence of and risk factors for malnutrition and its associated complications in children and adolescents in Limpopo Province, South Africa).

Ke lemoga gore:

- Thuto ye ya dinyakišišo e šoma ka go lekola maemo a mahlahlo a ngwana sekolong le ka lapeng, le go lekola nako yeo ngwana ae fetšago a dutše a sa dire selo;
- Tshepedišo goba mokgwa wa tswaro wo o letetšwego o ka ntšwarela dikotsi tše di sa bonelwego pele sebakeng se;
- Komiti ya melao ya Unibesiti ya Limpopo e dumeletše gore batho ba ka latwa ka o tee o tee gore ba tšeye karolo ka mo gare ga thutonyakišišo ye;
- Ke hlalosoditše melaoana ya teko, bjalo ka go naba maikemešetšo, le mekgwa ya nyakišišo;
- Melaona ye e bea pepeneng dikotsi tšeo di ka lebelelwago gape le dikgonagalo tša maemo a go se thabiše go batho bao ba tšeago karolo ka gare ga nyakišišo,, tlaloso ya mehola yeo e ka letelwago go nna goba ba bangwe;
- Ke tla tsebišwa ka tsebo yengwe le yengwe ye ntshwa yeo e ka bago gona bakong ya nyakišišo yeo e ka hlohleletšago boethaopo bjaka gore ke tswele pele ke tšea karolo;
- Tumelelo ya direkoto tšeo di amanago le go tšea karolo ga ka mo thutonyakašišong ye e tla felela fela go bao ba lego karolong ya nyakišišo;
- Dipotšišo di fe goba dife tšeo nka bago le tšona mabapi le nyakišišo ye, goba tše dingwe tšeo di amanago le yona, di tla arabjwa ke banyakišiši;

- Ge ke nale dipotšišo tše dingwe le tše dingwe ka ga thuto, goba mathata mabapi le thutonyakišišo, goba go itemogela ditlamorago tsa go se kgahliše nka ikgokagantša le leloko le lengwe la sehlopa sa banyakišiši;
- 10.Go tšea karolo ka gare ga nyakišišo ye ke ka go ithaopa gomme nka gogela morago go tšea karolo ga ka nako ye ngwe le yengwe;
- 11. Ge bothata bja kalafi bo ka lemogwa nakong efe goba efe ya nyakišišo goba ge ke nyakollwa gore ke kgone go tšea karolo, boemo bo bjalo botlo sekwasekwa le nna ka sephiring le motho wa maleba gob aka romelwa ngakeng ya ka;
- 12. Ke tshepiša go lefela Unibesithi ya Limpopo le batho ka moka bao ba amegago mo projekeng ye ya ka godimo dikotlo ka moka tšeo di ka tšwelelago go tšwa go tšeyeng karolo ga ka mo projekeng ya ka godimo goba tšeo di ka amanago le yona ntle le go šetša mabaka, go akaretša le bošaedi legatong la batho bao ba boletšwego;

ka

Mosaeno wa monyakišiši

Mosaeno wa hlatse

Mosaeno wa motsebiši

Mosaeno wa motswadi/ Mohlokomedi

E saenwe

letšatši la _____ 2016

UNIVERSITY OF LIMPOPO ETHICS COMMITTEE

 PROJEK TITEL: 'n Nie-ingrypende ondersoek na die voorkoms van en risikofaktore vir wanvoeding en die verwante komplikasies in kinders en adolesente in Limpopo Profinsie, Suid-Afrika
 PROJEK LEIERS: Dr M van Staden (PhD Physiology) Prof MJ Potgieter (PhD Botany)

TOESTEMMINGSVORM

Ek,

gee hiermee

vrywillig toestemming om aan die volgende projek deel te neem: 'n Nie-ingrypende ondersoek na die voorkoms van en risikofaktore vir wanvoeding en die verwante komplikasies in kinders en adolesente in Limpopo Profinsie, Suid-Afrika.

Ek begryp dat:

- Die studie verband hou met die meting van kinders se fisiese aktiwiteitsvlakke by die skool en by die huis en met die tyd wat kinders spandeer met sittende aktiwiteite;
- 2. Die prosedure of behandeling sekere risiko's vir my mag inhou wat nie op hierdie stadium voorsien kan word nie;
- 3. Die Etiese Kommitee van die Universiteit van Limpop het goedgekeur dat individue genader mag word om aan die studie deel te neem;
- 4. Die eksperimentele protokol, m.a.w. die omvang, oogmerke, en metodes van die navorsing is aan my verduidelik;
- 5. Die protokol bevat 'n uiteensetting van die risiko's wat binne redelike perke verwag kan word, sowel as moontlike ongemak vir persone wat aan die navorsing deelneem, 'n verduideliking van die verwagte voordele wat vir myself of ander uit die navorsing kan spruit en alternatiewe prosedures wat tot my voordeel kan strek;
- Ek sal inkennis gestel word van enige nuwe inligting wat tydens die navorsing beskikbaar mag word, wat my voortgesette bereidwilligheid om aan die projek deel te neem kan beïnvloed;
- Toegang tot die rekords wat verband hou met my deelname aan die studie sal tot individue wat direk by die studie betrokke is beperk word;

- 8. Enige vrae wat ek in verband met die navorsing, of verwante sake het, sal deur die navorsers beantwoord word;
- 9. Indien ek enige vrae oor, of probleme met die studie ondervind, of enige ongewenste gevolge ervaar, kan ek 'n lid van die navorsingspan kontak;
- 10. Deelname aan die navorsing is vrywillig en ek kan te enige tyd my deelname staak;
- 11. Indien enige mediese probleem op enige tydstip tydens die navorsing, of as ek vir die projek gewerf word, geïdentifiseer word, sal die toestand konfidensiëel met my bespreek word deur 'n gekwalifiseerde persoon en/of ek sal na my dokter verwys word;
- 12. Ek vrywaar die Universiteit van Limpopo en alle individue betrokke by bogenoemde projek van enige aanspreeklikheid wat uit my deelname aan bogenoemde projek mag spruit of wat daarmee verband mag hou, vir watter rede ook al, insluitende nalatigheid aan die kant van die genoemde individue.

Handtekening van deelnemer		Handtekening	g van getuie
Handtekening van persoon wat ingelig h	et	Handtekening	g van ouer/voog
Geteken te	hierdie	dag van	2016

ADDENDUM G: QUESTIONNAIRES (ENGLISH, SEPEDI AND AFRIKAANS)

Subject no:	
School name:	
Interviewer:	
Date:	

Questionnaire

Please complete the following questionnaire as accurately as possible. If you feel uncomfortable about answering some of the questions you may leave them blank.

SECTION A: DEMOGRAPHIC INFORMATION

- Child's name and surname: ______
 Child's date of birth:
- 3. Child's age _____ years and _____ months
- 4. Child's gender: male/female
- 5. You are the child's: Mother/Father/Guardian/Other (specify)
- 6. Your age: _____ years and _____ months
- 7. In which area/suburb do you stay _____
- 8. Do you own a motor vehicle? Yes/No
- 9. Including yourself, how many people aged 18 years and older live in your household?

Enter number_____

10. How many children (including babies) under the age of 18 years live in your household?

Enter number_____

11. What is your monthly household income, net income after taxes and fringe benefit deductions? (Optional) (Choose only one option)

Less than R1000 R1000 to R2000 R2000 to R3000 R3000 to R4000 R4000 to R5000 R5000 to R6000 R6000 to R7000 R7000 to R8000R8000 to R9000 R9000 to R10000 R10000 and more

SECTION B: MEDICAL HISTORY

- What was your (or the child's mother's age) when giving birth to him/her?
 ___years.
- 2. How many weeks were you (or the child's mother) pregnant when you gave birth to this child? _____ weeks
- 3. What was the birth weight of this child (as indicated on the "road to health" chart)? _____kg
- 4. Does the child's father, mother, grandparents, brother or sister suffer from any of the following conditions?
- a. Type 2 diabetes mellitus
- b. Hypertension
- c. Stroke
- d. High cholesterol

5. How many people in the household smoke?

SECTION C: DIETARY INFORMATION

Please answer the following questions as accurately as possible.

- 1. How many times per week does the child eat sweets, biscuits or cake? _____
- 2. How many times per week does the child eat take-away foods? _____
- 3. How many times per week does the child drink cold drinks or fruit juice?
- 4. How many times per week does the child eat fresh fruit? _____

SECTION D: PHYSICAL ACTIVITY

(We want to know how much time your child spends doing different things each day. Please put an X in one box for each question to tell us the number of hours your child spends doing each activity)

- 1. In a typical week, how much time does your child usually spend watching television, DVD's or videos?
- None Less than 1 hour 1–2 hours 3–5 hours 6–10 hours 11–14 hours 15–20 hours More than 20 hours

- In a typical week, how much time does your child spend playing video games, cell phone games, or internet to e-mail or chat online? (Example: Nintendo, XBOX, PlayStation, WhatsApp, Mxit, etc).
- None Less than 1 hour 1–2 hours 3–5 hours 6–10 hours 11–14 hours 15–20 hours More than 20 hours
- In a typical week, how much time does your child spend reading outside of school? This includes reading books, magazines, newspapers, doing homework.
- None Less than 1 hour 1–2 hours 3–5 hours 6–10 hours 11–14 hours 15–20 hours More than 20 hours
- 4. In comparison to your children how would you rate your physical activity level at present?

Less active/The same/More Active

5. When you were a child were you less or more physically active than your child? Less active/The same/More Active

6. How do your children get to school? Walk/Run/Bus/Car/Bicycle.

7. How did you get to school as a child? Did not go/Walk/Run/Bus/Car/Bicycle.

Contact details:

Parent/Guardian's name (Please print your name in full)

Parent/Guardian's gender (please select your answer):Male/Female

Parent/Guardian's age: _____years(Mengwaga).

Landline Phone No_____Cellular Phone No _____

Name of your child (Please print child's name in full)

Name of School your child attends _____

Gender of your child (Please select your answer) Male /Female.

(Thank you for completing this Questionnaire).

Subject no:	
School name:	
Interviewer:	
Date:	

SECTION A: DEMOGRAPHIC INFORMATION

SERIPA SA A : TSEBIŠO

1.	Child's name and surname
Leina	la ngwana le sefane):
2.	Child's date of birth
Letšat	ši la matswalo a ngwana:
3.	Child's age years andmonths
Meng	waga ya ngwana le dikgwedi tše
4.	Child's gender: male/female
Bong	ba ngwana: monna/mosadi
5.	You are the child's Mother/Father/Guardian/Other (specify)
O tswa	ala jwang le ngwana: Mma/papa/mohlokomedi (Ye ngwe hlalosa)
6.	Your age: years and months
Meng	waga ya gagomengwaga ledikgwedi tše
7.	In which area/suburb do you stay
O dula	a tulong efe
8.	Do you own a motor vehicle?(O nale koloi)? Yes/No (Ee/Aowa)

9. Including yourself, how many people aged 18 years and older live in your household?

Enter number _____

O bala/akaretša le wena, ke batho ba ba kae ba go ba ka godimo ga mengwa ye 18 bao ba dulago ka ntlong y a gago?

Lokela nomoro _____

10. How many children (including babies) under the age of 18 years live in your household?

Enter number _____

Ke bana ba ba kae (go akaretša le masea) ba go ba ka fase ga mengwa ye 18 bao ba dulago ka ntlong ya gago?

Lokela nomoro_____

- 11. What is your monthly household income, net income after taxes and fringe benefit deductions? (Optional)(Choose only one option)
- Less than R1000 R1000 to R2000 R2000 to R3000 R3000 to R4000 R4000 to R5000 R5000 to R6000 R6000 to R7000 R7000 to R8000R8000 to R9000 R9000 to R10000 R10000 and more
- Ke setšeka sefe se se tsenago ka lapeng la gago, ka morago ga motšhelo le go fokotšwa ga dikokeletšo) Kgetho) kgetha e tee fela)
- Less than R1000 R1000 to R2000 R2000 to R3000 R3000 to R4000 R4000 to R5000 R5000 to R6000 R6000 to R7000 R7000 to R8000R8000 to R9000 R9000 to R10000 R10000 and more

SECTION B (SERIPA SA B): MEDICAL HISTORY(HISTORI YA TŠA KALAFO)

- 7. What was your (or the child's mother's age) when giving birth to him/her? (Nna o be o nale (goba mengwaga ya mmago ngwana) mengwaga e me kae ge o be o belega ngwana o) _____years (Mengwaga).
- 8. How many weeks were you (or the child's mother) pregnant when you gave birth to this child? (O be o nale dibeke tše kae (goba mmago ngwana) o le mmeleng ge o belega ngwana o)? _____ weeks (dibeke)
- 9. What was the birth weight of this child (as indicated on the "road to health" chart)? (Nna ngwana o, o be a kala bokae (bjalo ka ge go laeditšwe godimo ga tšhate ya "road to health" _____kg
- 10. Does the child's father, mother, grandparents, brother or sister suffer from any of the following condition? (A e kaba papago, mmago, bomakgalo, buti goba sesi wa ngwana o nale bo bobongwe bja malwetši a a latelago)?
- a. Type 2 diabetes mellitus
- b. Hypertension
- c. Stroke
- d. High cholesterol
- 11. How many people in the household smoke? (Ke batho b aba kae bao ba kgogago motšoko ka mo lapeng)?_____

SECTION C (SERIPA C): DIETARY INFORMATION (TSEBIŠO YA MEKGWA YA GO JA)

Please answer the following questions as accurately as possible.

Ka kgopelo araba dipotšišo tše dilatelago ka go nepagala ka mokgwa o o ka kgonago ka gona.

How many times per week does the child eat sweets, biscuits or cake? _____
 Kega kae mo bekeng mo ngwana a jago malekere. dikuku, dikhekhe? _____

2. How many times per week does the child eat take-away foods? _____

Kega kae mo bekeng mo ngwana a jago dijo tša go tšwelela diapeilwe? _____

3. How many times per week does the child drink cold drinks or fruit juice? _____

Kega kae mo bekeng mo ngwana a nwago dino tša go tonya goba metsanakodi a dikenya?____

4. How many times per week does the child eat fresh fruit?

Kega kae mo bekeng mo ngwana a jago dikenywa tša tša go sepone? _____

5. Was this child ever breastfed/fed breast milk?

Aa ngwana o o ile a fepšwa matutu?

- a. Yes Ee
- b. No Aowa
- c. Don't know Ga ke tsebe
- 6. How long after birth was this child first breastfed?

O tšere nako ye kae ka morago ga go belegwa go nyantšhwa?

- a. Immediately Semetseng
- b. Less than one hour Pele ga iri ye tee
- c. Less than 24 hours Pele ga iri tše 24
- d. More than 24 hours Ka morago ga iri tše 24
- e. I don't know Ga ke tsebe
- 7. How was this child breastfed for the first 6 months?

Naa kgwedi tše 6 tša pele ngwana o be a nyantšwa jwang?

- a. Breastfeeding exclusively (Go nyantšhwa matutu feela a sa je goba go nwa selo ka ntle le matutu)
- b. Both breastfeeding and formula milk / cow milk (O be a nyantšwa ebile a fepšwa maswi a go rekwa)
- c. Other, specify Yengwe, hlalosa _____

8. How was expressed milk given?

Naa matutu a be a fiwa ngwana bjang?

- a. Bottle with teat/nipple lepotlolo la go ba le hlogo
- b. Cup Komiki
- c. Spoon Lelepola
- d. Other, specify Yengwe, hlalosa _____
- e. I don't know Ga ke tsebe
- 9. For how long was this child breastfed / fed breast milk? _____ months

(Ngwana o nyantšhitšwe matutu nako ye kaakang?) _____ dikgwedi.

10. How old was this child when milk/food other than breast milk was given for the first time? _____ months

Ngwana o be a nale nako ye kaakang ge a be a fiwa maswi/dijo tše dingwe ntle le matutu la mathomo?_____ dikgwedi

11. What was the first drink other than breast milk that this child received?

Naa ke seno sefe ka ntle le matutu seo ngwana a se nwelego la mathomo?

- a. Infant formula Maswi a bana a go rekwa
- b. Water Metsee
- c. Gripe water
- d. Sugar water Meetse a sokiri
- e. Other, specify Yenngwe, hlalosa _____

SECTION D SERIPA SA D: PHYSICAL ACTIVITY (MEŠOMO YA MMELE)

- (We want to know how much time your child spends doing different things each day. Please put an X in one box for each question to tell us the number of hours your child spends doing each activity)
- Re nyaka go tseba gore ngwana wa gago o tšea nako ye kaakang a dira dilo tše go fapanafapana letšatši ka letšatši. Ka kgopelo bea le swayo le X ka gare ga lepokisi le lengwe le lengwe la potšišo gore botša nomoro ya diiri tšeo ngwana wag ago a di fetšago a dira mešongwana o mongwe le o mongwe
- 1. In a typical week, how much time does your child usually spend watching television, DVD's or videos?
- Ke ga kae mo bekeng moo ngwana wa gago a fetšago nako a lebeletše Telebišene, DVD's or Bedio?

None Less than 1 hour 1–2 hours 3–5 hours 6–10 hours 11–14 hours 15–20 hours More than 20 hours

 In a typical week, how much time does your child spend playing video games, cell phone games, or internet to e-mail or chat online? (Example: Nintendo, XBOX, PlayStation, WhatsApp, Mxit, etc).

Ke ga kae mo bekeng moo ngwana wa gago a fetšago nako raloka dithaloko tša Bedio, tša sellathekeng, goba internet go fihla go imeile

None Less than 1 hour 1–2 hours 3–5 hours 6–10 hours 11–14 hours 15–20 hours More than 20 hours

 In a typical week, how much time does your child spend reading outside of school? This includes reading books, magazines, newspapers, doing homework.

Naa ngwana wag ago o tšea nako ye kae a bala a se ka sekolong? Se se akaretša go bala dipuku, magazines, dikuranta, le go dira mošomo wa gae?

None Less than 1 hour 1–2 hours 3–5 hours 6–10 hours 11–14 hours 15–20 hours More than 20 hours

4. In comparison to your children how would you rate your physical activity level at present?

Less active The same More Active

Ge o ke papetša le bana ba gago, o ka ke kala mahlahla a mmele wa gago bjang A mmanyane A tekano A godimo

5. When you were a child were you less or more physically active than your child?
Less active The same More Active
Ge obe o le ngwana, naa o be o nale mahlahla a ka godimo goba ka fase ga a ngwana wag ago?
A mmanyane A tekano A godimo
6. How do your children get to school

Walk Run Bus Car Bicycle

? Naa ban aba gago ba fihla jwang sekolong?

Ba sepela Ba kitima Ka pese Ka koloi Ka paesekela

7.	How	did you get to	o school as	a child?				
Did n	ot go	Wal	K R	un l	Bus Car		Bicycl	е
Naa v	wena d	o be o fihla bj	ang sekolor	ng?				
Ga se	e ke ts	ene sekolo	Ka go se	pela I	Ka go kitim	a	Ka	paesekela
		Ka koloi	Ka paes	ekela				
Cont	act de	etails:						
Parer	nt/Gua	rdian's name	(Please pri	nt your nam	e in full)			
Leina	la	motswadi/	mohlokom	edi (Ngwa	la leina	la ga	ago ka	a botlalo)
Guar	dian's	gender (plea	se select yc	our answer) I	Male/Fema	le		
Bong	ba mo	otswasi/mohlo	okomedi (kg	jetha karabo	ya gago)	Monna/N	/losadi	
Parer	nt/Gua	rdian's age: _	yea	ars.				
Meng	waga	ya motswadi/	mohlokome	edi:	mengwaga	a.		
Land	ine Pł	none No	Cell	ular Phone N	No (Nomor	o ya mo	gala)	
Nom	oro ya	mogala						
Name	e of yo	our child (Plea	se print chi	ld's name in	full)			
Leina	la ng	wana wa gag	o (Ngwala l	eina la ngwa	ina ka botla	alo)		
Name	e of So	chool your chi	ld attends _					
Leina	la sel	kolo seo ngwa	ana wa gag	o a se tsena	go		_	
Gend	er of y	our child (Ple	ase select	your answer) Male/Fer	nale		
Bong	ba ng	wana wa gag	o (Kgetha I	karabo ya ga	igo) Mon	ina/Fema	ale	
Parer	nt/Gua	ırdian's s	ignature	(Mosaeni	wa	motsw	/adi/mo	hlokomedi)
			Date					
Mosa	eni wa	a motswadi/m	ohlokomed	i		Letšatši		

Thank you for completing this Questionnaire

Re a leboga ge o kgonne go tlatša lenaeopotšišo le.

Subject no:	
School name:	
Interviewer:	
Date:	

SECTION A: DEMOGRAPHIC INFORMATION

AFDELING A: DEMOGRAFIESE INLIGTING

1. Child's name and surname: _____

Kind se naam en van:

2. Child's date of birth:

Kind se geboortedatum:

3. Child's age _____ years and _____ months

Kind se ouderdom _____ jaar en _____ maande

4. Child's gender: male/female

Kind se geslag: manlike/vroulik

5. You are the child's: Mother/Father/Guardian/Other (specify)

U is die kind se: Moeder/Vader/Voog/Ander (spesifiseer)

6. Your age: _____ years and _____ months

U ouderdom: _____ jaar en _____ maande

7. In which area/suburb do you stay?

In watter area/woongebied woon u?

8. Do you own a motor vehicle? Yes/No

Besit u 'n voertuig? Ja/Nee

9. Including yourself, how many people aged 18 years and older live in your household?

Enter number_____

Insluitende u, hoeveel individue van 18 jaar of ouer woon in u huishouding?

10. How many children (including babies) under the age of 18 years live in your household?

Enter number_____

Hoeveel kinders (insluitende babas) onder die ouderdom van 18 jaar woon in u huishouding?

11. What is your monthly household income, net income after taxes and fringe benefit deductions? (Optional) (Choose only one option)

Wat is u huishouding se maandelikse inkomste, die netto inkomste na all aftrekkings, insluitende belasting? (Opsioneel) (Kies slegs een opsie)

Less than / minder as R1000 R1000 to / na R2000 R2000 to / na R3000 R3000 to / na R4000 R4000 to / na R5000 R5000 to / na R6000R6000 to / na R7000 R7000 to / na R8000 R8000 to / na R9000R9000 to / na R10000 R10000 and more / en meer

SECTION B: MEDICAL HISTORY / AFDELING B: MEDIESE GESKIEDENIS

What was your (or the child's mother's age) when giving birth to him/her?
 ___years.

Hoe oud was u (of die kind se moeder) toe u aan hom/haar geboorte geskenk het?

 How many weeks were you (or the child's mother) pregnant when you gave birth to this child? _____ weeks

Hoeveel weke was u (of die kind se moeder) swanger toe die kind gebore is? ______ weke

- 3. What was the birth weight of this child (as indicated on the "road to health" chart)? _____kg
- Wat was die geboortegewig van die kind (soos op die "road to health" kaart aangetoon)? _____kg
- 4. Does the child's father, mother, grandparents, brother or sister suffer from any of the following conditions?
- Ly die kind se vader, moeder, grootouers, broer of suster aan enige van die volgende toestande?
- a. Type 2 diabetes mellitus / Tipe 2 diabetes mellitus
- b. Hypertension / Hipertensie
- c. Stroke / Beroerte
- d. High cholesterol / Verhoogde kolesterol
- 5. How many people in the household smoke? _____

Hoeveel mense in u huishouding rook? _____

SECTION C: DIETARY INFORMATION / AFDELING C: DIEËET INLIGTING

Please answer the following questions as accurately as possible. Beantwoord asseblief die volgende vrae so akkuraat as moontlik.

2. How many times per week does the child eat take-away foods? _____ Hoeveel keer per week eet die kind wegneem kos?

3. How many times per week does the child drink cold drinks or fruit juice? _____ Hoeveel keer per week eet die kind koeldrank of vrugtesap?

4. How many times per week does the child eat fresh fruit? ______ Hoeveel keer per week eet die kind vars vrugte?

4. Was this child ever breastfed/fed breast milk?

Is die kind ooit geborsvoed / moedersmelk gevoer?_____

a. Yes / ja

b. No / nee

- c. Don't know / weet nie
- 5. How long after birth was this child first breastfed?

Hoe lank na geboorte is die kind die eerste keer geborsvoed?_____

- a. Immediately / onmiddellik
- b. Less than one hour / minder as een uur
- c. Less than 24 hours / minder as 24 uur
- d. More than 24 hours / meer as 24 uur
- e. I don't know / weet nie
- 6. How was this child breastfed for the first six months?

Hoe is die kind vir die eerste ses maande geborsvoed

- a. Breastfeeding exclusively / slegs borsvoeding
- b. Both breastfeeing and formula milk / cow milk / beide moedersmelk en beesmelk
- c. Other, specify / ander, spesifiseer _____
- 7. How was expressed milk given?

Hoe is uitgedrukte borsmelk gegee?

- a. Bottle with teat/nipple / bottel met tiet
- b. Cup / beker
- c. Spoon / lepel

- d. Other, specify / ander, spesifiseer _____
- e. I don't know / weet nie
- 8. For how long was this child breastfed / fed breast milk? _____ months. Hoe lank is die kind geborsvoed / moedersmelk gegee? _____ maande.
- 9. How old was this child when milk/food other than breast milk were given for the first time? _____ months

Hoe oud was die kind toe melk/kos buiten moedersmelk vir die eerste keer gegee is? maande

10. What was the first drink other than breast milk that this child received?

Wat was die eerste vloeistof, buiten moedersmelk wat die kind ontvang het?

- a. Infant formula / baba formule
- b. Water
- c. Gripe water / krampwater
- d. Sugar water / suikerwater
- e. Other, specify / ander, spesifiseer_____

SECTION D: PHYSICAL ACTIVITY / AFDELING D: FISIESE AKTIWITEIT

(We want to know how much time your child spends doing different things each day. Please put an X in one box for each question to tell us the number of hours your child spends doing each activity)

(Ons wil weet hoeveel tyd u kind aan verskillende aktiwiteite spandeer. Maak asseblief 'n X in een van die opsies vir elke vraag om aan te toon hoeveel uur u kind aan elke aktiwiteit spandeer)

1. In a typical week, how much time does your child usually spend watching television, DVD's or videos?

In 'n tipiese week, hoeveel tyd kyk u kind televisie, DVD's of videos?

None

Geen Less than 1 hour / minder as 1 uur 1 - 2 hours / ure 3 - 5 hours / ure 6 - 10 hours / ure 11 - 14 hours / ure 15 - 20 hours / ure More than

20 hours / meer as 20 ure

- In a typical week, how much time does your child spend playing video games, cell phone games, or internet to e-mail or chat online? (Example: Nintendo, XBOX, PlayStation, WhatsApp, Mxit, etc).
- In 'n tipiese week, hoeveel tyd spandeer u kind aan die speel van videospeletjies, selfoonspeletjies, internet, e-pos of aanlyn gesels?

None

- Geen Less than 1 hour / minder as 1 uur 1 2 hours / ure 3 5 hours / ure 6 – 10 hours / ure 11 – 14 hours / ure 15 – 20 hours / ure More than 20 hours / meer as 20 ure
- In a typical week, how much time does your child spend reading outside of school? This includes reading books, magazines, newspapers, doing homework.
- In 'n tipiese week, hoeveel tyd spandeer u kind aan lees buite skool ure? Dit sluit in die lees van boeke, tydskrifte, koerante en die doen van huiswerk.

None

- Geen Less than 1 hour / minder as 1 uur 1 2 hours / ure 3 5 hours / ure 6 – 10 hours / ure 11 – 14 hours / ure 15 – 20 hours / ure More than 20 hours / meer as 20 ure
- 4. In comparison to your children how would you rate your physical activity level at present?

In vergelyking met u kinders, hoe sal u, u eie huidige fisiese aktiwiteitsvlak grader? Less active / minder aktief The same / dieselfde More active / meer aktief

5. When you were a child were you less or more physically active than your child?
Toe u 'n kind was, was u meer of minder aktief as u kind?
Less active / minder aktief The same / dieselfde More active / meer aktief

6. How do your children get to school?

Hoe kom u kinders by die skool?

Walk / loop Run / hardloop Bus / taxi Car / motor Bicycle / fiets

7. How did you get to school as a child?
Hoe het u as kind by die skool gekom? Did not go / nie gegaan Walk / loop Run / hardloop Bus / taxi Car / motor Bicycle / fiets Contact details: Kontakbesonderhede Parent/Guardian's name (Please print your name in full) Naam van ouer/voog (Skryf u naam asseblief volledig)_____ Parent/Guardian's gender (please select your answer) Male/Female Geslag van ouer/voog (kies die antwoord) Manlik/Vroulik Parent/Guardian's age: _____years. Ouer/voog se ouderdom _____jaar. Landline Phone No_____Cellular Phone No_____ Telefoonnommer ______Sellulêre telefoonnommer _____ Name of your child (Please print child's name in full) Naam van die kind (Skryf die kind se naam asseblief volledig) Name of School your child attends Naam van u kind se skool Gender of your child (Please select your answer) Male/Female Geslag van u kind (Kies die antwoord) Manlik/Vroulik Parent/Guardian's signature_____Date_____Date_____ Handtekening van ouer/voog Datum_____

(Thank you for completing this Questionnaire / Dankie dat u die vraelys voltooi het).



ADDENDUM H:

Private Bag X1106, Sovenga, 0727, South Africa

Tel: (015) 268 2817, Fax: (015) 268 2209, Email:Marlise.vanstaden@ul.ac.za

CHILD HEALTH PROJECT

Today your child _____ aged _____ participated in the Child Health Project that you gave consent for. The results are attached below.

Body fatness

To measure body fatness we use a term called BMI. A high BMI indicates large body fatness and a low BMI indicates low body fatness. We all need to have some fat in our bodies; however having too much fat is unhealthy. Below is an indication of your child's body fatness. If your child does not have a healthy BMI, you may speak to a health practitioner who will be able to assist you to solve the problem.

BMI	Body fatness	Your child's BMI
Less than 18.5 kg/m ²	Thin	
18.5 – 24.9 kg/m ²	Healthy	
25 – 29.9 kg/m ²	Overweight	
30 kg/m ² or more	Obese	

If you have any questions, please feel free to contact

Ms MN Mokabane, Department of Physiology and Environmental Health, University of Limpopo, Q-Block, 1st floor, Phone Number: 015 268 2209, E-mail address: donnah.mokabane@gmail.com,

OR

Dr. Marlise van Staden, Department of Physiology and Environmental Health, University of Limpopo, Q-Block, 1st floor, Phone Number: 015 268 2817, E-mail address: marlise.vanstaden@ul.ac.za,

Kind regards

N Made